The role of Toll-like receptors in Parkinson's disease



Catherine Beatrice Kennedy

Department of Clinical Neurosciences University of Cambridge

This thesis is submitted for the degree of Doctor of Philosophy

May 2023

DECLARATION

This thesis is the result of my own work and includes nothing which is the outcome of work done in collaboration except as declared in the Preface and specified in the text. I further state that no substantial part of my thesis has already been submitted, or, is being concurrently submitted for any such degree, diploma or other qualification at the University of Cambridge or any other University or similar institution except as declared in the Preface and specified in the text. It does not exceed the prescribed word limit for the relevant Degree Committee.

ABSTRACT

Inflammation in the brain has been strongly implicated in the development and progression of Parkinson's disease (PD). Neuroinflammation in PD is thought to occur in response to aberrant α -synuclein aggregates, and may be mediated by Toll-like receptors (TLRs). TLRs are pattern recognition receptors with the ability to detect damage-associated molecular patterns. The activation of TLRs, potentially occurring through the detection of aggregated α -synuclein, triggers the release of pro-inflammatory cytokines. This may result in chronic inflammation, which produces an environment toxic to neurons and thus causes neurodegeneration such as that seen in the substantia nigra of the PD brain. I hypothesise that the blockade of TLRs will result in the slowing of pathological progression in an animal model, and the blockade of an α -synuclein inflammatory response in human PBMCs.

Chapter 1 describes the clinical and pathological characteristics of PD and some of the suggested mechanisms behind the progression of pathology, focusing on neuroinflammation. Additionally, it introduces TLRs and summarises the current literature surrounding their connection to PD, in particular that of TLRs 2 and 4.

Chapter 2 describes the reproduction of the Kuan et al., (2019) α -synuclein animal model, including behavioural and neuropathological features. To test the effect of TLR blockade on the development of pathology the drug candesartan was used. This is a licensed drug used in the treatment of hypertension through its action as an AT₁ receptor blocker, but has also been identified as decreasing the expression of TLRs 2 and 4. This work demonstrated a protective effect of TLR blockade on the development of cholinergic neurodegeneration and olfactory deficits, both features of early human PD.

Chapter 3 investigates the mechanism behind the protective effects of candesartan seen in chapter 2. Candesartan showed only small trends towards decreasing the expression and protein levels of TLRs 2 and 4. Neuroinflammation was not identified within the brain, and thus candesartan did not appear to have any effects on this inflammation. Peripheral inflammation was identified at a 2-month timepoint, and candesartan decreased this inflammation, providing a potential protective mechanism.

Chapter 4 describes the optimisation of an *in vitro* assay to investigate the inflammatory response of human peripheral immune cells to different forms of α -synuclein. This work showed that human peripheral immune cells show a pro-inflammatory response to both monomeric and oligomeric α -synuclein, and that this is blocked using candesartan. This response was shown to be TLR4- but not TLR2-dependent.

Chapter 5 summarises the main conclusions of my thesis. Candesartan is having a protective effect on the development of neurodegeneration within the α -synuclein animal model, however the mechanisms behind this are unclear. This protective effect could arise through the suppression of peripheral inflammation, through the inhibition of autophagy through decreasing the expression of TLRs on neurons, or through the blockade of AT₁ receptors. Further evidence supporting an anti-inflammatory mechanism arises from the decrease in α -synuclein-mediated inflammation in *in vitro* human peripheral immune cells on candesartan treatment. This thesis finishes with suggestions for further work to provide more insight into the mechanisms behind the protective action of candesartan, and to provide further rationale for the use of candesartan in a clinical trial.

ACKNOWLEDGEMENTS

Firstly, I would like to extend my sincere gratitude to my PhD supervisor Dr Caroline Williams-Gray. Thank you for your support, your guidance, your willingness to answer all of my questions, and your help in dissuading my constant worries over the past 4 years. I am grateful for the time spent in your lab group and look forward to seeing all of the amazing research which continues to flow out of the group after I leave. I also thank my second supervisor Professor Roger Barker for his encouragement and wisdom.

I would like to thank my PhD examiners Professor Maria Grazia Spillantini and Dr Janelle Drouin-Ouellet. Thank you for your time in reading and assessing my thesis.

I am deeply grateful to Dr William Kuan for his assistance at so many stages of the research project. Thank you for your constant support, your advice (both lab and life-based), for allowing me to use your precious animal model, and for being willing to have drinks in the basement at little to no provocation. Shaline, you are an absolute star. Thank you for your unwavering support and belief in me. Thank you for all of the teaching, the guidance, the willingness for endless trouble-shooting, the laughter, and for keeping up my spirits even in the lowest times. I truly could not have got through the last 4 years without the both of you. I also would like to offer my special thanks to Dr Antonina Kouli for her patience in teaching me, for always being willing to answer any and all questions, for all of her generous and kind support when she was also incredibly busy, and finally for co-authoring the TLR review with me.

Thank you to the Mira staff, to Sarah Fox, and to Pam Tyers for all of their hard work with taking care of the rats, for dosing, and for their patience in teaching me.

Thank you to all members of the Williams-Gray and Barker lab groups for their welcoming attitude, their enthusiasm for science, and for all of their help over the years. I would particularly like to thank Sophie and Maha for always being willing to have a good complain with me, for sharing all the frustrations of a PhD, and for sharing well-earned drinks when necessary. Thank you also to Jonny for his advice and generous assistance with many aspects of the PBMC work, in particular helping out when I had broken my arm. Thank you to Lakmini for her help with the PBMC work, and for buying me chocolate when I was feeling

down. I would also like to offer my special thanks to Marta for her time in carrying out clinic database enquiries, and for being such a caring and supportive person.

Danya, thank you for being the absolute best housemate and friend. Thank you for always being up for a good science chat and for patiently listening to whatever the nightmare of the week was. Thank you for sharing in my love of trashy tv, musicals, chocolate, being very silly together, and for sharing your snacks with me. I am so happy to have shared this PhD journey with you and can't wait to see all of the successes coming from your own PhD.

Thank you to my best friends Ciara, Sophie, and Jess. Thank you for always being on my side, for listening to me complaining and stressing, for distracting me from my little PhD bubble, and for all your love throughout this time.

Thank you to many loving and supportive friends in Cambridge, and to church family who have kept me going through the difficult times.

Michael, you are my absolute favourite. Thank you for loving me and caring for me so well through this PhD. Thank you for being a shoulder to cry on, for lifting me up and keeping me going in the hard times, and for your endless and patient encouragement and support. Thank you also for all of the laughter and fun, for distracting me from the stress, for being interested and interesting, and for loving me through it all. I cannot wait to marry you.

Finally, thank you to my family. Mum and Dad I could not have done this without your unwavering love and support. Thank you for being there throughout.

TABLE OF CONTENTS

CHAPTER 1.	INTRODUCTION18	
1.1 WHAT IS PA	RKINSON'S DISEASE?	
1.1.1 Clinica	l characteristics	20
1.1.2 Patholo	peical characteristics	22
1.1.3 Disease	progression	$\overline{23}$
1.2 WHAT CAUS	es Parkinson's disease?	
1.2.1 Neurod	egenerative mechanisms	26
1.3 NEUROINFLA	AMMATION IN PARKINSON'S DISEASE	
1.3.1 Evidence	e that immune activation plays an aetiological role in PD	28
1.3.2 The neu	proinflammatory hypothesis of PD	30
1.3.3 Microg	lia	31
1.3.4 Periphe	eral immune system	33
1.4 Toll-like r	ECEPTORS	
1.4.1 Functio	n and localisation	34
1.4.2 Toll-lik	e receptors in Parkinson's disease	37
1.4.3 Potentie	al therapeutic strategies	48
1.5 SUMMARY A	ND HYPOTHESIS	
1.6 OBJECTIVES		
CHAPTER 2.	INVESTIGATING HOW THE USE OF CANDESARTAN AFFE	CTS
THE PROGRES	SION OF AN ANIMAL MODEL OF A-SYNUCLEINOPATHY54	
2.1 AIMS OF CH	APTER	
2.1.1 Objecti	ves	56
2.2 Introducti	ON	
2.2.1 Animal	models of Parkinson's disease	57
2.2.2 Candes	artan	60
2.3 Methods		
2.3.1 Study d	esign	61
2.3.2 Animal	s, PFF-injection, and candesartan treatment	62
2.3.3 Prepare	ation of α -synuclein pre-formed fibrils	62
2.3.4 Animal	culling and brain preparation	63
2.3.5 Blood s	amples	63
2.3.6 Immuno	oprecipitation	63
2.3.7 Behavio	oural tests	64
2.3.8 Immuno	ohistochemistry	65
2.3.9 Immuno	ofluorescence	0/
2.3.10 Statist	ical analysis	0/
2.4 RESULTS		
2.4.1 Study d	esign	68
2.4.2 What ef	tect does the injection of α -synuclein PFFs have on endogenous aggreg	ation
of α -synuclein	<i>i</i> within the rat brain?	08 73
2.4.5 Does co	indesarian ireatment affect the development of olfactory deficits?	/3 71
2.4.4 Does co 2.4.5 Does tr	eatment with candesartan affect the survival of cholinergic neurons?	74 77

2.4.6 Does candesartan treatment have an effect on dopaminergic deg	eneration in the
2 5 Discussion	85
2.5 Discussion	
2.5.1 Summary 2.5.2 Lack of dopamineroic degeneration	86
2.5.2 Each of a synuclein pathology $2.5.3$ Lack of a synuclein pathology	90
2.5.4 Lack of olfactory deficits	91
2.5.5 Parallel studies	92
2.5.6 The use of candesartan	93
CHAPTER 3. INVESTIGATING HOW CANDESARTAN IS	HAVING A
PROTECTIVE EFFECT ON THE PROGRESSION OF THE A-SYNUC	LEINOPATHY
RAT MODEL	98
3.1 AIMS OF CHAPTER	
3.1.1 Objectives	100
3.2 INTRODUCTION	
3.2.1 Neuroinflammation in Parkinson's disease	101
3.2.2 Investigating inflammation in α -synuclein PFF model	101
3.2.3 Determining inflammation within the animal brain	102
3.3 Methods	
3.3.1 Study design	106
3.3.2 Liquid chromatography mass spectrometry	106
3.3.3 qPCR	106
3.3.4 Western blot	107
3.3.5 Immunohistochemistry	108
3.3.6 Meso Scale Discovery assay	113
3.3.7 Statistical analysis	114
3.4 Results	
3.4.1 Does candesartan cross the blood brain barrier?	115
3.4.2 Is candesartan successfully reducing the expression of TLR2/4?	118
3.4.3 Does candesartan treatment impact microglial activation?	125
3.4.4 Is candesartan treatment associated with changes in downstrea	im inflammatory
markers? 3.4.5 Does candesartan treatment influence serum peripheral immune fe	134 actors? 137
2.5 Discussion	140
3.5 DISCUSSION	
3.5.1 Summary	140
3.5.2 Lack of neuroinflammation 2.5.3 Alternative mechanism of action of earldesartan	141
5.5.5 Alternative mechanism of action of canaesarian	144
CHAPTER 4. INVESTIGATING WHETHER BLOCKING TLR	S ALTERS A-
PEOPLE WITH PARKINSON'S DISEASE	NE CELLS OF 148
4.1 AIMS OF CHAPTER	150
4.1.1 Objectives	150
4.2 Introduction	
4.2.1 Monocytes	151
4.2.2 Toxic species of α -synuclein	153

4.2.3 TLR blockers		154
4.3 Methods		
4.3.1 PBMC extraction		155
4.3.2 PBMC thawing		155
4.3.3 Monocyte separation		155
4.3.4 Stimulation assays		156
4.3.5 Preparation of α -synuclein pre-formed fibrils		156
4.3.6 Preparation of α -synuclein oligomers		157
4.3.7 Endotoxin testing		157
4.3.8 Super-resolution microscopy		157
4.3.9 Enzyme-linked immunosorbent assays (ELISA)		158
4.4 Results	161	
4.4.1 Monocytes show a high basal level of inflammation		161
4.4.2 The α -synuclein used is endotoxin-free		162
4.4.3 PBMCs are responsive to monomeric and oligomeric α -synuclein		163
4.4.4 PBMCs respond to oligometric α -synuclein in a dose-dependent manner		167
4.4.5 The inflammatory response is TLR4 but not TLR2- dependent		167
4.5 DISCUSSION	171	
4.5.1 Summary		171
4.5.2 Assay failures		172
4.5.3 The inflammatory response to monomers versus oligomers		176
4.5.4 TLR blockers		177
CHAPTER 5. DISCUSSION		
5.1 Introduction		
5.2 Key findings		
5.3 GENERAL DISCUSSION		
5.4 FUTURE DIRECTIONS		
5.4.1 Further exploration of candesartan		188
5.4.2 Human PBMC studies		191

REFERENCES 195

LIST OF TABLES

TABLE 1.1 SUMMARY OF STUDIES EVALUATING TLR2 IN PD PATIENTS. 39
TABLE 1.2 Summary of studies evaluating TLR2 in in vitro and in vivo PD models42
TABLE 1.3 SUMMARY OF STUDIES EVALUATING TLR4 IN PD PATIENTS. 43
TABLE 1.4 SUMMARY OF STUDIES EVALUATING TLR2 IN IN VITRO AND IN VIVO PD MODELS. $.47$
TABLE 2.1 ANTIBODIES USED FOR IMMUNOHISTOCHEMISTRY ON BRAIN SECTIONS
TABLE 2.2 ANTIBODIES USED FOR A-SYNUCLEIN IMMUNOHISTOCHEMISTRY ON BRAIN SECTIONS
TABLE 3.1 TAQMAN PRIMERS USED FOR QPCR. 107
TABLE 3.2 ANTIBODIES USED FOR WESTERN BLOTTING. 108
TABLE 3.3 ANTIBODIES USED FOR IMMUNOHISTOCHEMISTRY ON BRAIN SECTIONS

LIST OF FIGURES

FIGURE 1.1 SELF-PERPETUATING CYCLE OF DEGENERATION
FIGURE 1.2 CELLULAR LOCALISATION AND SIGNAL TRANSDUCTION OF HUMAN TOLL-LIKE
RECEPTORS (TLRS)
FIGURE 2.1 TIMELINE OF STUDY DESIGN
FIGURE 2.2 THE STAIRCASE TEST
FIGURE 2.3 MODEL CHARACTERISATION
FIGURE 2.4 PFF AND CONTROL STUDIES WERE CARRIED OUT IN PARALLEL
FIGURE 2.5 A-SYNUCLEIN-42 STAINING IN THE VAGUS NERVE NUCLEUS71
FIGURE 2.6 A-SYNUCLEIN 2A7 STAINING IN THE VAGUS NERVE NUCLEUS
FIGURE 2.7 A-SYNUCLEIN-211 STAINING IN THE VAGUS NERVE NUCLEUS72
FIGURE 2.8 A-SYNUCLEIN STAINING IN THE SUBSTANTIA NIGRA
FIGURE 2.9 TIMECOURSE OF OLFACTORY DEFICITS
FIGURE 2.10 TIMECOURSE OF MOTOR DEFICITS
FIGURE 2.11 WEIGHT AS A CONFOUNDING VARIABLE
FIGURE 2.12 CANDESARTAN TREATMENT HAS A PROTECTIVE EFFECT ON CHOLINERGIC NEURON
SURVIVAL IN THE VAGUS NERVE NUCLEUS
FIGURE 2.13 CANDESARTAN TREATMENT HAS A PROTECTIVE EFFECT ON THE SURVIVAL OF
CHOLINERGIC NEURONS IN THE HYPOGLOSSAL NUCLEUS
FIGURE 2.14 DOPAMINERGIC DEGENERATION IN THE SUBSTANTIA NIGRA IS NOT OCCURRING IN
THIS WORK DESPITE DEGENERATION BEING SEEN IN THE ORIGINAL KUAN MODEL 82
FIGURE 2.15 NO EFFECT OF CANDESARTAN ON STRIATAL DOPAMINERGIC FIBRE DENSITY84
FIGURE 3.1 MICROGLIAL ACTIVATION SPECTRUM
FIGURE 3.2 SCANNED SLIDES FOR IHC QUANTIFICATION OF IBA1 STAINING109
FIGURE 3.3 IMAGESCOPE SOFTWARE ANALYSIS
FIGURE 3.4 IMAGEJ SOFTWARE ANALYSIS
FIGURE 3.5 CANDESARTAN IS ABLE TO ENTER THE BRAIN, AND PERSISTS FOR 24 HOURS116
FIGURE 3.6 CANDESARTAN PERSISTS IN THE LIVER
FIGURE 3.7 CANDESARTAN TREATMENT DOES NOT HAVE AN EFFECT ON THE EXPRESSION OF
TLRS IN THE RAT BRAINSTEM118
FIGURE 3.8 CANDESARTAN TREATMENT RESULTS IN A TREND TOWARDS DECREASED PROTEIN
LEVELS OF TLRS 2 AND 4 IN THE RAT MIDBRAIN

Figure 3.9 Candesartan treatment results in a trend towards decreased protein
LEVELS OF TLR2 BUT NOT OF TLR4 IN THE RAT BRAINSTEM122
Figure 3.10 Candesartan treatment does not alter protein levels of MyD88124
Figure 3.11 Candesartan treatment does not alter AT_1 expression
Figure 3.12 Candesartan does not have any effect on microglial number or
MORPHOLOGY IN THE HYPOGLOSSAL AND VAGAL NERVE NUCLEI
Figure 3.13 Candesartan does not have any effect on microglial number or
MORPHOLOGY IN THE SUBSTANTIA NIGRA127
Figure 3.14 Candesartan does not have any effect on microglial number or
MORPHOLOGY IN THE STRIATUM
Figure 3.15 Candesartan has no effect on MHCII expression at 6 months130 $$
$Figure \ 3.16 \ Candesartan \ \text{has no effect on } MHCII \ \text{expression at different timepoints}.$
FIGURE 3.17 NO DIFFERENCES IN INFLAMMATION (AS ASSESSED USING MHCII) ARE SEEN
BETWEEN PFF-INJECTED AND SALINE-INJECTED CONTROL ANIMALS WHEN AT COMPARABLE
AGES
FIGURE 3.18 CANDESARTAN IS NOT ASSOCIATED WITH CHANGES IN MHCII PROTEIN LEVELS.
Figure 3.19 ED1 staining is difficult to identify in an animal with an inflammatory $% \mathcal{F}(\mathcal{F})$
INSULT IN THE CEREBELLUM BEING USED AS A POSITIVE CONTROL
Figure 3.20 $CD163$ is not differentially expressed in different treatment groups.
Figure 3.21 Candesartan treatment is not associated with alterations in the
EXPRESSION OF DOWNSTREAM PRO-INFLAMMATORY CYTOKINES IN THE BRAINSTEM 135
Figure 3.22 Western blots of TNF-a were unsuccessful at identifying a band at the
CORRECT MOLECULAR WEIGHT FOR TNF-A (26 KDA)136
Figure 3.23 Western blots were unsuccessful at identifying band at the correct
MOLECULAR WEIGHT FOR NF-KB (65 KDA)137
FIGURE 3.24 PERIPHERAL SERUM CYTOKINE LEVELS OVER TIME IN CANDESARTAN TREATED
VERSUS UNTREATED ANIMALS
FIGURE 4.1 FAILED ELISA PLATES USING THE INVITROGEN KIT
FIGURE 4.2 DILUTION GRADIENTS OF VARIOUS SUBJECT'S CELLS
FIGURE 4.3 TNF-A CONCENTRATION IN SUPERNATANT OF FRESH OR FROZEN CELLS IN RESPONSE
TO STIMULATION 162

FIGURE 4.4 TNF-A CONCENTRATION IN SUPERNATANT OF PBMCs in response to
STIMULATION164
Figure 4.5 Super resolution microscopy and SiMPull of a-synuclein aggregates in
SUPERNATANT FROM STIMULATED CELLS
FIGURE 4.6 INFLAMMATORY RESPONSE ON STIMULATION WITH OLIGOMER CONCENTRATION
GRADIENT
Figure 4.7 Effect of TLR blocking treatment on TNF-a release from stimulated
CELLS
Figure 4.8 Effect of TLR blocking treatment on TNF-a release from stimulated
CELLS

Chapter 1. INTRODUCTION

DECLARATION

Chapter 1.4 "Toll-like receptors" is a modified excerpt of the published review article "Toll-like receptors and their therapeutic potential in Parkinson's disease and α -synucleinopathies". This article was published in Brain, Behaviour, and Immunity: <u>https://doi.org/10.1016/j.bbi.2019.06.042</u> under Green open access with a 12-month embargo which has now lifted. The accepted manuscript is freely available to view within Apollo – the University of Cambridge repository at:

https://www.repository.cam.ac.uk/items/2abbd448-b6b5-4aa5-a22c-22279f75ecd9

The author contributions are as follows: I, Catherine Kennedy (née Horne), and Dr Antonina Kouli are joint first authors and equally contributed to the writing of the text. The figures were produced by Dr Antonina Kouli. Critical feedback was provided by Dr Caroline Williams-Gray in addition to the editing of the final manuscript.

The use of this article in this thesis is according to the Elsevier permission guidelines whereby "Authors can include their articles in full or in part in a thesis or dissertation for non-commercial purposes".

1.1 What is Parkinson's disease?

Parkinson's disease (PD) is the second most common neurodegenerative disease after Alzheimer's disease. It was first described by James Parkinson in 1817. The current estimate for PD prevalence in the UK puts number of cases at around 145,000 according to Parkinson's UK. The prevalence of PD is growing (Ou et al., 2021), and indeed is predicted to grow at an even faster rate within the next 20 years (Dorsey et al., 2018). This growth is driven by its being an age-related disease and the continuing improvements in medicine resulting in increased longevity and consequently in a growing ageing population. This has made Parkinson's a key area for research to improve quality of life for millions of people.

1.1.1 Clinical characteristics

There are four key motor characteristics used in the diagnosis of PD: bradykinesia, tremor at rest, rigidity, and postural instability. Bradykinesia is the slowness of movement; this results in a slowness in performing daily living activities. Tremor at rest is a very common feature of PD, though not universal. It is often seen at disease onset as described in a study by Hughes et al., (1993) where 68% of 100 patients had a resting tremor at disease onset. It was also reported in this study that 75% of patients experienced tremor at some point during disease duration. Rigidity is manifested through increased resistance to the passive movement of limbs. Rigidity in certain areas can lead to postural deformities, for example axial rigidity can result in spinal deformities such as camptocormia (Protopsaltis et al., 2016). Postural instability is the loss of postural reflexes leading to poor balance and falls. (Reviewed in Palakurthi and Burugupally, 2019).

PD is generally diagnosed when motor symptoms develop. However, it is thought that the disease begins many years prior to this with prodromal non-motor symptoms. Olfactory dysfunction (or hyposmia) is very common in PD, with up to 90% of 81 patients in one study showing deficits, though interestingly it was found that over 70% of these PD patients were unaware of their having this symptom prior to olfactory testing (Doty et al., 1988). Idiopathic olfactory dysfunction is linked to an increase in the probability of developing PD (Haehner et al., 2007; Ponsen et al., 2004). Autonomic disturbances, including gastrointestinal function, are also a common non-motor feature, with constipation occurring as a prodromal feature in some cases. A systematic review of constipation and PD carried out by Adams-Carr et al., (2016) has shown that people with constipation are at a greater risk of developing PD than those without. Furthermore, Camacho et al., (2021) have shown in a long-term incident cohort that the severity of constipation at diagnosis is predictive of faster progression to dementia in PD. Sleep disturbance is a common non-motor symptom of PD, and the sleep disorder RBD (REM sleep behavioural disorder) is another prodromal feature of PD and other synucleinopathies (including multiple system atrophy – MSA – and Dementia with Lewy Bodies – DLB). In a cohort of patients diagnosed with RBD (89 patients) 30% developed a synucleinopathy after 3 years, and by 7.5 years this was at 66% (Postuma et al., 2015). Within 12 years of diagnosis over 70% will develop a form of parkinsonism or dementia (Roguski et al., 2020). In PD, the prevalence of RBD is nearly 50% (Zhang et al., 2017).

As mentioned above, many non-motor symptoms are thought to precede the development of motor symptoms. Some of these symptoms, including gastrointestinal dysfunction, are found outside of the central nervous system – that is to say peripherally. It has been hypothesised that PD is initiated peripherally and then spreads to the brain where the further development of disease leads to the characteristic motor symptoms (Reviewed by Klingelhoefer and Reichmann, 2015). Clinical symptom progression consists of the development in severity of motor symptoms; these continue to worsen for the duration of the disease. Postural instability is a major disease milestone with regards to this having a large impact on quality of life; this occurs around 4 years into the disease course (Evans et al., 2011). In addition, other non-motor related symptoms develop. Pain is a common development in people with PD and is seen in over 70% of patients according to a study carried out by Valkovic et al (2015). Cognitive dysfunction can be a feature of early PD but worsens over disease course and can evolve into dementia later in disease progression; dementia is considered to be a symptom of late-stage PD. In an incident population-representative PD cohort 46% had developed dementia by 10 years from diagnosis (Williams-Gray et al., 2013), and another study reported 83% of surviving patients had dementia after 20 years from PD diagnosis (Hely et al., 2008). Symptoms of psychosis, including hallucinations, are a further feature of late-stage PD. Visual hallucinations are reported in around 30% of patients (Fénelon, 2008). These symptoms are often linked to severe cognitive deficits and dementia.

1.1.2 Pathological characteristics

There are two main pathological characteristics of PD. These are the degeneration of dopaminergic neurons in the substantia nigra (SN), and the presence of α -synuclein aggregates (Lewy bodies) within the brain.

The death of SNpc neurons and the loss of the corresponding neuron terminals in the striatum results in decreased dopamine levels in the nigrostriatal system and subsequent motor deficits. It is thought that motor symptoms occur when neuron loss in the SNpc reaches around 50% (Marsden, 1990; Ross et al., 2004), though this differs depending on study, for example Fearnley and Lees estimated only a 30% loss at the point of symptom development (Fearnley and Lees, 1991). The exact percentage loss is difficult to calculate with high accuracy but it is clear that there is already substantial loss by the time of motor-symptom development. It is thought that the striatal dopaminergic neuron terminals could be the initiation point of degeneration, and that the subsequent loss of dopaminergic neurons in the SN could result from the 'dying-back' of the striatal neurite projections (Bernheimer et al., 1973; Tagliaferro and Burke, 2016). This theory is supported by work in the MPTP monkey model where it was shown that the degeneration of dopamine axon terminals in the striatum preceded the loss of dopaminergic cell bodies in the nigra (Herkenham et al., 1991). The loss of striatal dopaminergic markers at motor symptom onset is also thought to be substantial, with estimates taken from post-mortem studies and radioligand imaging analysis coming to around 50-70% loss (Lee et al., 2000; Scherman et al., 1989). The progression of motor symptoms in PD patients occurs due to the continued degeneration of these dopaminergic neurons and striatal fibres, in addition to other neurotransmitter networks, such as the cholinergic system which is implicated in gait and balance dysfunction (Karachi et al., 2010; Rochester et al., 2012). These lead to further loss of the ability to control and coordinate movement. The amount of degeneration is dependent on the duration of disease (Fearnley and Lees, 1991) and correlates with motor deficit severity, however there is variability in this (Asenbaum et al., 1997).

The presence of α -synuclein aggregated to form Lewy bodies is the second key feature of the Parkinson's brain. The aggregation of proteins within the brain is a characteristic of multiple neurodegenerative diseases including Parkinson's and Alzheimer's. Though the specific function of the protein involved and the location of deposition vary between diseases, this commonality suggests that the protein aggregation and dysfunction in protein handling and clearance mechanisms is in some way toxic to neurons. A-synuclein is a 140 amino acid long protein coded for by the SNCA gene. It is made up of 3 regions: an amino terminus which confers the ability to form α -helices on binding to membranes; a central hydrophobic region which is key to the β -sheet forming potential; and a carboxyl terminus which is unstructured (Stefanis, 2012). It localises to pre-synaptic terminals (George, 2002) and has the ability to bind to membranes; these abilities suggest a potential role in pre-synaptic transmitter release (Cabin et al., 2002; Murphy et al., 2000). A-synuclein was first linked to PD through the discovery of a mutation in the SNCA gene being linked to familial PD (Polymeropoulos et al., 1997). Further alterations to the gene including duplication and triplication have also been found in familial Parkinson's cases (Miller et al., 2004; Singleton et al., 2003). Further study showed that a gene dosage effect was present: a duplication of the gene resulted in late-onset PD whereas a triplication resulted in early-onset PD and dementia (Fuchs et al., 2007). This could indicate that the amount of α -synuclein in the brain is key. Following on from this work α -synuclein was found to be the main component of Lewy bodies (Spillantini et al., 1997). These initial studies have led to the generally accepted theory that aggregated α -synuclein is the toxic species in PD, and that through as yet undetermined mechanisms its aggregation results in the death of dopaminergic neurons in the SN.

1.1.3 Disease progression

Pathological progression is thought to occur via the spread of α -synuclein through the brain along specific anatomical circuits as hypothesised by Braak et al., (2003) in a theory known as Braak's hypothesis. In this study post-mortem PD brains were used to assess the progression of pathology through the brain over time. The first α -synuclein pathology was identified in the dorsal IX/X motor nucleus (the glossopharyngeal/vagal nerve nucleus respectively) or the olfactory bulb; this is identified as Braak stage 1. Pathology continues to spread to other areas of the brain in stage 2, and it isn't until stage 3 that pathology reaches the substantia nigra leading to motor deficits. In the final 3 stages pathology spreads to the temporal cortex followed by the neocortex, this alongside associated cholinergic deficits may underlie the cognitive impairment and dementia seen in late-stage PD. This theory points towards a peripheral first insult, resulting in α -synuclein aggregation which then spreads through the vagal nerve from the enteric nervous system of the gastrointestinal tract. This would explain the presence of hyposmia and gastrointestinal dysfunction prior to the development of motor symptoms and the diagnosis of PD. These two areas are particularly vulnerable to

environmental pathogenic exposure, and as such, a dual-hit hypothesis has been suggested (Hawkes et al., 2007). In this theory a neurotropic pathogen enters the brain through two routes: nasally, with anterograde transport from the olfactory system into the temporal lobe; and gastric, with transsynaptic transmission to the vagus nerve from the gut and retrograde transport into the midbrain. Alternatively, a single-hit hypothesis has been suggested whereby Parkinson's either arises in a 'body-first' situation – α -synuclein pathology initiates in the peripheral nervous system before spreading to the brain through the vagal nerve - or a 'brainfirst' situation – α -synuclein pathology initiates in the olfactory system before secondary spreading to other brain regions (Borghammer et al., 2022). Both of these hypotheses explain the presence of non-motor symptoms prior to the motor symptoms. Further evidence for pathology arising outside of the brain comes from the presence of α -synuclein aggregates in the enteric nervous system (Braak et al., 2007, 2006; Del Tredici et al., 2010; Kupsky et al., 1987; Wakabayashi et al., 1988). These are seen mainly in post-mortem studies, but also in gut biopsies of relatively early PD patients (diagnosis <5 years) (Lebouvier et al., 2008) and in gut biopsies prior to the onset of PD motor symptoms (Hilton et al., 2014; Shannon et al., 2012; Stokholm et al., 2016). Further evidence arises from the discovery that people who have had a vagotomy (the cutting of one or more branches of the vagal nerve; carried out as a treatment for peptic ulcers) have their risk of developing PD reduced by approximately 40% when study results are combined (Liu et al., 2017; Svensson et al., 2015a, 2015b). In addition, recent work investigating a model of prodromal Parkinson's disease – produced through the injection of adeno-associated viruses containing mutated α -synuclein gene (A53T) into the vagus nerve – found that phosphorylated α-synuclein was deposited in the vagal Schwann cells. This resulted in vagus nerve dysfunction, a neuroinflammatory response, and peripheral autonomic dysfunction - in the form of gastrointestinal deficits - which preceded motor deficits and central pathology (Cheng et al., 2023).

The mechanism of propagation of α -synuclein throughout the brain, as suggested by Braak, is a controversial one; there are many differing theories on how this could take place. One hypothesis is that α -synuclein has the ability to act in a prion-like manner. Prions are infectious particles composed solely of a modified protein. Transmission occurs through converting the normal protein into the modified protein through posttranslational misfolding into β -sheet-rich structures; the original misfolded protein acts as a template for this conversion into pathogenic protein (Prusiner, 1998). Supporting evidence for α -synuclein acting in this manner is provided by various studies. The analysis of post-mortem brains of PD patients who received grafts of human embryonic mesencephalic tissue showed that when patient death occurred 18 months after the transplant, the dopaminergic phenotype of the graft appeared normal, however 11-16 years post-transplant the grafts showed evidence of Lewy body-like inclusions (Kordower et al., 2008, 1995; Li et al., 2008; Lindvall et al., 1994). This transmission of pathology could be occurring through α -synuclein acting in a prion-like manner, resulting in the pathogenic form of α -synuclein present in the healthy grafts (Olanow and Prusiner, 2009) Further evidence for this mechanism arises from animal and cellular models of PD. These show similar transmission of pathology into healthy grafts (Desplats et al., 2009; Hansen et al., 2011; Kordower et al., 2011), and also that a PD model can be produced through the injection of α -synuclein pre-formed fibrils (PFFs) which act as a seed to trigger the formation of endogenous α -synuclein into Lewy-like inclusions (Luk et al., 2012, 2009). This supporting evidence does not necessarily provide clear support for the prion hypothesis, and could be argued simply to support a non-cell autonomous neurodegenerative process. Other hypotheses exist to support this, including the role of glial cells in the pathological spread of α -synuclein (Xia et al., 2019), and selective vulnerability of specific neuronal populations (Surmeier et al., 2017).

There is not yet a treatment to delay or halt the progression of PD. Only symptomatic treatments are available, which act to stimulate the dopaminergic system. Dopaminergic drugs are effective at alleviating motor symptoms, however this diminishes as the disease progresses due to the underlying neurodegeneration (Schapira, 2009). No disease-modifying drugs currently exist. In order to successfully modify the disease course, a clearer understanding of the mechanisms which drive disease initiation and development is needed.

1.2 What causes Parkinson's disease?

As discussed above, the pathology of PD involves the spread of aggregated α synuclein through the brain, this is thought to result in the death of dopaminergic neurons. The cause of the initial aggregation of α -synuclein and of its toxicity is unknown. Some theories for the causes of PD do not even involve aggregated α -synuclein, which may be a by-product of other failed cellular mechanisms which are actually responsible for the neurodegeneration. Furthermore, there are some genetic forms of PD which lack Lewy bodies at post-mortem such as *Parkin*-associated PD where Lewy bodies are seen in only a minority of patients, however this is known as a very different form of PD (Doherty and Hardy, 2013). Determining the cause of PD and its progression could provide a solution to preventing, halting, or slowing the progression of the disease.

1.2.1 Neurodegenerative mechanisms

1.2.1.1 Oxidative stress and mitochondrial deficits

Mitochondrial deficits have long been thought to play a role in the development of PD. Initial work on this arose from the discovery that the toxin MPTP caused parkinsonian symptoms in those accidentally exposed to it (Burns et al., 1985; Langston et al., 1983); this was found to occur through its selective inhibition of mitochondrial complex I (Nicklas et al., 1987). This resulted in the death of dopaminergic neurons in the substantia nigra and subsequent parkinsonism. Oxidative damage has been seen in post-mortem idiopathic Parkinson's brains (Reviewed by Greenamyre et al., 2001). The deficit in mitochondrial complex I results in the release of reactive oxygen species (ROS) and it is these which are thought to be toxic to the dopaminergic neurons and result in their degeneration. The susceptibility of the dopaminergic degeneration may arise from these neurons being particularly vulnerable to oxidative stress. This is due to their already producing high levels of ROS through the metabolism of dopamine (Hermida-Ameijeiras et al., 2004). Oxidative stress as a cause of PD does not immediately involve α -synuclein, however studies have shown that oxidative damage can cause the nitration of a-synuclein thus increasing its propensity to misfold and aggregate. This potentially implicates oxidative stress as the primary event resulting in the onset of pathogenesis (Giasson et al., 2000). Further evidence for a role of oxidative stress and mitochondrial deficits in the development of PD arises from studies of gene mutations in familial forms of PD. Multiple gene mutations which result in mitochondrial dysfunction have been linked to Parkinson's including mutations in Parkin (Palacino et al., 2004), PINK1 (Gautier et al., 2008), and DJ-1 (Krebiehl et al., 2010). However, it has also been suggested that oxidative stress and the evidence for this within Parkinson's patients could be a secondary effect of the death of neurons, or of α -synuclein aggregation, and not specifically involved in the aetiology of PD. Evidence for this arises from work showing that the aggregation of α -synuclein in mitochondria resulted in mitochondrial deficits through decreased complex I activity (Devi et al., 2008; Hsu et al., 2000; Liu et al., 2009).

1.2.1.2 Altered proteolysis

A build-up of aggregated α -synuclein could occur due to impaired processes of proteolysis. There are two main systems through which damaged/dysfunctional proteins are degraded in cells: the ubiquitin-proteasome system (UPS) and the autophagy-lysosome pathway. A-synuclein can be broken down by both of these systems (Cuervo et al., 2004; Liu et al., 2003; Webb et al., 2003), and a failure of these systems has been identified in human PD (Alvarez-Erviti et al., 2010; Anglade et al., 1997; Chu et al., 2009; Dehay et al., 2010; McNaught and Jenner, 2001; McNaught et al., 2003; Murphy et al., 2015; Ullrich et al., 2010) and in Parkinson's models (Bai-Yun Zeng et al., 2006a, 2006b). Dysfunction of the UPS/autophagy-lysosome pathway is thought to play a role in the spread of neuropathology through aiding α -synuclein propagation between cells (Poehler et al., 2014). This has been shown through the increased exocytosis of α -synuclein when autophagy is inhibited and the subsequent decreased survival of neighbouring neurons (Alvarez-Erviti et al., 2011; Danzer et al., 2012; Lee et al., 2013). A further failure of the autophagic/lysosomal pathway is seen in those with GBA mutations; mutations in this gene are the most common genetic risk factor for PD. This gene codes for the lysosomal enzyme glucocerebrosidase; mutations can result in the loss of function of this enzyme and can result in dysfunction in the autophagic/lysosomal pathway (Smith and Schapira, 2022). Hence the failure of these protein degradation systems may result in the accumulation of the presumed toxic species of α -synuclein in addition to the propagation of α -synuclein through the brain, and thus may be a key feature of neurodegeneration in PD. Alternatively, it has been shown that α -synuclein overexpression can inhibit autophagy (Tanik et al., 2013; Winslow et al., 2010; Yan et al., 2014); accordingly the altered proteolysis seen in PD could be a secondary effect of the development of the pathology rather than involved in the aetiology.

1.2.1.3 Neuroinflammation

Neuroinflammation has been strongly implicated in both the development and progression of PD. This thesis will focus on the role of inflammation in PD, and in particular on the role that Toll-like receptors might play in the inflammatory response. Hence these topics will now be discussed in detail below.

1.3 Neuroinflammation in Parkinson's disease

The function of the inflammatory response in the brain is to protect and restore neurons and glia within the CNS, particularly neurons, as these cells lack the ability to regenerate (Ransohoff, 2016). However, in addition to neuroprotective functions – such as phagocytosis and the subsequent destruction of misfolded proteins – neuroinflammation can also be detrimental. Neuroinflammation involves the activation of immune cells resident to the central nervous system such as microglia, the infiltration of peripheral immune cells, a resulting increase in the release of pro-inflammatory cytokines/chemokines, and the production of reactive oxygen species (ROS). Chronic neuroinflammation has been shown to occur in many neurodegenerative diseases including Alzheimer's disease and Parkinson's disease, and may accelerate disease development and progression (Frank-Cannon et al., 2009).

1.3.1 Evidence that immune activation plays an aetiological role in PD

There has been some debate over whether brain inflammation and immune activation are a secondary effect of the neurodegenerative process or a primary aetiological driver.

1.3.1.1 Genetics

Mutations in a number of immune system-related genes are known to increase susceptibility to developing PD. These genes include those in the HLA region (Saiki et al., 2010; International Parkinson Disease Genomics Consortium et al., 2011) and those in LRRK2 (Russo et al., 2014) in addition to other immune-related genetic susceptibilities found through pathway analysis of GWAS data (Holmans et al., 2013). HLA (human leucocyte antigen) is region of the genome encoding the major histocompatibility complex which plays a critical role in the presentation of antigens and hence is essential for an effective targeted immune response (Saiki et al., 2010). LRRK2 (leucine-rich repeat kinase 2) mutations are implicated in multiple immune-related disorders including inflammatory bowel disease (IBD) and tuberculosis. They are also known to result in an autosomal-dominant familial PD, and are found in approximately 1% of sporadic PD cases (Simón-Sánchez et al., 2009; Zimprich et al., 2004). How LRRK2 mutations cause PD is not well understood. LRRK2 is highly expressed in many immune cells, and an increase in this expression has been seen in response to inflammatory stimuli such as LPS and IFN- γ (Gardet et al., 2010; Hakimi et al., 2011; Moehle et al., 2012; Reviewed by Wallings and Tansey, 2019). It has been suggested that mutated

forms of LRRK2 may push microglia (immune cells found in the brain) towards an activated pro-inflammatory phenotype in addition to regulating their migration and phagocytic properties (Russo et al., 2014).

1.3.1.2 Epidemiological studies

Further evidence arises from epidemiological studies which have shown a reduced risk of developing PD in people regularly taking nonsteroidal anti-inflammatory drugs (NSAIDs) (Gagne and Power, 2010) and studies showing a reduced PD incidence of a third in patients who have taken immunosuppressant drugs (Racette et al., 2018). Furthermore, studies have shown an increased risk of PD associated with autoimmune disorders. A GWAS study identified 17 loci which overlap between PD and autoimmune disorders including type I diabetes and Crohn's disease (Witoelar et al., 2017). In one study, patients with autoimmune diseases – including multiple sclerosis and Hashimoto's disease – were shown to have a 33% higher risk of developing PD (Li et al., 2012). In addition, there is growing evidence for a link between IBD, a disease involving severe intestinal inflammation, and PD. Individuals with IBD are at a greater risk of developing PD than the general population (Zhu et al., 2021).

1.3.1.3 Animal models

In addition, multiple animal studies have implicated inflammation as an initiating factor in PD. PD models can be produced specifically through the induction of inflammation, such as the LPS model. Central injection of LPS (bacterial endotoxin) into the nigra causes specific dopaminergic degeneration (Castaño et al., 1998); this has been shown to occur through the induction of microglial activation and the subsequent production of proinflammatory cytokines including TNFa and nitric oxide species (Herrera et al., 2000; Lu et al., 2000). Peripheral injection of LPS, such as intraperitoneal injection, also results in the death of nigral dopaminergic neurons specifically. This effect is thought to be due to the induction of peripheral inflammation resulting in the translocation of peripheral immune cells to the site of inflammation and the production of cytokines which are able to cross the blood-brain barrier and induce neuroinflammation and subsequent neurodegeneration (Qin et al., 2007). In further work it was found that the overexpression of TNF- α in the substantia nigra results in the activation of microglia, the recruitment of peripheral immune cells to the CNS, and the death of dopaminergic neurons and associated motor symptoms (De Lella Ezcurra et al., 2010). These models provide compelling evidence for the role of inflammation in the aetiology of Parkinson's. In this work the induction of inflammation is sufficient for the specific death of dopaminergic neurons and no direct toxic insult is required. Double-hit models also exist where an α -synuclein-overexpressing animal is also given an inflammatory stimulus such as LPS. This has been shown to potentiate the toxicity in the model, where this dual hit resulted in persistent neuroinflammation (only early acute inflammation was seen in WT animals), progressive degeneration of dopaminergic pathways, and the formation of Lewy-like pathology (Gao et al., 2011). Neuroinflammation can also be seen in toxin models of PD, such as the MPTP model, the 6-OHDA model, or the rotenone model (Członkowska et al., 1996; Purisai et al., 2007; Srivastava et al., 2012). While inflammation is not the direct cause of degeneration in these models, it does exacerbate the effects of the toxin. Protective effects have been seen in these toxin-induced models through manipulation of the immune system - including the knockout of mature T-lymphocytes (Brochard et al., 2009); the knockout of MHCII (Martin et al., 2016); the knockout of TNF- α (Sriram et al., 2002); and the use of immunosuppressant drugs (Khan et al., 2013; McCoy et al., 2006; Mogi et al., 2000; Purisai et al., 2007; Świątkiewicz et al., 2013). The knockout of key components of the immune system has also been seen to show a protective effect in α -synuclein overexpressing animal models, for example the knockout of MHCII was shown by Harms et al., (2013) to prevent the α -synucleininduced inflammation and dopaminergic degeneration.

1.3.2 The neuroinflammatory hypothesis of PD

The link between neuroinflammation and neurodegeneration is thought to arise through the release of proinflammatory factors by glial cells which have a detrimental effect on neuron survival. Neuron death also leads to further microglial activation. This forms a self-perpetuating cycle of degeneration whereby neuronal death causes microglial activation, resulting in the increased production of pro-inflammatory cytokines & chemokines, and increased oxidative stress, which in turn lead to further neuron loss (Gao and Hong, 2008). (**Figure 1.1**). This hypothesis has been expanded to include the role of the peripheral immune system in this cycle of degeneration. There is evidence in animal models for the activation of peripheral immune cells by antigens which leak into the periphery through cervical lymph nodes (Benner et al., 2008). In addition, peripheral immune cells can be activated by α -synuclein aggregates external to the brain, such as those in the gut (Stolzenberg et al., 2017). These activated peripheral immune cells may then traffic back into the CNS to further activate microglia or to have toxic effects directly on neurons (Sommer et al., 2018). Post-mortem

studies have shown the presence of T-cells in Parkinson's brains, providing further evidence for this hypothesis (Brochard et al., 2009).



Figure 1.1 Self-perpetuating cycle of degeneration. Neuronal death causes the release of antigens such as pathological α -synuclein aggregates. These cause immune activation, resulting in increased production of pro-inflammatory cytokines & chemokines, and increased oxidative stress, which in turn leads to further neuron loss. Antigens, such as aggregated α -synuclein, are released from neurons and can leak into the periphery and activate peripheral immune cells. These may also be involved in this cycle through their trafficking back into the CNS to further activate microglia or to have toxic effects directly on neurons.

1.3.3 Microglia

Microglia are the resident immune cells of the brain. They are found throughout the brain but are particularly concentrated at certain regions, including the hippocampus and substantia nigra (Lawson et al., 1990). Microglia are generally categorised as existing in either a surveying state – originally termed the resting state – or an activated state. Historically this dual categorisation has been strict, however new approaches are moving towards a continuum of activation with 'activated' and 'surveying' states forming the extremes of the spectrum (Mosser and Edwards, 2008). Microglia in the surveying state are described as having ramified

morphology with many extending processes to allow them to constantly survey the local environment for changing conditions (Nimmerjahn et al., 2005). Microglia in the activated state are described as having amoeboid morphology where the cell has a large cell body and few cell processes. The gene expression profiles also differ dramatically between these two states. The activated state is associated with pro-inflammatory functions including the release of pro-inflammatory cytokines, such as TNF- α and IL-1 β , in addition to other inflammatory phenotypic features, such as the upregulation of inducible nitric oxide synthase and the release of reactive oxygen species (Le et al., 2001). The surveying state is associated with immunoregulatory functions, the dampening of the immune system, and tissue repair through the production of anti-inflammatory cytokines (Reviewed by Colton, 2009).

1.3.3.1 Microglia in Parkinson's disease

Activated microglia were first observed in the substantia nigra of patients with PD at post-mortem by McGeer and colleagues in 1988 (McGeer et al., 1988). Subsequent studies reported an increase in the levels of pro-inflammatory cytokines in the striatum of PD brains compared to healthy controls; these included TNF- α , IL-1 β , and IL-6 as well as other proteins related to microglial activation (Knott et al., 2000; Mogi et al., 1994a, 1994b). In agreement with these post-mortem studies, positron emission tomography (PET) neuroimaging using the tracer [¹¹C]-PK11195 has demonstrated increased microglial activation in the brains of PD patients compared to age-matched healthy individuals, and this appears to start early on in the disease course (Gerhard et al., 2006; Ouchi et al., 2005). Furthermore, recent studies using advanced techniques, such as single-nuclei RNA sequencing, have also shown an increase in activated microglia in the nigra of PD patients and that these cells are enriched in cytokine secretion (Smajić et al., 2022).

Microglia have the ability to be activated by multiple extracellular signals, including those from pathogens, dead or dying cells, or foreign material. The activation of microglia in PD has been hypothesised to occur in response to oligomeric α -synuclein. The mechanism by which aggregated α -synuclein activates microglia is still not fully established but a key theory involves α -synuclein working as a damage-associated molecular pattern (DAMP) to activate Toll-like receptors (Beraud et al., 2011).

1.3.4 Peripheral immune system

In addition to inflammation occurring in the Parkinson's brain, it also occurs in the periphery. The overactivation and dysregulation of both the innate and adaptive immune responses results in the overproduction and release of pro-inflammatory cytokines. These then contribute to the further activation and recruitment of immune cells, resulting in chronic inflammation and an environment which is toxic to cells, including dopaminergic neurons. Mean levels of inflammatory cytokines are elevated in the serum in people with Parkinson's compared to age-matched controls, and a pro-inflammatory cytokine response at diagnosis is associated with a more rapid disease progression over time (Brodacki et al., 2008; Qin et al., 2016; Reale et al., 2009; Williams-Gray et al., 2016). In addition to changes to inflammatory cytokines, alterations in peripheral immune cell subsets also occur in people with PD. An increase in classical (inflammatory) monocytes and monocyte activation markers – including TLR2 and 4 – was shown in people with Parkinson's as compared to healthy controls, and this was most pronounced in cases at a higher risk of dementia (Wijeyekoon et al., 2020). Differential T-call subsets have also been identified in PD, with increased effector/memory Tcells and regulatory T-cell dysfunction being associated with disease severity (Saunders et al., 2012) and decreased CD8+ T-cell replicative senescence and an increase in CD8+ cell activation markers in PD (Williams-Gray et al., 2018). A-synuclein may play a key role in the activation of the peripheral immune system. A-synuclein specific T cells have been identified and are present at higher frequency in PD cases than controls; α -synuclein peptides act as antigens to trigger helper and cytotoxic T-cell responses (Sulzer et al., 2017). A-synucleinspecific antibodies, which may act to propagate inflammation, are increased in people with PD; this is particularly seen in early PD (Scott et al., 2018). A-synuclein is also able to activate monocytes as shown through work with the immortalised monocytic cell line THP-1 (Klegeris et al., 2008) and in cultured patient monocytes/PBMCs and other monocytic/microglial cell lines (Codolo et al., 2013; Grozdanov et al., 2019; Liu et al., 2022; Nissen et al., 2019; White et al., 2018).

1.4 Toll-like receptors

A specific aspect of the immune system which has been gaining traction as playing an important role in PD is the involvement of Toll-like receptors (TLRs). These receptors have been implicated in mediating the inflammatory response of microglia and peripheral monocytes

to pathological α -synuclein, in addition to having important roles in the transmission of α -synuclein, and in the altered proteolysis of α -synuclein.

1.4.1 Function and localisation

TLRs are a family of transmembrane pattern recognition receptor (PRR) proteins key to the activation of the innate immune response. Their primary role is in the detection of foreign microbial and viral molecules which they achieve through their recognition of pathogen-associated molecular patterns (PAMPs). TLRs also have the ability to recognise damage-associated molecular patterns (DAMPs) which are found on molecules endogenous to the host (Reviewed in Bianchi, 2007). On their recognition of either PAMPs or DAMPs, TLRs induce a signalling cascade resulting in the activation of an immune response to clear the pathogen from the host or to resolve the damage. TLRs are involved in the activation of both the innate immune system – consisting of non-specific responses such as the initiation of pro-inflammatory cytokine production and phagocytosis – and in a less direct manner the adaptive immune system – consisting of a response specific to the pathogen mediated by T and B lymphocytes, through their upregulation of co-stimulatory molecules required for robust T-cell responses (Hoebe and Beutler, 2004).

The TLR family is made up of 10 functional TLRs in humans. Each of these receptors has the ability to recognise distinct molecular patterns, allowing a wide range of patterns to be recognised and an appropriate immune response to be initiated (Hasan et al., 2005). TLRs are type I transmembrane proteins formed of an extracellular leucine-rich repeat (LRR) domain responsible for specific pattern recognition, a transmembrane region, and a cytoplasmic TIR (Toll/interleukin-1 receptor) domain which transduces the signal to activate the correct downstream signalling pathway (Reviewed by Kawai and Akira, 2011). The human TLR family can be split into two groups based on where the receptors are located in the cell: TLR1, TLR2, TLR4, TLR5, TLR6, and TLR11 are found in the plasma membrane, while TLR3, TLR7, TLR8, and TLR9 are found in intracellular compartments (see **Figure 1.2**). This differential localisation occurs due to the specialisation of the different members of the TLR family to recognise specific pathogens.

TLRs normally function in homodimers, with the exception of TLR2 which forms heterodimers with either TLR1 or TLR6, as well as homodimers (Jin et al., 2007; Kang et al., 2009). Some TLRs work in combination with additional molecules or co-receptors that enable

more sensitive ligand recognition; notably TLR4 requires both MD2 (Shimazu et al., 1999) and CD14 (Wright et al., 1990) to recognise and be activated by bacterial lipopolysaccharides (LPS).



Figure 1.2 Cellular localisation and signal transduction of human Toll-like receptors (TLRs). TLR1, TLR2, TLR4, TLR5, TLR6 and TLR10 are located on the plasma membrane while TLR3, TLR7, TLR8 and TLR9 are found intracellularly on endosomal or lysosomal membranes. Most TLRs are believed to function in homodimers, though TLR2 forms heterodimers with either TLR1 or TLR6. All TLRs except for TLR3 signal through a MyD88 (myeloid differentiation 88)- dependent downstream pathway. Signalling via TLR2 and TLR4 requires an additional intracellular adapter protein TIRAP (TIR domain-containing adapter protein), which in turn recruits the central adapter MyD88. This triggers a downstream signalling cascade resulting in the nuclear translocation of the transcription factor NF- κB and the expression of pro-inflammatory cytokines such as TNF α , IL-1 β and IL6. There is also a MyD88-independent pathway which involves TLR3 and TLR4. Activation of TLR3 engages the adapter protein TRIF (TIR-domain-containing adapter-inducing interferon- β), while TLR4 signalling requires both TRAM (TRIF-related adapter molecule) and TRIF. This results in the downstream activation of the IRF transcription factors (interferon-regulatory factor) which regulate the expression of interferon type 1 genes. Abbreviations: ssRNA: single stranded RNA, dsRNA: double stranded RNA, CpG DNA: cytosine-guanine rich DNA. Image and legend reproduced from review coauthored by myself (Kouli, Horne et al., 2019).

Chapter 1

On PAMP or DAMP recognition by the extracellular LRR domain, the signal is transduced intracellularly, and the correct adaptor proteins (each with a TIR domain) are recruited to and activated by the cytoplasmic TIR domain of the TLR to trigger an appropriate immune response (Akira et al., 2006). There are multiple adaptor proteins which can be activated on pattern recognition by the TLR, and the combination of these proteins determines the specific downstream pathway. MyD88 (myeloid differentiation 88) is one such adaptor protein; it can be activated by all TLRs, with the exception of TLR3, and as such is known as the central adaptor protein (Deguine and Barton, 2014). MyD88 recruitment results in the downstream activation of NFkB (nuclear factor kappa-light-chain-enhancer of activated B cells) and mitogen-activated protein kinases (MAPKs), leading to the activation of the NLRP3 inflammasome and the expression of multiple genes involved in the immune response. These include genes responsible for the production of pro-inflammatory cytokines such as $TNF-\alpha$, IL-1β, and IL-6 (Kawai and Akira, 2011). NFκB can also be activated in a MyD88-independent pathway involving the TLR3 and TLR4 adaptor protein TRIF (TIR-domain-containing adaptor-inducing interferon- β). (Figure 1.2). (Yamamoto et al., 2003). Most signalling pathways resulting from TLR activation can be classified as either MyD88-dependent or TRIFdependent. There are several other molecules and adaptor proteins involved in the TLR signalling pathways which have been extensively reviewed elsewhere (Drouin-Ouellet and Cicchetti, 2012).

In the central nervous system TLRs are expressed in microglia, astrocytes, and neurons. Human and mouse microglia express all members of the TLR family, and their expression varies depending on the stimulation state of the cell and the TLR type (Bsibsi et al., 2002; Olson and Miller, 2004). The most highly expressed microglial TLRs are TLRs 1-4, with TLR2 having the highest expression (Bsibsi et al., 2002). A healthy individual shows low TLR expression on astrocytes, with upregulation on activation, though at a slower rate than that seen in microglia (Trudler et al., 2010). Microglia and astrocytes have distinct functional responses on exposure to TLR ligands, such as differential pro-inflammatory cytokine induction (Jack et al., 2005). Both human and mouse neurons have been shown to express various TLRs, however the expression levels show high variation dependent on the specific TLR (Barajon et al., 2009; Lafon et al., 2006; Ma et al., 2006; Tang et al., 2007; Zhou et al., 2009). In the periphery, TLR expression is highest in tissues which are in close contact with the environment, namely the lungs and the gastrointestinal tract, as well as in peripheral blood mononuclear cells and the spleen (Zarember and Godowski, 2002). Several cell types, both immune and non-immune,
have been found to express TLRs, including dendritic cells, macrophages, B and T lymphocytes (both cytotoxic and helper T lymphocytes, although expression is lower in the latter), natural killer (NK) cells, as well as epithelial, endothelial and fibroblast cells (Fitzner et al., 2008; McClure and Massari, 2014; Muzio et al., 2000; Yao et al., 2015). Expression levels are dependent on the cell type, the specific member of the TLR family, and the stimulation state of the cell (Hornung et al., 2002).

The activation of microglia through TLR engagement by PAMPs results in an inflammatory response induced to clear pathogens, but it can also be induced by endogenous signals (DAMPs) such as heat shock proteins, uric acid, HMGB-1 (high mobility group box chromosomal protein 1), and aggregated, modified or misfolded proteins (Beraud et al., 2011).

1.4.2 Toll-like receptors in Parkinson's disease

Microglial activation in α -synucleinopathies is thought to be triggered by α -synuclein aggregation. The mechanism by which aggregated α -synuclein activates microglia is still being established, but a key theory involves α -synuclein working as a DAMP to activate Toll-like receptors (Beraud et al., 2011). Initial experiments using the BV2 microglial cell line demonstrated that treatment with α -synuclein led to altered expression of several TLRs, including upregulation of TLR1, TLR2, TLR3, and the adapter MyD88, and an increase in expression and release of pro-inflammatory molecules (Beraud et al., 2011). This is similarly seen in other microglial and monocytic cell lines, in addition to in human peripheral blood mononuclear cells (PBMCs) (Bearoff et al., 2023; Codolo et al., 2013; da Rocha Sobrinho et al., 2021). In transgenic animal models, such as Thy1- α -synuclein mice which overexpress wild-type human α-synuclein, there is an increase in TLR1, TLR2, TLR4, and TLR8 mRNA expression in brain regions showing microglial activation (Watson et al., 2012). This upregulation of TLRs is widely seen across a number of different PD models, but the level and timecourse of upregulation depend on the specific model (McCabe et al., 2017). It is worth noting that TLRs have also been shown to be activated in PD models which do not show asynuclein pathology, such as the MPTP model (Drouin-Ouellet et al., 2011; Panaro et al., 2008; Ros-Bernal et al., 2011). Activation of TLRs in this and similar toxin models may be induced by DAMPs related to oxidative stress, such as cell debris or molecules released by damaged or dying neurons including heat shock proteins, uric acid, and others (Bianchi, 2007).

Of particular note, TLR2 and TLR4 have been consistently reported to be upregulated in many α -synuclein-overexpressing or toxin-induced animal models (Drouin-Ouellet et al., 2015; Kim et al., 2013; Noelker et al., 2013; Watson et al., 2012), and accumulating evidence from human studies further implicates these receptors in the pathogenesis of PD.

1.4.2.1 Toll-like receptor 2

In human PD, upregulation of TLR2 has been observed in post-mortem brains. (Summarised in Table 1.1). Specifically, transcriptomic analysis revealed increased TLR2 expression in the substantia nigra, accompanied by an increase in the expression of genes involved in the TLR2 signalling pathway including CD14, IRAK2, and NF-κB (Kim et al., 2013). An increase in TLR2 protein levels was also observed in the caudate/putamem of postmortem PD brains (Drouin-Ouellet et al., 2015). Furthermore, TLR2 was found to co-localise with amoeboid microglia, suggesting upregulation of TLR2 expression by activated microglia in PD brains (Doorn et al., 2014; Kim et al., 2013; Ping et al., 2018). TLR2 has also been reported to be expressed on neurons in human PD brains, and the level of neuronal TLR2 expression was strongly correlated with disease stage and disease duration (Dzamko et al., 2017). However, analysis of publicly available transcriptomic data from the cortex of control post-mortem brains has indicated that TLR2 is almost exclusively expressed on microglia (Hughes et al., 2019), hence neuronal TLR2 expression may be disease specific. TLR2 upregulation, as well as NF-KB activation, have also been identified in peripheral immune blood cells of PD patients (da Rocha Sobrinho et al., 2021; Drouin-Ouellet et al., 2015; Ping et al., 2018; Wijeyekoon et al., 2020).

Table 1.1 Summary of studies evaluating TLR2 in PD patients. (Modified from Tables in Kouli, Horne et al., 2019).

References	Study design	Key observation
Dzamko et al., 2017; Drouin-Ouellet et al., 2015	Human PD and control postmortem brains	TLR2 level is increased in the striatum and anterior cingulate cortex of PD brains.
Ping et al., 2018; Dzamko et al., 2017; Doorn et al., 2014; Kim et al., 2013	Human PD and control postmortem brains	Increased co-localization of TLR2 with activated microglia in PD. Neuronal TLR2 correlates with disease burden and duration.
Ping et al., 2018; Drouin- Ouellet et al., 2015; Shin et al., 2015; da Rocha Sobrinho et al., 2021	Human PD and control PBMCs	TLR2, TLR4, and NF-κB are upregulated in PD PBMCs.
Wijeyekoon et al., 2020	Human PD and control PBMCs and serum	Higher percentage of TLR2+ monocytes in PD patients

PD: Parkinson's disease; PBMCs: peripheral blood mononuclear cells

There is also a growing body of evidence from *in vitro* and *in vivo* studies in animal models implicating TLR2 in the pathogenesis and progression of PD. (Summarised in Table **1.2**). Endogenous α -synuclein binds to TLR2 to trigger microglial activation, and it has been suggested that this occurs in a conformation-dependent manner, with the oligomeric form, but not monomeric or dimeric forms, acting as a TLR2 agonist (Kim et al., 2013). This was further verified with the use of microglia from TLR2-deficient mice. Kim showed diminished microglial activation indicated by a complete loss of cytokine and chemokine upregulation on treatment with endogenous α -synuclein released by human α -synuclein overexpressing SHSY5Y cells, compared to microglia from wild-type mice. The same group later demonstrated that endogenous α -synuclein released from neurons induced TLR2-mediated microglial activation and this appeared to be neurotoxic. This neurotoxicity was entirely abolished by TLR2 gene depletion (Kim et al., 2016). Similarly, Daniele et al (2015) showed that an oligometric form (with higher-ordered structure) of α -synuclein activates microglia, and additionally reported that this is mediated through a TLR1/2 heterodimer, resulting in the nuclear translocation of the p65 NF-kB subunit. However, work by others has contradicted that it is only the oligometric form of α -synuclein which can activate TLRs. Gustot et al., (2015) reported that a-synuclein fibrils, not oligomers or monomers, cause the activation of the NFκB pathway through TLR2 activation *in vitro*. Another study in primary human monocytes suggested that both monomeric and fibrillary a-synuclein instigated a TLR2-mediated inflammatory response, though α -synuclein oligomers were not used in this work (Codolo et al., 2013). A combination of different sized fibrils in the form of PFFs were shown to activate primary mouse microglia and the BV2 microglial cell line, and this was prevented through the use of a TLR2-blocking agent (Dutta et al., 2021). It is possible that these discrepancies result partly from a lack of consistency in the definition and size of oligomers and fibrils across different studies. Although the precise aggregate size which is most pathologically relevant in this context remains uncertain, it seems clear that at least one, if not several species of the protein, can act as DAMPs (Roberts and Brown, 2015). Studies in models of other diseases have similarly suggested that other misfolded proteins can also act as DAMPs, for instance Liu et al (2012) identified TLR2 as the receptor to which amyloid- β binds and initiates microglial inflammatory responses.

Activation of neuronal TLR2 has been suggested to result in increased intracellular asynuclein protein. This was seen in work by Chung et al., (2022) showing an increase in TLR2 activation and a concomitant increase in neuronal a-synuclein on exposure to activated microglia. This was also seen in studies by Dzamko et al., (2017) on the SHSY5Y cell line and iPSC-derived (induced pluripotent stem cell) human neural progenitor cells using the TLR2 agonist PAM3CSK4. It is interesting to note that the use of rapamycin – an inducer of autophagy – ameliorated this TLR2-induced increase in neuronal α -synuclein, potentially implicating autophagy in this pathway. This is further confirmed in a recent study where the activation of neuronal TLR2 impaired the autophagy/lysosomal pathway and potentiated asynuclein pathology; a small molecule TLR2 inhibitor was seen to ameliorate this (Chedid et al., 2022). Building on their earlier work, Kim et al., (2015) proposed that TLR2 activation can lead to an AKT/mTOR-mediated inhibition of autophagy resulting in accumulation of αsynuclein within neurons. In this study, TLR2^{-/-} mice had reduced neuronal α -synuclein accumulation, reduced microgliosis and astrogliosis, and showed improved motor behavioural deficits compared to control mice (Kim et al., 2015). A recent study by Kim et al., (2021) provided further evidence for this showing that stimulation of TLR2 increased α -synuclein fibril uptake in neurons, and delayed the degradation of the α -synuclein within the neurons. Additionally, they showed a change in morphology of microglia on TLR2 activation, confirming previous work on the activation of microglia in response to α -synuclein through TLR2 activation.

TLR2 has also been implicated in the MPTP toxin model of PD, as shown by its upregulation on MPTP treatment, and subsequent upregulation of proinflammatory cytokines

(Chao et al., 2016; Drouin-Ouellet et al., 2011; Jang et al., 2017; Koo et al., 2017; Sun et al., 2021). In α -synuclein overexpressing mice, immunization with anti-TLR2 antibodies resulted in reduced α -synuclein accumulation, reduced microgliosis and pro-inflammatory cytokine secretion, and a reduced memory deficit as shown using behavioural tests (Kim et al., 2018; La Vitola et al., 2018). TLR2 was shown to be activated in the vagal Schwann cells of a model overexpressing mutated α -synuclein (AAV-A53T injected into the vagus nerve); silencing TLR2 expression attenuated α -synuclein-induced GI dysfunction and vagus nerve demyelination and axonal loss. This model shows a role for TLR2 in a prodromal model of PD (Cheng et al., 2023). Furthermore, the use of a TLR2 neutralising antibody in an *in vitro* set of experiments revealed additional roles for TLR2 in the neuron-to-neuron and neuron-to-astrocyte transfer of α -synuclein. Specifically, after TLR2 inhibition the neuron-to-neuron transfer of α -synuclein was markedly reduced. A similar reduction in α -synuclein transfer was observed in a co-culture of SHSY5Y neurons with primary human astrocytes (Kim et al., 2018).

References	Study design	Key observation
Beraud et al., 2011	α-Synuclein-treated BV2-microglia	Upregulation of microglial TLR2.
Daniele et al., 2015; Kim et al., 2013	α-Synuclein-treated WT & TLR2 ^{-/-} mouse microglia/TLR2 antagonists	TLR2-mediated microglial activation; inhibition of TLR1/2 ameliorates inflammation.
Gustot et al., 2015	THP1 cells treated with α -synuclein and anti-TLR2 antibodies	Fibrillar α-synuclein activates TLR2.
Codolo et al., 2013	Human monocytes treated with anti-TLR2 antibodies prior to α -synuclein	Both fibrillar and monomeric α -synuclein activate TLR2.
Dzamko et al., 2017	SHSY5Y and IPSC-derived neural cells treated with a TLR2 agonist.	Neuronal TLR2 activation leads to increase α -synuclein.
Chung et al., 2022	SHSY5Y cells co-cultured with LPS-activated microglia and treated with TLR2 inhibitor	Inhibition of TLR2 mitigates α -synuclein aggregation and improves autophagic flux.
Chedid et al., 2022; Kim et al., 2015	SHSY5Y cells treated with TLR2 agonist and autophagy inducer or inhibitor	Neuronal TLR2 regulates autophagy.
Kim et al., 2021	SHSY5Y cells treated with TLR2 stimulator; upregulation of TLR2 in a synucleinopathy mouse model	TLR2 stimulation accelerates α -synuclein uptake, delays its degradation <i>in vitro</i> , increases Lewy-like inclusions, and activates microglia <i>in vivo</i> .
Kim et al., 2016	WT & TLR2 ^{-/-} mice	TLR2-mediated microglial activation is neurotoxic.
Kim et al., 2015	A53T ⁺ TLR2 ^{+/+} and A53T ⁺ TLR2 ^{-/-} mice	TLR2 ablation reduces neuronal α -synuclein accumulation and improves motor function.
Dutta et al., 2021	PFF-seeded A53T ^{+/+} and A53T ^{ΔTLR2} mice treated with a TLR2 inhibitor.	TLR2 inhibition reduces inflammation and α -synuclein spreading, and protects neurons.
Cheng et al., 2023	AAV-A53T injected into vagal nerve of mice; TLR2 silencing	TLR2 activated in Schwann cells; TLR2 silencing reduced inflammation and pathology.
Chao et al., 2016; Drouin-Ouellet et al., 2011; Sun et al., 2021	MPTP mice	MPTP treatment upregulates TLR2.
Kim et al., 2018	Anti-TLR2 immunisation of α -synuclein overexpressing mice	Anti-TLR2 immunization alleviates neuroinflammation, neurodegeneration and behavioural deficits.
La Vitola et al., 2018	Anti-TLR2 immunisation of WT mice	Anti-TLR2 pre-treatment ameliorates α -synuclein-induced memory deficits.

Table 1.2 Summary of studies evaluating TLR2 in in vitro and in vivo PD models. (Modified from Tables in Kouli, Horne et al., 2019).

PD: Parkinson's disease; WT: wild-type; SHSY5Y: human neuroblastoma cell line; BV2: mouse microglia cell line; THP1 cells: human monocyte cell line

1.4.2.2 Toll-like receptor 4

In human studies, analysis of transcriptomic data from post-mortem control brains reveals ubiquitous expression of TLR4 and the adapter MyD88 throughout the brain, with their expression being higher in the substantia nigra and the putamen (Hughes et al., 2019). Additional studies comparing PD with control brains showed increased TLR4 protein levels in PD cases, specifically in the substantia nigra, the frontal cortex, the amygdala, and the caudate/putamen (Drouin-Ouellet et al., 2015; Kouli et al., 2020; Shin et al., 2015). This increase was also observed in peripheral immune cells in PD patients (Drouin-Ouellet et al., 2015; Ping et al., 2018; Shin et al., 2015) and one study also showed that the increased TLR4 is most prominent in PD cases with a higher risk of dementia (Wijeyekoon et al., 2020). Additional evidence supporting a role of TLR4 in human PD is provided from genetic studies and the discovery that a polymorphism in the TLR4 gene is linked with the risk of PD in a Chinese Han population (Zhao et al., 2015). (Summarised in **Table 1.3**).

Table 1.3 Summary of studies evaluating TLR4 in PD patients. (Modified from Tables in Kouli, Horne et al., 2019).

References	Study design	Key observation
Drouin-Ouellet et al., 2015; Hughes et al., 2019; Kouli et al., 2020; Shin et al., 2015	Human PD and control postmortem brains	TLR4 protein levels are increased in the substantia nigra, caudate/putamen, frontal cortex, and amygdala of PD brains.
Drouin-Ouellet et al., 2015; Ping et al., 2018; Shin et al., 2015	Human PD and control PBMCs	TLR2, TLR4, and NF- κ B are upregulated in PD PBMCs.
Perez-Pardo et al., 2019	Human PD and control intestinal mucosa biopsies	TLR4 and pro-inflammatory cytokine expression is upregulated in the intestinal mucosa of PD patients compared to controls.
Zhao et al., 2015	PD and healthy Chinese Han population	TLR4 polymorphisms are linked to PD.
Wijeyekoon et al., 2020	Human PD and control PBMCs and serum	Higher percentage of TLR4+ monocytes in PD patients, significant increase in TLR4+ monocytes in higher dementia risk group

PD: Parkinson's disease; PBMCs: peripheral blood mononuclear cells

TLR4 has also been implicated through *in vitro* and *in vivo* work in animal models of PD, though some studies support a neuroprotective role while others argue that it acts in a detrimental manner. (Summarised in **Table 1.4**).

Chapter 1

Stefanova et al., (2011) showed that TLR4 ablation leads to impaired microglial phagocytosis of a-synuclein in an *in vitro* model using BV2 microglia. In the same study, TLR4 deficiency was found to increase the dopaminergic neuron loss in the SN of an α -synuclein overexpressing mouse model, in addition to exacerbating the motor problems seen in the model. This neuronal death was proposed to occur due to the lack of clearing of α -synuclein by microglial phagocytosis, implicating TLR4 in this process (Stefanova et al., 2011). Further implicating TLR4 in neuroprotection, work by Mariucci et al., (2018) demonstrated that overexpression of α -synuclein mRNA occurs in TLR4^{-/-} mice in several brain areas, though the causes of this have not been elucidated. One suggestion is that TLR4 plays an important role in the clearance of α -synuclein through the autophagy/lysosomal pathway. Venezia et al., (2021) showed that TLR4 deficiency facilitated the spread of α -synuclein, and that this is associated with a deficit in autophagy-lysosomal activity. Further evidence comes from work specifically showing that neuron-released α -synuclein activates microglia, which then engulf the α -synuclein and clear it through selective autophagy. It was shown that this process is dependent on microglial TLR4 due to its role in upregulating the autophagy receptor p62, a receptor key to degradation processes in autophagy (Choi et al., 2020).

Conversely, Tu et al., (2021) showed that microglial autophagic impairment occurred on the activation of TLR4 by α -synuclein, though through a different pathway than that investigated by Choi. In addition, Lee et al., (2019) showed that the activation of TLR4 resulted in the suppression of autophagy-related genes in microglia, however it is worth noting that the TLR4 activation was carried out using LPS and not α -synuclein which may result in the different downstream effects. In this way, TLR4 may be acting in a detrimental manner. This has consistently been seen in toxin-induced models of PD. TLR4 has been implicated in the MPTP model of PD, with MPTP treatment resulting in TLR4 upregulation, and inhibition of TLR4, either by gene knockout or with siRNA silencing, ameliorating MPTP induced deficits. For instance, TLR4^{-/-} mice appeared to be less vulnerable to the toxin as they showed less dopaminergic cell death, in addition to fewer activated nigral microglia (Conte et al., 2017; Mariucci et al., 2018; Noelker et al., 2013; Zhao et al., 2016). Campolo et al., (2019) added to this data by showing that the absence of TLR4 also leads to inhibition of inflammasome activation and an improvement in MPTP-induced motor impairments. Zhou et al., (2016) using an in vitro TLR4 siRNA silencing approach found a reduction in microglial activation after MPTP treatment and decreased NF-kB activation. Upregulation of TLR4 was observed in an additional toxin model induced by 6-OHDA; in this model TLR3 expression was also upregulated (McCabe et al., 2017). In a rotenone-induced model, motor deficits and dopaminergic cell loss were ameliorated in TLR4^{-/-} animals (Perez-Pardo et al., 2019). The paraquat toxin-induced production of proinflammatory cytokines by BV2 microglia was reduced on TLR4 knockdown (Huang et al., 2020).

Recent work suggests that not only TLR2 but also TLR4 is necessary for the recognition of α -synuclein and initiation of inflammatory responses. Upon α -synuclein stimulation, TLR4-deficient microglia produced significantly less TNF-α and reactive oxygen species compared to wild-type microglia (Fellner et al., 2013). In agreement with this, Shao et al., (2018) recently showed that either pharmacological inhibition or gene depletion of TLR4 in microglia *in vitro* attenuates α -synuclein-induced TNF- α secretion. Further work showed TLR4 to be involved in mediating the astrocyte response to α -synuclein as well, resulting in upregulation of pro-inflammatory cytokines through activation of the NF-kB pathway (Rannikko et al., 2015). This data suggests that TLR4 is responsible for detecting α -synuclein as a DAMP and initiating an immune response. Work by Hughes et al., (2019) revealed that physiological levels of soluble oligomeric α-synuclein (but not monomers or fibrils) could induce TLR4-mediated cytokine production in microglia in vitro, with a 10- to 100-fold reduction in TNF-α production induced by α-synuclein in TLR4^{-/-} microglia compared to wildtype. The reduction in the inflammatory response to α -synuclein in TLR2^{-/-} microglia compared to wild type was much more modest, suggesting that the TLR4 pathway is the primary mechanism underlying the microglial response to oligometric α -synuclein at physiological concentrations.

Finally, recent evidence implicates TLR4 activation in the gastrointestinal tract with PD pathogenesis. It has been widely proposed that α -synuclein pathology in PD may be initiated in the periphery, specifically in the intestinal and/or olfactory mucosa (Hawkes et al., 2007), and α -synuclein aggregates are present in human colonic mucosa even in very early disease (Malek et al., 2014). In further support of early gut involvement in PD, patients have been found to have an altered microbiome in the intestine (Keshavarzian et al., 2015; Scheperjans et al., 2015). Furthermore, studies in germ-free α -synuclein-overexpressing mice have demonstrated that colonisation of the gut with microbiota from PD patients exacerbates disease severity, compared with colonisation with microbiota from healthy human donors (Sampson et al., 2016). As such, the microbiome itself may play a role in driving disease progression by triggering an abnormal inflammatory response. TLR4 is primarily activated by

Chapter 1

lipopolysaccharides found on the surface of Gram-negative bacteria (Poltorak et al., 1998) and is therefore uniquely poised to sense and be activated by dysbiotic gut microbiota. In line with this theory, Perez-Pardo et al., (2019) have demonstrated increased expression of TLR4 mRNA in human intestinal mucosa biopsy samples from PD patients compared to controls, with a concomitant increase in the expression of pro-inflammatory cytokines (IL-1 β , IFN- γ) and chemokines (CCL2, CCL5) in the sigmoid mucosa. In a second set of experiments using the rotenone mouse model of PD, the same authors showed that TLR4^{-/-} mice were protected to some extent from the toxic effects of rotenone both in the gut and the brain, and their motor deficit was significantly less severe compared to wild-type rotenone-treated mice. However, in a different study it was found that the knockout of MyD88 had no effect on the MPTP treatment-induced degeneration of the dopaminergic nigrostriatal pathway (Drouin-Ouellet et al., 2011) despite the increased survival of myenteric dopaminergic neurons seen in the same MPTP-treated MyD88^{-/-} mice (Côté et al., 2011). In another animal study, transplantation of faecal microbiota from wild-type to MPTP mice led to reduced activation of the TLR4 pathway, dopaminergic neuron rescue, and improved motor function (Sun et al., 2018).

In summary, there is evidence to suggest that TLR4 may have both neuroprotective and detrimental roles in PD, although this may depend on the model system, or on the timepoint in the disease process. Specifically, in an acute setting or in early disease, TLR4 activation by α -synuclein may promote microglial phagocytosis to facilitate intracellular degradation of aggregated α -synuclein. Conversely in the setting of chronic inflammation, in either the gut or brain, TLR4 may play a critical role in escalating the inflammatory response (Fellner et al., 2013; Lee et al., 2008).

References	Study design	Key observation
Shao et al., 2018b; Fellner et al., 2013	α-Synuclein-treated WT or TLR4 ^{-/-} mouse microglia	α-Synuclein triggers microglial activation through TLR4.
Rannikko et al., 2015	α-Synuclein-treated WT or TLR4 ^{-/-} mouse astrocytes	α-Synuclein triggers astrocytic activation through TLR4.
Stefanova et al., 2011	BV2 microglia with anti-TLR4 antibody	TLR4 inhibition impedes microglial α -synuclein phagocytosis.
Zhou et al., 2016	TLR4 knockdown in MPTP-treated BV2 microglia	TLR4 silencing reduces MPTP-induced microglial activation.
Chung et al., 2022	SHSY5Y cells co-cultured with LPS-activated microglia treated with TLR4 inhibitor	Inhibition of TLR4 mitigates α -synuclein aggregation and improves autophagic flux activity.
Tu et al., 2021	Primary microglia from WT and TLR4 ^{-/-} mice stimulated with human α -synuclein	A-synuclein-induced microglial autophagy inhibition is TLR4-dependent.
Mariucci et al., 2018	WT & TLR4 ^{-/-} mice	$\alpha\text{-}Synuclein expression is upregulated in the brain of TLR4 ^{\prime\text{-}} mice.$
Venezia et al., 2021	WT & TLR4-/- mice with stereotaxic injection of PFFs	TLR4 deficiency facilitates spreading of α -synuclein, reduces lysosomal function of microglia, and accelerates neurodegeneration.
Mariucci et al., 2018; Conte et al., 2017; Zhao et al., 2016; Noelker et al., 2013; Shao et al., 2019	WT & TLR4 ^{-/-} MPTP mice	MPTP treatment upregulates TLR4. TLR4 ablation rescues MPTP-induced neuroinflammation, neurotoxicity and motor deficits.
Campolo et al., 2019	WT & TLR4-/- MPTP mice	TLR4 ablation blocks inflammasome activation.
McCabe et al., 2017	WT & 6-OHDA treated rats	On 6-OHDA treatment, striatal TLR4 expression is increased.
Perez-Pardo et al., 2018	WT & TLR4-/- rotenone-treated mice	TLR4 knockout partially rescues dopaminergic cell loss in the substantia nigra and ameliorates the motor deficits.
Huang et al., 2020	BV2 and TLR4-/- microglia treated with paraquat	TLR4 knockout reduced paraquat-induced production of pro-inflammatory cytokines.
Choi et al., 2020	Mice injected with AAV9-human α -synuclein	Neuronal α -synuclein activates microglia which engulf α -synuclein in a TLR4-dependent process.

Table 1.4 Summary of studies evaluating TLR2 in in vitro and in vivo PD models. (Modified from Tables in Kouli, Horne et al., 2019).

PD: Parkinson's disease; WT: wild-type; SHSY5Y: human neuroblastoma cell line; BV2: mouse microglia cell line; THP1 cells: human monocyte cell line

1.4.3 Potential therapeutic strategies

Immunomodulatory drugs already exist for a multitude of conditions, however so far none are regularly prescribed to patients with PD. Currently clinical trials of immunomodulatory agents are underway in PD – for example the AZA-PD trial (Greenland et al., 2020) – but are using broad spectrum agents with the intention of gaining proof of concept. This means that off-target immunosuppressive effects may be problematic. For increased efficiency and tolerability, agents focusing on specific immune pathways have been suggested. Establishing the drug targets requires further exploration of the fundamental processes underlying the role of inflammation in Parkinson's progression. There is strong evidence for a role of Toll-like receptors in Parkinson's disease, as discussed above, and as such these may provide a specific therapeutic target.

1.4.3.1 Candesartan cilexetil

A potential candidate of interest is Candesartan cilexetil. This is the prodrug of candesartan, and is rapidly converted to candesartan during the process of gastrointestinal absorption. Candesartan specifically and potently inhibits angiotensin II type I (AT₁) receptors. AT₁ receptors are found in vascular smooth muscle, in addition to main organs and glands including the heart and the brain. Angiotensin II works through its binding to AT₁ receptors, and through this action has effects including vasoconstriction and increased sodium retention. As a result of its antagonism of AT₁ receptors and the subsequent downstream effects, candesartan is prescribed for the treatment of hypertension and causes a dose-dependent decrease in diastolic blood pressure in response to angiotensin II (Gleiter et al., 2004). Candesartan is well tolerated and has limited side effects.

Candesartan has recently been shown to suppress TLR expression, specifically TLR2 and TLR4 (Dasu et al., 2009). This was demonstrated both *in vitro* in primary human monocytes and *in vivo* in mice peritoneal macrophages. In this study it was shown that candesartan treatment reduced the expression of TLR2 and TLR4, inhibited the secretion of pro-inflammatory mediators including TNF- α , IL-1 β , and IL-6, and was protective against challenge by LPS. Qie et al., (2020) similarly showed a protective effect against LPS challenge whereby candesartan reduced the BV2 microglial activation phenotype, including a reduction in pro-inflammatory cytokines. This was shown to occur in a TLR4-dependent manner. Another angiotensin II receptor blocker (telmisartan) also induced a reduction in TLR2 expression in LPS-stimulated BV2 microglia (Elkahloun et al., 2019). A different study using human renal tubular epithelial cells found that candesartan reversed the LPS-induced upregulation of TLR4 expression and inhibited downstream inflammatory effects. Interestingly, this was not altered by knockdown of AT₁ receptors, thus suggesting that the anti-inflammatory effects of candesartan seen in this study could be through a novel pathway independent of AT_1 receptors (Zhao et al., 2013). Furthermore, it has been shown that candesartan can act to rescue microglial phenotype and function after *in vitro* stimulation with a-synuclein, providing direct evidence for a potential therapeutic role in preventing the asynuclein-induced inflammation in PD (Daniele et al., 2015). Additionally, candesartan can reduce TLR 2 and 4 in the brain *in vivo*. A study investigating the use of candesartan to treat experimentally-induced ischemic stroke in mice showed that orally-dosed candesartan reduced the insult-induced rise in TLR2 and TLR4 expression in the brain and the subsequent rise in pro-inflammatory factors (Barakat et al., 2014). This study is not directly applicable to PD, but does show that candesartan appears to be able to cross the blood-brain barrier and to result in anti-inflammatory effects in the brain. This study also shows that candesartan is having this effect independent of changes in blood pressure as a dose of candesartan was chosen specifically to be under this threshold; the study focuses solely on the effect of candesartan on the TLR signalling cascade. Candesartan has been similarly shown to have a neuroprotective effect alongside its anti-inflammatory action in response to α-synucleinopathy. (Rodriguez-Perez et al., 2018). This was carried out using an α -synuclein-overexpressing animal model. Rats injected with an AAV9 virus expressing α -synuclein were dosed daily with candesartan or telmisartan or vehicle. Animals treated with candesartan or telmisartan showed less dopaminergic cell loss in the substantia nigra, reduced microglial activation, and reduced release of pro-inflammatory factors than the untreated AAV9-injected animals. The role of TLRs was not investigated in this work however, so the mechanistic basis of the reduction in neurodegeneration and neuroinflammation was not clear.

Candesartan has been considered as a potential candidate for the treatment of neurodegenerative diseases, such as Alzheimer's, though so far these studies have focused on its role as an anti-hypertensive and have not investigated inflammation. A double-blind randomized placebo-controlled clinical trial has been carried out to assess the safety of treatment with candesartan for one year in non-hypertensive Alzheimer's patients. The effects of treatment were assessed using biomarkers and clinical measures of Alzheimer's disease severity. Candesartan was found to be safe for treatment over this time, and crucially did not

Chapter 1

result in hypotension. It was also found that candesartan treatment was associated with reduced amyloid accumulation (as assessed using CSF biomarkers), with an increase in functional network connectivity in subcortical networks, and with improved executive function. There was no proof of target engagement in this trial. (Hajjar et al., 2022; Okafor et al., 2021). A recent randomized double-blind placebo-controlled clinical trial (parallel phase II) using candesartan to treat cognitive impairment in 14 people with PD did not show positive effects on cognitive outcome over 28 weeks (the main trial outcome), or on motor symptoms (Kulisevsky et al., 2023). The authors discussed that this trial may not have been long enough to show clinical effect. Candesartan was chosen for this trial due to its function as an AT_1 receptor blocker through which it has a protective effect on oxidative stress and inflammation, however the effects of candesartan treatment on these factors were not explored, and so no conclusions can be based on this.

Many of these studies provide evidence to support further investigation of candesartan as a candidate treatment for PD. However the mechanism of action and whether this is through its TLR-blocking actions needs to be determined.

1.5 Summary and hypothesis

Parkinson's disease is a common neurodegenerative disorder. Patients experience characteristic motor symptoms in addition to other non-motor features including sleep disorders and gastrointestinal dysfunction; the disease commonly results in the eventual development of dementia. Patients experience a substantial decrease in quality of life as the disease progresses. There is currently no treatment to slow the progression of PD and existing therapies work only to alleviate symptoms. A deeper understanding of the mechanisms underlying the disease may be key to the development of treatments.

There is strong evidence for neuroinflammation playing a crucial role in the development and progression of PD. This evidence arises from animal models, *in vitro* work, and post-mortem and neuroimaging studies in humans. Further evidence implicates TLRs in mediating this inflammation through their action as pattern recognition receptors. TLRs have the ability to recognise damage-associated molecular patterns such as aggregated α -synuclein. TLR activation results in the release of proinflammatory factors which may go on to activate other immune cells and eventually to create an environment detrimental to the survival of neurons.

I hypothesise that Toll-like receptors play an important role in the development and progression of PD through mediating the response of microglia and peripheral monocytes to the binding of aberrant forms of α -synuclein, leading to neuroinflammation and exacerbation of neurodegeneration. I propose that TLR blockade will have a protective effect on the development of pathology in an α -synuclein animal model (*Chapter 2*) through its suppression of inflammation (*Chapter 3*) in addition to reducing the inflammatory response of patient-derived immune cells stimulated with α -synuclein (*Chapter 4*). This has important implications for potential new therapeutic routes for the treatment of Parkinson's disease.

1.6 Objectives

- To reproduce the α-synuclein PFF model developed by Kuan et al., (2019) and to use this model to determine the effects of the inhibition of TLRs 2 and 4 on the progression of pathology. TLR inhibition will be carried out using candesartan, a licensed drug for the treatment of hypertension repurposed in this study.
- To determine whether any protective action of candesartan seen in the model is occurring through the suppression of TLR expression and subsequent suppression of inflammation.
- 3. To optimise a protocol to be used in the investigation of the α -synuclein-induced inflammatory response, and for testing whether candesartan and other TLR-blockers may be protective against this inflammation.

Chapter 2.

INVESTIGATING HOW THE USE OF CANDESARTAN AFFECTS THE PROGRESSION OF AN ANIMAL MODEL OF A-SYNUCLEINOPATHY

DECLARATION

The preparation of α -synuclein pre-formed fibrils and their intravenous injection into the rats was carried out by Dr Wei-Li Kuan. Drug dosing was carried out by myself, Sarah Fox, and staff at The Mira Building animal facility (University of Cambridge Biomedical Services). Blood samples were taken by staff at The Mira Building. Animal culling and tissue collection was carried out with the assistance of Dr Wei-Li Kuan and Pam Tyers.

Immunohistochemistry slide scanning was carried out by the Histopathology and ISH core facility at CRUK Cambridge.

2.1 Aims of Chapter

Though many studies have investigated the mechanistic basis of PD using animal models, few of these models accurately recapitulate the early phase of PD. The PFF:RVG9R method of transporting aggregated α -synuclein from the periphery into the brain results in a slowly progressing degenerative model of PD featuring behavioural deficits representative of early/prodromal PD. The nature of this model also allows the exploration of how a drug intervention may slow the progression of pathology in the model, and allows for the testing of drug efficacy over a long period of time in addition to the testing of drug tolerability.

In this chapter I aimed to reproduce the α -synuclein PFF model developed by Kuan et al., and use this model to determine the effect of candesartan, a putative TLR2 and TLR4 antagonist, on the progression of pathology and of behavioural deficits.

2.1.1 Objectives

- 1. To confirm that PFF:RVG9R peripheral injections lead to α -synuclein aggregation in the brain.
- 2. To determine the effect of candesartan treatment on behavioural deficits in the PFF:RVG9R model.
- 3. To determine the effect of candesartan on neuronal loss in the PFF:RVG9R model.

A further objective of the experiment was to investigate whether candesartan has an effect on neurodegeneration, however this will be described in *Chapter 3*.

2.2 Introduction

2.2.1 Animal models of Parkinson's disease

The key clinical and pathological characteristics of PD, as discussed in *Chapter 1*, need to be recapitulated by an animal model to allow mechanistic study of disease pathogenesis. Animal models of Parkinson's disease are useful in researching organisms as a whole. They are a helpful way to monitor the progression of disease pathology long-term, and allow the use of modifying therapies whereby the effects of this intervention can be monitored in a whole system and measured over time. In spite of the usefulness of animal models, the vast majority of drugs successful in slowing the progression of pathology in an animal model eventually fail when transitioned to a clinical trial. This failure is in part due to the lack of success in recapitulating features of the human disease in animals. This is obviously impossible to do fully, however an improvement in similarity of models to disease is needed for higher success rates in the transfer of drugs to clinical trial. The accuracy with which an animal model follows the pathophysiology of the human disease is key to the applicability of the results of work using the model. Many animal models of PD exist; these replicate PD to different extents and emphasise different features of PD.

2.2.1.1 Toxin models

Toxin models were used as the first models of PD, discovered through the observation that the contamination of an illicit drug with MPTP resulted in parkinsonian symptoms in users (Langston et al., 1983). MPTP is one of the most popular toxin models having been found to produce a reliable dopaminergic lesion in mice and monkeys (Visanji and Brotchie, 2005), and showing reproducible behavioural deficits including akinesia and rigidity (Sedelis et al., 2001). On administration through either IV injection or infusion (Jackson-Lewis and Przedborski, 2007), MPTP crosses the blood-brain barrier and is converted to MPP⁺ by monoamine oxidase B. MPP⁺ is transported into dopaminergic neurons through the DAT transporters. MPP⁺ acts in a toxic manner through blocking mitochondrial complex I activity (Nicklas et al., 1987). Another popular model is the 6-hydroxydopamine (6-OHDA) model produced through the injection of the toxin directly into the nigrostriatal system, commonly into the substantia nigra, the medial forebrain bundle, or the caudate-putamen complex (Deumens et al., 2002). This neurotoxin is transported into dopaminergic neurons and results in the degeneration of nerve terminals. Its toxicity occurs through its inhibition of mitochondrial complexes I and IV (Deumens et al., 2002; Glinka et al., 1997). Other neurotoxins used in the production of PD models include rotenone and paraquat. Both of these toxins similarly work to induce oxidative stress in dopaminergic neurons though mitochondrial-related mechanisms (Betarbet et al., 2000; Fei et al., 2008). These models are effective at replicating the vast loss of dopaminergic neurons in the substantia nigra as is seen in the human disease and the resulting motor deficits. However, these models do not produce the characteristic Lewy bodies seen in human PD and thus appear to reproduce the pathological end-point of PD – i.e. the death of dopaminergic neurons – but not the processes involved to get there.

2.2.1.2 A-synuclein overexpression models

The α -synuclein overexpressing animal model is now frequently utilised. Overexpression is induced either by transgenics – commonly through α -synuclein overexpression under the Thy1 promotor; most common in mice - or through a viral vector generally using the stereotactic injection of lentiviruses or adeno-associated viruses into the substantia nigra; most common in rats (Chesselet, 2008). These models were produced after the discovery that some familial forms of PD show a duplication or triplication mutation in the SNCA gene coding for α -synuclein. (Miller et al., 2004; Singleton et al., 2003). Fuchs et al., (2007) showed that a gene dosage effect was present, in that a duplication of the gene resulted in late-onset PD, whereas a triplication resulted in early-onset and dementia. This indicated that the level of overexpression of α -synuclein in the brain is key to the development of the disease. Different iterations of these overexpression models show different effects, these include synucleinopathy, neurodegeneration, and motor deficits (Decressac et al., 2012; Kirik et al., 2003, 2002; Lo Bianco et al., 2002). These models are effective in regards to these pathologies, however, they are based on the rare familial forms of PD not the more common idiopathic PD. Consequently, it is difficult to know how relevant these models are due to α synuclein being expressed at supra-physiological levels. These models are exposed to asynuclein at a much higher level than would be seen in idiopathic Parkinson's where there are differing reports of whether α -synuclein gene expression is elevated or not (Gründemann et al., 2008; Kingsbury et al., 2004; Neystat et al., 1999; Su et al., 2017; Tan et al., 2005). This not only fails to recapitulate the brain environment experienced by people with idiopathic PD, but also results in excessive neuroinflammation which may mask other more subtle neuroinflammatory effects seen in the idiopathic PD brain.

2.2.1.3 Pre-formed fibril models

A-synuclein pre-formed fibrils (PFFs) are artificially-produced aggregated α synuclein species. PFFs are able to act as a template for the recruitment of soluble endogenous α -synuclein into insoluble Lewy body-like inclusions (Volpicelli-Daley et al., 2011). This model was first developed by Volpicelli-Daley in mouse primary neuron cultures. They showed that PFFs were able to act as a template for the recruitment of soluble endogenous α -synuclein into insoluble Lewy body-like inclusions. This model was revolutionary as it uses endogenous levels of α -synuclein, and so produces an environment where physiological levels of α synuclein are aggregating to form pathological inclusions. The toxicity of these injections of α -synuclein PFFs does not arise from elevated levels of α -synuclein, but instead from the PFFinduced recruitment of endogenous α -synuclein into intracellular aggregates. This was proven through the use of α -syn^{-/-} neurons whereby the addition of α -synuclein pre-formed fibrils did not induce neuron death (Volpicelli-Daley et al., 2011). This model has been extended to the production of animal models, including mice, rats, and primates (Abdelmotilib et al., 2017; Duffy et al., 2018; Luk et al., 2012; Paumier et al., 2015; Shimozawa et al., 2017).

The use of PFFs in the production of animal models generally requires injection into the striatum; this then causes the development of Lewy-like pathology in areas which innervate the striatum including, crucially, the substantia nigra. However, one big limitation of this is that the location of these inclusions is injection site-specific. In addition, it has been suggested that PD begins in the periphery (in the gut or olfactory bulb) and then spreads to the brain where degeneration of dopaminergic neurons leads to subsequent motor symptoms (Hawkes et al., 2007). This means that models created by injection directly into the brain may be less relevant to the human disease. Furthermore, injection into the brain can cause site-specific trauma from the needle resulting in effects, such as inflammation, that cannot be entirely separated from this trauma.

2.2.1.4 A novel PFF model

To circumvent the limitations with injecting directly into the brain, as discussed above, a novel model was developed by Kuan et al., (2019), which is produced through the injection of α -synuclein PFFs into the tail vein of Sprague Dawley rats. PFFs on their own have

Chapter 2

been shown to have the ability to cross the blood-brain barrier, however not in sufficient volumes to trigger inclusions or neurodegeneration. Kuan therefore utilised the pre-existing transport system of conjugation to the rabies glycoprotein RVG9R which had previously been utilised to transport RNA into the brain (Kuan et al., 2011; Kumar et al., 2007). This ligand is thought to work through binding a neural-specific acetylcholine receptor which results in a transient membrane inversion, thus allowing the passage of the ligand and its cargo into the brain (Zeller et al., 2015). Kuan showed that this RVG ligand is able to reversibly complex to PFFs, and that they are transvascularly delivered to the brain without affecting their conformation. This model therefore provides an effective periphery to central nervous system transport of α-synuclein PFFs. This replicates processes that are thought to occur in human PD, though in this model the α -synuclein enters the brain through the transport in blood and crossing of the blood-brain barrier, not through neuronal connections with the gut or olfactory bulb, thus making it a less effective model than one incorporating these conditions. Kuan then went on to show that once in the brain, and over a 6 month time period, the PFFs induced site-specific and time-dependent α -synucleinopathy, behavioural deficits associated with early PD – including olfactory deficits (Kuan et al., 2019) and deficits in fine motor ability (Kuan et al., 2022) - and neurodegeneration of both dopaminergic neurons in the substantia nigra and cholinergic neurons in the vagus nerve nucleus. Hence this was shown to be an effective model and potentially better replicates the early stages of human PD as pathology develops relatively slowly with subtle effects even after 6 months.

2.2.2 Candesartan

Candesartan is a licensed drug used in the treatment of hypertension through its actions as an angiotensin II type I receptor blocker. It has recently been shown to act as a blocker of Toll-like receptors 2 and 4 (TLR2, TLR4) *in vitro* and *in vivo*, though the mechanism of this has not been determined (Dasu et al., 2009). Given its ability to act as a TLR blocker and hence potentially reduce neuroinflammation candesartan has been suggested as a candidate for further exploration as a disease modifying treatment for PD. The role of TLRs in PD has been discussed in *Chapter 1.4*, and will be further discussed in *Chapter 3*.

2.3 Methods

2.3.1 Study design

Adult male Sprague Dawley rats (aged 14 weeks at start of experiments, N=72) were used in this study. Baseline behavioural tests were carried out and a blood sample obtained from each of the rats. The rats were split into 3 groups: a control group injected with saline only (N=12), a candesartan-treated pre-formed fibril-injected (PFF-injected) group (N=30), and an untreated PFF-injected group (N=30). All of these groups were given an injection at Day 0, with the candesartan treatment being started at Day -1. The PFF injection was a one-off intravenous injection of pre-formed fibrils conjugated to a RVG9R peptide which allowed the transvascular delivery of the fibrils to the brain. This has been shown to lead to a 20% loss of dopaminergic neurons in the substantia nigra and a 20% loss of cholinergic neurons in the vagus nerve nucleus at 6 months, as well as aggregation of α -synuclein in both of these brain regions (Kuan et al., 2019). Behavioural tests and blood sampling were repeated at 2, 4, and 6 month timepoints post-injection. A subset of rats were culled for post-mortem analysis at each 2 month timepoint: 3 from the control group, and 6 from each PFF-injected groups at the 2 and 4 month timepoints. The remaining 6 from the control group and 18 from each PFF-injected group were culled at the final 6 month timepoint. (Figure 2.1). At 2 months some animals were chosen for culling if they refused to voluntarily take the drug or control treatment (thus making daily administration more difficult), at 4 months the animals were chosen for culling on a random cage-by-cage basis.



Figure 2.1 Timeline of study design.

A parallel experiment in a larger group of control (non-PFF injected) rats was carried out at a later timepoint to establish whether the effects of the drug were disease model-specific. Adult male Sprague Dawley rats (aged 8 weeks at start of experiments, N=30) were used in this parallel study. Baseline behavioural tests were carried out and a blood sample obtained from each of the rats. The rats were split into 2 groups: a candesartan-treated group (N=16) and an untreated group (N=14). Candesartan treatment was started at Day -1, and a saline injection mimicking the PFF injection of the parallel study was carried out at Day 0. Behavioural tests and blood sampling were repeated at 2, 4, and 6 months post-injection.

2.3.2 Animals, PFF-injection, and candesartan treatment

The rats were housed (3 per cage in the initial experiment and 2 per cage in the second control experiment) in standard laboratory conditions: 12-hour light and dark cycle, 22°C room temperature, 55% relative humidity, food and water ad libitum. When required, food restriction was carried out prior to behavioural tests. All food was removed at 16:00 PM the day prior to the tests, and the tests were carried out between 08:00 AM and 13:00 PM with food being replaced by 14:00 PM. Animals were anaesthetised using isoflurane via inhalation prior to injection of α -synuclein pre-formed fibrils (see below for preparation) into the tail veins of the rats in the initial experiment, and saline was injected in the same way into the rats in the second parallel experiment. This injection was carried out by a researcher who was licensed and competent in this technique (Dr William Kuan), not by me. Candesartan treatment was delivered to the rats in the treated groups on a daily basis at 10 mg/kg/day. The dose was selected based on previous studies using Sprague Dawley rats. In these studies doses between 1 mg/kg/day and 10 mg/kg/day were used, which were well-tolerated, and for up to 40 weeks no severe adverse effects were seen (Fu et al., 2011; Hashikawa-Hobara et al., 2012; Jin et al., 2009; Otsuka et al., 2004). The rats were dosed orally using syringes placed into their mouths, and the candesartan was made palatable to the rats through crushing and mixing with 6ml chocolate spread. Untreated rats were fed with chocolate spread only at the same volume.

2.3.3 Preparation of α-synuclein pre-formed fibrils

Human WT full-length α -synuclein pre-formed fibrils were prepared as previously described in the literature (Volpicelli-Daley et al., 2014). A-synuclein monomers (rPeptide) were resuspended in 10 mM Tris-HCl, 50 mM NaCl, pH 7.6 at a concentration of 5 mg/ml. A 37°C thermomixer was used to shake them for 7 days at 1000 r.p.m., they were then aliquoted

and transferred to a -80°C freezer. Prior to use, the PFFs were thawed and diluted to 0.1 μ g/ μ l and sonicated using a Soniprep 150 Plus ultrasonic disintegrator (MSE) with 60 pulses at 10% power with a pause every 10 pulses to allow the solution to cool.

2.3.4 Animal culling and brain preparation

Animals were culled through overdose of carbon dioxide followed by decapitation. The brains were removed and dissected into each hemisphere. The left hemisphere was fixed immediately in 4% PFA for 24 hours and then transferred to 30% sucrose solution with sodium azide. The fixed brains were stored in the sucrose solution at 4°C until sectioning. The right hemisphere was dissected into 5 brain regions, namely the olfactory bulb, the brainstem, the cerebellum, the cortex, and the midbrain, and then flash frozen. The dissected brain regions were stored at -80°C until use. If on removal the left hemisphere was damaged then the right hemisphere was fixed instead and the damaged left hemisphere dissected. The PFA-fixed brains were cut into 40 μ m sections in the sagittal plane using a microtome. 12 series of consecutive sections were obtained and stored as free-floating sections in TBS-azide at 4°C until use in immunohistochemistry.

2.3.5 Blood samples

Blood was obtained through bleeding the animals from the saphenous vein. This was collected using uncoated tubes. The blood was allowed to clot at room temperature for 15 minutes before being centrifuged for 15 minutes at 2000 rpm. The serum was then collected and stored at -80°C until use. More details will be provided in the next chapter.

2.3.6 Immunoprecipitation

Immunoprecipitation was carried out to isolate α -synuclein from the frozen rat brains. Flash-frozen rat cortex was homogenised in 1% Triton X-100 buffer supplemented with Complete protease inhibitors, and a BCA Protein Assay Kit (Thermo Fisher #23227) was used to quantify protein concentration. Lysate containing 1 mg protein in a final volume of 500 ul was spun in an ultracentrifuge at 120000 g and 4°C for 60 min (Beckman Optima Max-XP). The supernatant was collected (triton-soluble fraction). 500 ul Triton X-100 buffer supplemented with Complete protease inhibitors was added to the pellet and a needle used to redistribute the pellet, this formed the triton-insoluble fraction which was then used in the immunoprecipitation. A 5% input was removed from the sample and NuPAGE LDS sample buffer added before being stored at -20°C until use in immunoblot. To the remaining 95% of the sample the IP antibody against α -synuclein (Abcam *ab212184*; 1:100) was added. Samples were then incubated at 4°C overnight on a rotating wheel. DynabeadsTM (Thermofisher Protein G immunoprecipitation kit #10007D) were prepared and added to the samples, before a further incubation for 30 min at 4°C followed by incubation for 10 min at room temperature. The Dynamag-2 magnet was used to isolate the beads; these were washed 3 times in buffer before being eluted using NuPAGE LDS sample buffer. The eluted sample was stored at -20°C with the DynabeadsTM until use in the immunoblot.

2.3.6.1 Immunoblot

Immunoblotting was carried out to establish the presence of aggregated α -synuclein in the rat brains. IP samples were defrosted on ice before being put onto the Dynamag-2 magnet to isolate the beads while the eluted sample was removed. NuPAGE reducing agent was added to both the Input samples and the IP samples before they were denatured at 70°C for 10 minutes. All IP samples were run on one 4-12% polyacrylamide gel; all input samples were run on a second 4-12% polyacrylamide gel. Transfer onto a polyvinylidene fluoride membrane was carried out for 90 min at 30 V. Both membranes were then fixed for 1 hour in 4% PFA according to the known practise when blotting for α -synuclein (Lee & Kamitani, 2011). Each membrane was blocked with 5% milk-TBST and then incubated overnight at 4°C with a second anti- α -synuclein antibody (*BD610787*; 1:1000). The membrane was washed and then incubated with the appropriate HRP-conjugated secondary antibody for 1 hour (Sigma *A9917*; 1:20,000). Immunoreactivity was visualised with chemiluminescence.

2.3.7 Behavioural tests

2.3.7.1 The buried food test

The buried food test was carried out to assess the ability of the rats to discriminate odours. Animals were familiarised with the food pellets (Kelloggs Coco pops) and fasted overnight before this test was carried out. Rats were individually placed in a cage where a single pellet had been buried in the wood shavings lining the bottom of the cage. A timer was started when the animal was placed in the cage and stopped either when the pellet had been found or when 15 minutes had passed. This test was carried out once per 2 month timepoint and the position of the pellet within the cage was changed at each of these timepoints. This test was carried out only in the first study with the PFF-injected animals.

2.3.7.2 The staircase test

The staircase test was carried out to assess the fine motor ability of the rats in the form of reaching and grasping ability. This test was carried out immediately following the odour discrimination test. Animals were food-deprived overnight before the tests. The staircase was made up of 7 steps, with 2 food pellets (Kelloggs Coco pops) placed into wells on the right and left of the step (making 28 pellets total). Rats were individually placed into the staircase box and left for 30 minutes. (**Figure 2.2**). After the test interval, animals were removed from the staircase and the uneaten pellets were counted. This test was carried out once per 2 month timepoint. Animals were discounted from the final analysis if they failed to eat any pellets at all 4 of the study timepoints (baseline, 2 months, 4 months, and 6 months).



Figure 2.2 The staircase test. Equipment used in the staircase test with demonstration of how rat uses the equipment.

2.3.8 Immunohistochemistry

2.3.8.1 Dopaminergic and cholinergic

A 1:12 series of post-mortem brain sections was immunostained for either tyrosine hydroxylase or choline acetyltransferase to stain dopaminergic or cholinergic neurons respectively. Sections were first quenched with 10% methanol and 10% hydrogen peroxide before being blocked in TBS with 0.3% Triton X-100 and 5% serum for 1 hour. Sections were then incubated overnight at 4°C with primary antibodies diluted in TBS with 3% Triton X-100 and 1% serum, as listed in **Table 2.1**, followed by a 3 hour incubation with biotinylated secondary antibody, and 1 hour 30 minute incubation with avidin-biotin complex (Vector; *#PK-6100*). Antigen visualisation was carried out using 3,3'-Diaminobenzidine (DAB

solution; Vector; *#SK-4100*). DAB solution was left on the tissue from 1 minute to 4 minutes, dependent on tissue colour change. Sections were mounted on coverslips and dehydrated in sequential transfers in 70% EtOH, 90% EtOH, and 10 0% EtOH, followed by immersion in 100% xylene for 30 minutes. Finally, they were mounted with DPX mounting medium (Thermo Scienific).

Marker	Species	Dilution	Manufacturer/Catalogue No.
Tyrosine hydroxylase (TH)	Mouse monoclonal	1:500	Millipore Sigma #MAB318
Choline acetyltransferase (ChAT)	Goat polyclonal	1:500	Millipore Sigma AB144P

Table 2.1 Antibodies used for immunohistochemistry on brain sections.

Quantification of tyrosine hydroxylase (TH) and choline acetyltransferase (ChAT) staining was carried out in a similar fashion. To ensure non-bias the animal code and group number was blinded. Using a Leica DM6000 microscope with a fixed digital camera, and using the Stereo Investigator 9.10.3 software (MBF Bioscience), the boundaries of the regions of interest (ROI) were drawn. The ROI for TH staining was the substantia nigra, and the ROI for ChAT staining was the hypoglossal nerve nucleus and the vagus nerve nucleus. The identification of these regions was carried out with the assistance of the Paxinos rat brain atlas (5th edition). Neurons were counted using unbiased stereology.

2.3.8.2 A-synuclein

A 1:12 series of post-mortem brain setions were immunostained for a selection of α synuclein antibodies. Sections were first mounted on SuperFrost Plus Adhesion slides (Epredia #10149870) and left to dry at 37°C for 4 hours. Antigen retrieval was carried out using sodium citrate buffer for 20 minutes in the microwave. Sections were then quenched with 20% methanol and 3% H₂O₂ for 30 minutes before being blocked in TBS with 0.3% Triton X-100 and 5% serum for 1 hour. Sections were then incubated overnight at 4°C with primary antibodies diluted in TBS with 0.3% Triton X-100 and 1% serum, as listed in **Table 2.2**, followed by a 2 hour incubation with biotinylated secondary antibody, and 55 minute incubation with avidin-biotin complex (Vector; #*PK-6100*). Antigen visualisation was carried out using 3,3'-Diaminobenzidine (DAB solution; Vector; #*SK-4100*). DAB solution was left on the tissue from 1 minute to 4 minutes, dependent on antibody used. Counterstaining was carried out using haematoxylin. Sections were dehydrated in sequential transfers in 70% EtOH, 90% EtOH, and 100% EtOH, followed by immersion in 100% xylene for 30 minutes. Finally, they were mounted with DPX mounting medium (Thermo Scienific).

Table 2.2 Antibodies used for a-synuclein immunohistochemistry on brain sections

Marker	Species	Dilution	Manufacturer/Catalogue No.
A-synuclein-42	Mouse monoclonal	1:500	BD Biosciences #610787
A-synuclein-211	Mouse monoclonal	1:500	Invitrogen MA5-12272
A-synuclein 2A7	Mouse monoclonal	1:500	Antibodies.com A85290

2.3.9 Immunofluorescence

A 1:12 series of post-mortem brain sections was immunostained for tyrosine hydroxylase to identify dopaminergic striatal fibres. Sections were blocked in TBS with 0.3% Triton X-100 and 5% serum for 1 hour before being incubated overnight at 4°C with primary antibodies diluted in TBS with 3% Triton X-100 and 1% serum. Sections were then incubated for 3 hours with DAPI and anti-rabbit-568 at room temperature in the dark. Sections were mounted and then coverslipped using FluorsaveTM reagent (Millipore Sigma).

Quantification of TH striatal immunofluorescence was carried out using the Leica DMI8 microscope. An area of interest was drawn around the striatum and the tile scanning function was used to form a single image of the entire striatum. These images were analysed using ImageJ and a mean greyscale was calculated for each region; all sub-sections containing striatum were used to produce a mean striatal intensity through the whole striatum.

2.3.10 Statistical analysis

Data analysis and graph generation was carried out using GraphPad Prism 8.3. Data normality was assessed using the Shapiro-Wilk test. Pairwise comparisons were carried out between untreated and candesartan-treated PFF or control groups using an unpaired T-test or a Mann-Whitney test (for parametric or non-parametric data respectively). For data comparing >2 groups a one-way ANOVA was carried out followed by Tukey's posthoc test or a Kruskal-Wallis test followed by Dunn's posthoc analysis or a (for parametric or non-parametric data respectively).

2.4 Results

2.4.1 Study design

The model used in this study was developed and established by Dr Wei-Li Kuan. My study focus was to compare disease development and progression in a PFF-injected group treated with the drug candesartan, (which has been shown to block Toll-like receptors 2 and 4), with a PFF-injected group not treated with candesartan. I included a small saline-injected control group in the original study design but limited this to only 6 animals, given that this model was already established and I aimed to reduce the total number of animals used (in accordance with the 3 Rs of animal research). This small control group was included to ensure that the degeneration had occurred as expected, and also because a new behavioural test was carried out which wasn't used in the original paper so controls were required to compare to a non-disease state.

2.4.2 What effect does the injection of α -synuclein PFFs have on endogenous aggregation of α -synuclein within the rat brain?

The original model produced by Kuan et al., (2019) showed aggregation of α synuclein in the brains of rats injected with PFFs conjugated to an RVG9R peptide, but not in the brains of rats injected with unconjugated PFFs or those injected with saline only. We used the conjugated PFF:RVG9R and saline only to inject our animals – no unconjugated PFFs were used in this study. For simplicity, from here on in our PFF:RVG9R-injected animals will be referred to as PFF-injected or PFF-only. To confirm that α -synuclein aggregation in the brain was occurring as expected in my study, sequential ultracentrifugation was carried out to isolate brain homogenate containing only triton-insoluble α -synuclein, followed by immunoprecipitation to confirm the presence of high molecular weight species in the brain. This was only carried out in the cortex due to a lack of sufficient amounts of other brain tissue which is required in large amounts for this experiment. The immunoprecipitation showed the presence of these high molecular weight species in the cortex of PFF-injected animals. High molecular weight species were also unexpectedly seen in cortical samples from the control

animals. (Figure 2.3a). Due to this unforeseen result, it was clear that an error had occurred in the experimental set-up and it seemed likely that the control animals were accidentally injected with α -synuclein PFFs. For this reason, I decided in conjunction with my supervisors to exclude the original controls and conduct a repeat study using control animals. This second study also allowed me to test the effect of candesartan in non-PFF rats to determine whether any effect of the drug on behaviour and pathology was specific to the PFF-model. (Figure 2.4). Though every effort was made to directly replicate the first study in the PFF-injected animals in order to allow for comparison, due to some specific unavoidable differences – namely the age of the rats, the number of rats housed per cage, and the timing of the post-mortem work carried out – it was decided that this second control study would be carried out as a parallel study and data analysis would not involve direct comparison between the 2 studies.

Immunoprecipitation was carried out in the new control groups, as in the first study, to confirm the absence of high molecular weight α -synuclein species in the new controls. No high molecular weight species were seen in either of the new control groups (candesartan-treated or untreated), but were present in the PFF-injected animals, confirming the validity of control versus PFF status of the experimental groups. (**Figure 2.3b**).



Figure 2.3 Model characterisation. Immunoprecipitation on homogenised rat cortex in (a) original study with comparison between original control animal and PFF-injected animal, and (b) original and parallel control study with comparison between original control animals, and untreated/drug-treated PFF-injected/new control animals. Loading controls can be seen in the 5% input.





Figure 2.4 PFF and control studies were carried out in parallel. Diagram showing study design of parallel PFF and control studies, and the timeline of these studies.

Further work was carried out to determine whether the α -synuclein could be seen in key pathological brain areas in this model, i.e. the brainstem and substantia nigra, using immunohistochemistry techniques. This was carried out in 4 PFF-injected animals and 4 saline injected control animals (controls from the parallel study) using multiple different α -synuclein antibodies to ensure that any changes would not be missed through the use of an ineffective antibody. Suprisingly, there were no clear differences in levels of α -synuclein staining in the brainstem of PFF-injected and control saline-only injected rats (controls from the parallel study). This was consistently seen on staining brainstems using the α -synuclein-42 (**Figure 2.5**) and α -synuclein-2A7 (**Figure 2.6**) antibodies, however no staining at all was seen in either animal group using the α -synuclein-211 antibody (**Figure 2.7**). Positive staining was identified using the α -synuclein-2A7 and α -synuclein-42 antibodies, however this was difficult to interpret fully in regards to differences between PFF-injected and saline-injected control rats due to high non-specific background staining. No positive staining was seen in the substantia nigra using either the α -synuclein-42 or α -synuclein-2A7 antibodies, though high levels of non-specific background staining was seen. (**Figure 2.8**).



Figure 2.5 α-synuclein-42 staining in the vagus nerve nucleus. Representative images of α-synuclein staining in the vagus nerve nucleus (outlined) in (a) control (red outline) and (b) PFF-only (blue outline) rats at (I) 4X, (II) 40X, and (III) 100X magnification.



Figure 2.6 α-synuclein-2A7 staining in the vagus nerve nucleus. Representative images of α-synuclein staining in the vagus nerve nucleus (outlined) in (a) control (red outline) and (b) PFF-only (blue outline) rats at (I) 4X, (II) 40X, and (III) 100X magnification.



Figure 2.7 α -synuclein-211 staining in the vagus nerve nucleus. Representative images of α -synuclein staining in the vagus nerve nucleus in (a) control and (b) PFF-only rats at (I) 4X and (II) 10X magnification.



Figure 2.8 a-synuclein staining in the substantia nigra. Representative images of (a) α -synuclein 2A7 and (b) α -synuclein -42 staining in the substantia nigra in control (left; red-outline) and PFF-only (right; blue-outline) rats at 4X magnification.
2.4.3 Does candesartan treatment affect the development of olfactory deficits?

Behavioural tests were carried out to assess hyposmia, a clinical feature of PD. The behavioural test chosen was the buried food test whereby animals are tested on their ability to use olfactory cues to forage for a food pellet after overnight fasting. Previously published work using this animal model (Kuan et al., 2019) showed significantly increased retrieval latency – indicative of olfactory deficits - in the PFF-injected animals as compared to the controls. In the current study, there was a trend towards an increase in mean retrieval latency in the untreated PFF-injected animals which was not seen to the same extent in the candesartantreated group, though this did not reach statistical significance (Friedman test; p=0.4612). (Figure 2.9a). Comparison between treatment groups at each individual timepoint did not show any significant differences between the untreated and drug-treated olfactory ability. (Multiple Mann-Whitney tests adjusted for False Discovery Rate). (Figure 2.9b). There appeared to be a greater proportion of animals that failed to find the pellet (as shown by a retrieval latency of 900 seconds which was the cut-off) in the untreated PFF-injected group as compared to the candesartan-treated group, though this does not reach significance (Fisher's exact test p=0.0877). (Figure 2.9b). This test was not repeated in the second control study due to time constraints.



Figure 2.9 Timecourse of olfactory deficits. (a) Retrieval latency over time in the treated and untreated group. Only animals surviving to the end of the study used, to allow for analysis over time. N = 18 per treatment group per timepoint. Friedman test; p=0.4612. (b) Retrieval latency in candesartan treated versus untreated PFF-injected animals at each measurement timepoint. Baseline N=30 per group; 2m N=30 per group; 4m N=24 per group; 6m N=18 per group. Multiple Mann-Whitney tests adjusted for False Discovery Rate; Fisher's exact test comparing number of successes to number of failures; p=0.0877.

2.4.4 Does candesartan treatment protect animals against motor deficits?

Previous work using this model used common motor behavioural tests such as the ladder climbing test and adjusted step test, however these tests did not show significant behavioural deficits (Kuan et al., 2019). I therefore used a theoretically more sensitive test to determine whether fine motor deficits developed in the PFF model over 6 months, and whether this was altered by candesartan treatment. The staircase test was used to determine fine motor ability in the form of reaching and grasping ability. Initial results showed that from the 2 month timepoint there was a significant difference in motor ability between the drug-treated PFF-injected animals compared to the untreated PFF-injected animals with the drug-treated animals having better motor ability (Mann-Whitney test; p<0.05). This significantly better motor ability was sustained at the 4 and 6 month timepoints (Mann-Whitney test; p<0.01). These differences between drug-treated and untreated were not seen in the parallel control study. (**Figure 2.10a**-

b). In addition, there was a trend towards decreased motor ability in the untreated PFF-injected group which was not seen in any other treatment groups. (**Figure 2.10c**). However, I also found that many of those animals unable to retrieve any pellets were heavier than the others. When tested it was found that weight significantly inversely correlated with motor ability in both the PFF-injected and saline-injected studies (simple linear regression; PFF p=0.0014, Control p=0.0363) (**Figure 2.11a**). The untreated PFF-injected animals weighed significantly more than the candesartan-treated animals (Unpaired T-test; p<0.0001) (**Figure 2.11b**). This occurred due to the weight-loss effects of candesartan (Müller-Fielitz et al., 2011). Due to these results, an ANCOVA was carried out to assess whether adding in weight as a covariance would remove the significance seen in the initial analysis. Significance did not withstand the ANCOVA; this indicates that the beneficial effect of candesartan on motor ability might be mediated through its effect on weight rather than a protective effect on the development of pathology in the disease model.



Figure 2.10 Timecourse of motor deficits. Motor ability as determined by number of pellets eaten in untreated vs drug-treated (a) PFF-injected animals and (b) saline injected control animals at each measured timepoint. Mann-Whitney test; $p^*<0.05$; $p^{**}<0.01$. (c) Timecourse of motor ability in each individual treatment group.



Figure 2.11 Weight as a confounding variable. (a) Correlation between animal weight and number of pellets eaten in PFF-injected animals (left)(p=0.0014) and control saline-injected animals (right)(p=0.0363). Simple linear regression. (b) Comparison between animal weights in untreated and drug treated PFF-injected (left) and saline injected control (right) animals. Unpaired T-test; $p^{****} < 0.0001$. Mann-Whitney test.

2.4.5 Does treatment with candesartan affect the survival of cholinergic neurons?

The vagus nerve nucleus is an easily identifiable region of interest within the brainstem containing cholinergic neurons which degenerates early in PD. According to Braak's hypothesis, Parkinson's pathology begins in the brainstem (Braak et al., 2003), and as such the vagus nerve nucleus was chosen as a key region to examine for evidence of neurodegeneration. This analysis was carried out in the majority of animals, though not all due to the lack of useable tissue in a small number of animals. The untreated PFF-injected rats showed significant degeneration of cholinergic neurons in the vagus nerve nucleus over 6 months post PFF-injection (One-way ANOVA followed by Tukey's posthoc test; p<0.05). The candesartantreated PFF-injected rats did not show degeneration over the same period of time, indicative of a protective effect of candesartan. (**Figure 2.12a**). Due to differences in study design it is not

known whether degeneration over time was also seen in either of the saline-injected control groups (control animals were only culled at the 6 month timepoint). Candesartan treatment protected cholinergic neurons in this brain region in the PFF-injected rats as evidenced by a higher density of cholinergic neurons in the candesartan-treated PFF-injected rats at 6 months as compared to the untreated PFF-injected rats (Mann-Whitney test; p<0.01). No such difference was observed between the untreated and candesartan-treated saline-injected rats. (**Figure 2.12b-c**). The degeneration of cholinergic neurons was not found to correlate to behavioural tests (Simple linear regression p=0.8584). (**Figure 2.12d**). This suggests a disease-state specific protective effect. The degeneration of cholinergic neurons in the PFF-injected rats (One-way ANOVA followed by Tukey's posthoc test; p<0.05) and the protective effect of candesartan treatment on cholinergic neuron survival (Unpaired T-test; p<0.01) was similarly seen in the hypoglossal nerve nucleus. (**Figure 2.13**). This region has not been established as being pathologically relevant in human PD but is located directly adjacent to the vagus nerve nucleus.



Figure 2.12 Candesartan treatment has a protective effect on cholinergic neuron survival in the vagus nerve nucleus. (a) Timecourse of cholinergic neuron (ChAT⁺) density in the vagus nerve nucleus of untreated (top; n=3 at 2m, 3 at 4m, 17 at 6m) and drug-treated (bottom; n=4 at 2m, 3 at 4m, 16 at 6m) PFF-injected animals. One-way ANOVA followed by Tukey's posthoc test; $p^*<0.05$. (b) Cholinergic neuron density in the vagus nerve nucleus of untreated and drug-treated PFF-injected animals (left; n=17 and 16 respectively) and saline injected control animals (right; n=14 and 16 respectively). Mann-Whitney test; $p^{**}<0.01$. (c) Representative images of the vagus nerve nucleus stained with anti-ChAT antibody in an untreated PFF-injected animal (top) and a drug-treated animal (bottom). (d) Correlation between cholinergic neuron density and number of pellets eaten indicative of motor ability. Simple linear regression.



Figure 2.13 Candesartan treatment has a protective effect on the survival of cholinergic neurons in the hypoglossal nucleus. (a) Cholinergic neuron density in the hypoglossal nucleus of untreated and drug-treated PFF-injected animals (left; n=16 for each group) and saline-injected control animals (right; n=14 and 15 respectively). Unpaired T-test; $p^{**}<0.01$. (b) Representative images of the hypoglossal nerve nucleus stained with anti-ChAT antibody in an untreated (top) and drug-treated (bottom) PFF-injected animal. (c) Timecourse of cholinergic (ChAT⁺) density in the hypoglossal nerve nucleus of untreated (drug; n=3 at 2m, 2 at 4m, 16 at 6m) and drug-treated (bottom; n=3 at 2m, 4 at 4m, 16 at 6m) PFF-injected animals. One-way ANOVA followed by Tukey's posthoc test; $p^{*<0.05}$.

2.4.6 Does candesartan treatment have an effect on dopaminergic degeneration in the substantia nigra?

The substantia nigra is a key pathological site in the Parkinson's brain, making it an important region to identify the degeneration of dopaminergic neurons. This degeneration was identified in the Kuan et al., (2019) paper, however it was harder to characterise in my study due to the lack of controls to directly compare to. Surprisingly, no degeneration of the

dopaminergic neurons in the substantia nigra was seen over time, and the average number of neurons remained relatively constant regardless of timepoint or treatment group. (Figure 2.14a). Though no significant differences were seen between treatment groups, within these groups there was a large spread of neuron number. (Figure 2.14b-c). I hypothesised that degeneration was only occurring in specific animals. To test this, relationships between dopaminergic neuron number and other pathological and behavioural measures were explored. No correlations were found between the number of dopaminergic neurons and either motor deficits, olfactory dysfunction, or cholinergic neuron density. (Figure 2.14d). This demonstrates that the number of dopaminergic neurons is not indicative of whether other pathology is progressing in the model.



Figure 2.14 Dopaminergic degeneration in the substantia nigra is not occurring in this work despite degeneration being seen in the original Kuan model. (a) Timecourse of number of dopaminergic (TH^+) neurons in the substantia nigra of untreated animals (left; n=6 at 2m, 6 at 4m, 18 at 6m) and drug-treated animals (right; n=6 at 2m, 6 at 4m, 18 at 6m) and drug-treated animals (right; n=6 at 2m, 6 at 4m, 18 at 6m). (b) Number of dopaminergic neurons in the substantia nigra of untreated and drug-treated PFF-injected animals (left; n=18 per group) and saline-injected control animals (right; n=14 and 16 respectively). (c) Representative images of substantia nigra stained with anti-TH antibody in an untreated (top) and drug-treated (bottom) animal. (d) Correlation between (i) number of dopaminergic neurons and motor ability, (ii) cholinergic neuron density, or (iii) olfactory ability. Simple linear regression; p>0.05 for all.

Determining fibre density in the striatum is another way to assess dysfunction in dopaminergic tracts, and may be a better reflection of early dopaminergic dysfunction. Immunofluorescence staining of TH-positive nerve fibres was carried out in the striatum. No significant differences were seen between untreated and candesartan-treated animals in the PFF-injected group which is where the differences were expected. Surprisingly a significant

difference was observed between the untreated and candesartan treated control animals with the drug-treated animals showing higher levels of TH⁺ fibres in the striatum. (**Figure 2.15a**). There was no correlation between striatal density and TH neuron number, so not supporting the theory that the lack of dopaminergic neuron degeneration is due to the degeneration being yet to spread along the dopaminergic tracts. (**Figure 2.15b**). There was also no correlation between striatal fibre density and motor deficits, providing further evidence that the motor effects were not due to a dopaminergic deficit in the PFF-injected animals. (**Figure 2.15c**).



Figure 2.15 No effect of candesartan on striatal dopaminergic fibre density. (a) Striatal fibre density (as determined by mean grey area of IF images) of untreated and drug-treated PFF-injected animals (left) and saline injected control animals (right) Mann-Whitney test; $p^{**}<0.01$. Correlation between striatal density and (b) number of dopaminergic (TH⁺) neurons or (c) number of pellets eaten in staircase behavioural test in PFF-injected (left) or control (right). Simple linear regression; p>0.05 in all. PFF only N=18; PFF+drug N=18; Control only N=14; Control+drug N=16.

2.5 Discussion

2.5.1 Summary

In this chapter I sought to identify how long-term treatment with candesartan affected the development of pathology in a slow-progressing model of α -synucleinopathy. The results suggest that candesartan has a protective effect on the survival of cholinergic neurons in both the vagus nerve nucleus (an area seen to degenerate in human PD) and the hypoglossal nucleus (not seen to degenerate in human PD but found directly adjacent to the vagus nerve nucleus). A protective effect is also seen on motor ability, however this effect could be due to weight differences rather than an improvement in dexterity. Furthermore, a trend towards a candesartan-mediated protective effect on olfactory ability was seen but did not reach significance. There was a lack of dopaminergic degeneration in the substantia nigra, possibly due to the fact that this model represented an early (prodromal) disease stage and so the effect of candesartan on dopaminergic pathology could not be adequately tested. However overall, these results show a trend towards a protective effect of candesartan in specific (but not all) areas of disease progression in this model, particularly on the progression of cholinergic degeneration in the brainstem.

The protective effects of candesartan demonstrated in this study using a peripherallyinitiated animal model of early PD provide a novel angle on the TLR-blockage as a protective agent hypothesis. The model used is subtle, similarly to the early stages of PD, and has meant that the protective effects of candesartan treatment, though present, are also subtle. The most important result from this study is the protective effect of candesartan on the survival of cholinergic neurons in the vagus nerve nucleus. According to Braak's hypothesis neuronal Lewy bodies first appear in the dorsal motor vagal nucleus in the brainstem before spreading to other brain regions including the substantia nigra in the midbrain. I have not been successful in quantifying α -synuclein levels in specific brain regions, however I have identified specific areas of neurodegeneration. The degeneration of cholinergic neurons in the vagus nerve nucleus of the brainstem, according to Braak's hypothesis, could be the first indication of pathology in the rat α -synucleinopathy model. In this way, this model forms a good representation of early PD before neurodegeneration has spread elsewhere in the brain, such as to the substantia nigra. This model also provides compelling evidence for PD to begin peripherally, as the peripheral injection of α -synuclein PFFs results in the initiation of brain degeneration in the same brain region as suggested by Braak.

Nevertheless, although the results from this work alone are not sufficient for the immediate transferral to the human disease through the repurposing of candesartan into a clinical trial, there is sufficient data for making candesartan a clear target for further study.

2.5.2 Lack of dopaminergic degeneration

The most obvious limitation with my study is that although the rats developed cholinergic degeneration as expected, they did not develop dopaminergic degeneration in the substantia nigra. Alongside α -synuclein inclusions, dopaminergic degeneration is the most important aspect of a model of PD; these are the two main pathophysiological characteristics of the disease. There are several possible reasons for the lack of development of a dopaminergic deficit in the model which are discussed below.

2.5.2.1 Slow development of neurodegeneration

As mentioned above, this model is a slowly developing model making it a good model for PD which is a slowly progressing disease. According to Braak's hypothesis, Parkinson's pathology begins in the dorsal motor nucleus of the vagus before spreading to other regions of the brain (Braak et al., 2003). This may explain why neurodegeneration is seen in the brainstem in this study, but not in the substantia nigra – i.e. pathology is yet to spread to the nigra. Dopaminergic degeneration was observed in the Kuan et al.,(2019) study after 6 months, and this continues to develop further post 6 months as was shown in an extended study (Kuan et al., 2022). There are possible sex differences in progression rates in PD, and it is possible that the males rats used in this study showed a slower rate of progression than the females in the Kuan study (Cerri et al., 2019; Iwaki et al., 2021). If left for longer than 6 months, dopaminergic pathology may have developed in the nigra in my study. This feature of the model used in this study may actually be a positive. As discussed in *Chapter 1.1.2*, at the point of diagnosis – generally occurring on the development of motor features – neuron loss in the SNpc is at around 50% (Marsden, 1990; Ross et al., 2004). PD is thought to initiate many years prior to diagnosis, and thus the model used in this study – where α -synuclein pathology is present and degeneration seen in the vagus nerve nucleus but little to no degeneration in the SN – may be a good representation of the earliest disease processes.

2.5.2.2 Model mechanism

As described earlier, the mechanism through which this model is produced utilizes the rabies virus glycoprotein RVG9R peptide to transport the conjugated PFFs transvascularly into the brain. This peptide works through its binding specifically to neural acetylcholine receptors; these receptors are found on cholinergic neurons. As a result, the PFFs may be entering cholinergic neurons through this mechanism, inducing α -synuclein aggregation, and resulting in the cholinergic degeneration. Acetylcholine receptors are not found on dopaminergic neurons. As a result, the RVG9R-conjugated PFFs will not be directly entering the dopaminergic neurons after injection, however they may spread through the brain from their origins in the cholinergic neurons. The underlying mechisms for this spread are a key field of interest in Parkinson's research and theories include their being spread through anatomical circuits (Braak et al., 2003), through α -synuclein acting in a prion-like manner (Olanow and Prusiner, 2009), or through transfer via the actions of glial cells (Xia et al., 2019). The lack of dopaminergic degeneration in my work may be a result of the failure of the PFFs, or the PFFinduced aggregated α-synuclein, spreading to other areas of the brain. The spread of RVGconjugated a-synuclein and of RVG-conjugated GFP wa shown to occur in work by Kuan et al., (2019). In this work it was identified in the cortex, hippocampus, thalamus, and nigra. However, as discussed, there are sufficient differences between the work carried out in this chapter and the work carried out by Kuan that it cannot be assumed that this spread of α synuclein from cholinergic neurons in the brainstem to other neuron types and other areas of the brain is occurring in the animals used in my work. If the spread of α -synuclein has failed then this could explain why degeneration is only seen in cholinergic neurons.

2.5.2.3 Sex differences

The main difference between the study carried out here and the original study carried out by Kuan et al., (2019) was the sex difference between the animals. The original study used female animals whereas we used male animals. Male rats were used in keeping with the majority of animal studies due to concerns about hormonal changes having a neuroprotective effect (Liu and Dluzen, 2007). PD is known to be more prevalent in males as opposed to females with a ratio of approximately 2:1 (Wooten et al., 2004). There is also data showing that the immune response in PD patients may differ between males and females, which – as

suggested by our overall hypothesis – could alter the development of the disease in different sexes (Nissen et al., 2019). It has been suggested that the decrease in Parkinson's prevalence in females might occur due to the protective effect of oestrogen on dopaminergic neuron survival. Evidence for this arises from toxin-induced Parkinsonism where females seem to experience a protective effect on dopaminergic degeneration in response to exposure to various environmental toxins. This is seen in both animal models (including rotenone and paraquat neurotoxin models) and in human epidemiological studies (investigating the link between PD risk and occupation where men and woman hold equivalent occupations) (Adamson et al., 2022; Liu and Dluzen, 2007). However, though evidence strongly supports differences in propensity to develop Parkinson's between males and females, this doesn't help to explain the differences seen between the Kuan study and our study. Based on the above evidence, it would be expected that the male animals used in our study would experience higher levels of dopaminergic degeneration due to a lack of oestrogen, however this is not the case. Consequently, this does not explain the lack of degeneration seen in this study.

2.5.2.4 Weight differences

The method used for dosing the animals was crushing candesartan tablets and mixing them with chocolate spread. The control for this was chocolate spread on its own. Dosing was carried out in this way as a preliminary study showed that when the drug was crushed and mixed with water the rats would not drink the water and so were not getting the correct dose of candesartan. The method of dosing with chocolate spread was very effective in terms of ease of dosing as the rats were very keen to eat it. However, this was ultimately detrimental to the study as after 6 months of eating 0.6 ml of chocolate spread every day the rats gained weight. This was particularly problematic as the weight was not gained equally between the treatment groups – candesartan is known to cause weight loss and it is likely that this counteracted the weight gain from the chocolate spread in these treatment groups (Müller-Fielitz et al., 2011). This created a confounding effect in the motor behaviour test as weight correlated negatively with motor ability. The differences between the candesartan-treated and untreated PFF-injected groups in terms of motor ability no longer reached significance when weight was added as a covariate. Differences in weight and fat content could also be relevant to the lack of dopaminergic degeneration in this study, when compared to Kuan's study.

A link between a high fat diet or increased weight and dopaminergic degeneration has been suggested, however available evidence indicates that high fat/weight is associated with increased dopaminergic degeneration, in contrast to our findings. The phosphatase PP2A is one of the only phosphatases that dephosphorylates tyrosine hydroxylase (the enzyme which converts L-tyrosine to L-Dopa – the first step in the synthesis of dopamine in dopaminergic neurons) (Nagatsu, 1995). Through its dephosphorylation of TH, PP2A acts to reduce TH activity (Zhang et al., 2007). PP2A shows increased expression after exposure to a high fat diet (Jun et al., 2008), hence the high fat diet in our study might be expected to result in increased expression of PP2A, and the subsequent dephosphorylation and decrease in activity of TH. Further evidence for this arises from Supplementary Figure 2.1 where a western blot was carried out to determine the levels of TH in the midbrain. This was carried out in a selection of animals from this study, and a selection of animals from the published Kuan et al., (2019) study. The blot showed that the TH in the animals used in this study had undergone posttranslational modifications – potentially the dephosphorylation discussed above – which have resulted in bands of a lower molecular weight than that seen in the animals from the Kuan et al., (2019) study which did not receive a high fat diet. This could indicate a potential increase in PP2A. This does not explain the lack of degeneration seen over time in our study, as it would be expected that TH levels would be lower in my study compared to the Kuan et al., (2019) study due to the effect of a high fat diet-induced rise in PP2A. However, it could be the case that this high fat diet rapidly effects the downregulation of TH meaning that by the 2 month timepoint TH is already downregulated. Consequently, it might not have been possible to detect further dopaminergic neurodegeneration over time in my study. Unfortunately, due to a lack of any baseline post-mortem rat brains I was unable to confirm this. This may also explain the lack of significant difference in dopaminergic neuron number between the candesartan-treated and untreated animals, as even though a difference in weight was seen, the amount of fat quantity in their diet is the same.

Finally, weight may have an effect on dopaminergic neurodegeneration due to differences in insulin signalling. Insulin is thought to act in a protective manner towards neuron survival. This has been shown to occur in various types of neurons and against various toxic insults (Ribeiro et al., 2014; Sun et al., 2010). Evidence points to this protective effect also occurring in dopaminergic neurons in an insulin dose-dependent manner through reduction of reactive oxygen species, nitric oxide release, and calcium ion influx (Pang et al., 2016; Ramalingam and Kim, 2016). Evidence for the neuroprotective effect further occurs through the link between type-2 diabetes-mediated insulin resistance and the development of PD (Reviewed in Athauda and Foltynie, 2016). Could it be possible that a temporary rise in insulin

on the addition of daily chocolate spread to the diets of the animals used in this study could be having a protective effect on dopaminergic neuron survival in the substantia nigra, and so negating any detrimental effects of PFF-injection? Though possible, this is unlikely. A high fat diet is strongly associated with insulin resistance (Hancock et al., 2008). Various studies suggest this would make a subject more prone to developing PD (Hu et al., 2007; Xu et al., 2011), the progression of disease would be faster (Cereda et al., 2012; Kotagal et al., 2013), and the dopaminergic neuron loss would be greater (Choi et al., 2005; Morris et al., 2011, 2010). This is not what is seen in our study (with a high fat diet) compared to the Kuan study (with a normal diet). In addition, insulin is a very transient hormone, so to measure levels of insulin and assess how this may be linked to the survival of dopaminergic neurons in our model would require more than the post-mortem tissue we have available, so making this unfeasible to test.

2.5.3 Lack of α-synuclein pathology

In work carried out by Kuan et al., (2022) using this RVG9R:PFF model, α -synuclein pathology was suggested to be diffuse throughout the brain. They did not carry out immunohistochemistry identifying any aggregates in this paper, and instead demonstrated the presence high molecular weight α-synuclein using immunoprecipitation. of Immunohistochemistry was used in the earlier Kuan et al., 2019 paper and showed α -synuclein aggregates through pFTAA and Ser-129 co-staining – though this is not conclusive evidence of α -synuclein aggregates as pFTAA recognises any oligometric and fibrillar proteins and is not specific to α -synuclein. I similarly found high molecular weight α -synuclein species in the cortex of my PFF-injected rats using immunoprecipitation. However, when using immunohistochemistry for α -synuclein, there was no clear evidence of α -synuclein aggregates and no differences in α -synuclein staining in the brainstem or substantia nigra between the PFF-injected and saline-injected control animals. The lack of any α -synuclein staining in the substantia nigra in the PFF-injected animals could explain the lack of dopaminergic neurodegeneration seen in the PFF-injected animals, and therefore why I couldn't show a protective effect of candesartan on dopaminergic neurodegeneration (as there is no deficit which candesartan could act upon). The lack of differences in α -synuclein staining between the PFF-injected and control animals in the brainstem counfounds the assumption that the cholinergic degeneration seen in the PFF-injected animals was due to α -synuclein pathology.

However, the α -synuclein staining in the control animals was unexpectantly high, hence non-specific staining may have affected the interpretability of the results.

A potential explanation for these unexpected results could be that the PFF-injected animal brain sections are 2 years older than the control animal brain sections. Though these were stored appropriately, it could be that the age of the sections has resulted in their degradation and so are less able to be effectively stained. The staining level in different animals may also have been influenced by anatomical differences in the exact area of the brain being compared as it is not possible to select sections through the brain which cut through the region of interest in exactly the same way in two different animals. This could result in the section from the control animal featuring a more central section through the vagus nerve nucleus whereas the section from the PFF animal could only be catching the edge of the nucleus. To test this co-staining would have to be carried out with an anti-ChAT antibody. The use of further antibodies identifying phosphorylated α -synuclein may also have been useful to look for pathological α-synuclein aggregates. Unfortunately attempts to do this using an anti-Serine-129 antibody did not produce any useable images, and there was not sufficient time to replicate this work. It is also a possibility that neurotoxicity in this model is being driven by small oligometric α -synuclein species rather than larger aggregated species. These are not detectable using standard immunohistochemistry staining, and instead would require a much higher sensitivity imaging process such as that used by Emin et al., (2022). This work uses super resolution imaging techniques to image oligomeric a-synuclein, however homogenised or soaked brain tissue samples are required for this and as a result it is not possible to image these species in precise brain regions.

These explanations require further work to establish the true cause of the lack of α -synuclein staining seen in this model. Due to these results, I am unable to conclusively state that the cholinergic degeneration seen in the animals used in this study is due to α -synuclein.

2.5.4 Lack of olfactory deficits

The original study carried out by Kuan et al., (2019) showed olfactory deficits in PFFinjected animals after 6 months which were not seen in the control animals. This was less significant in our study, though a trend towards a deficit in the untreated PFF-animals was seen. As mentioned above, the sex of the animals used in this study was different to those used in the Kuan study. Sex may also underlie this difference in olfactory ability. In humans, women

Chapter 2

are known to show higher olfactory sensitivity than men, and similarly this is thought to be the case in most mammals, including rodents (Kass et al., 2017; Sorokowski et al., 2019). In human PD there are sex-dependent differences in the development of non-motor symptoms including hyposmia. One study looking at 951 patients found that a significantly higher percentage of female PD patients reported loss of taste/smell than male PD patients (Martinez-Martin et al., 2012). Hence tests of olfaction may be less sensitive in male rats because of inherent differences in olfactory ability, or pathological differences in the male versus female disease model. The potential differences in olfaction in the male and female α -synuclein rats would require further work in order to be clarified, possibly requiring a more sensitive olfactory test.

Weight gain could also have affected the rats' ability to perform in this behavioural test. The paradigm of this test requires starvation overnight to ensure that animals are keen to find food. Dosing with chocolate spread was carried out immediately prior to starvation (as this was the time when dosing was carried out every day). It is possible that due to the high fat content nature of the chocolate spread, starvation was not as effective, resulting in the animals being less motivated to find the pellet than they would otherwise have been. It also may hold that the allure of the Coco pop food pellet was less to these animals as they were accustomed to the smell and taste of chocolate spread. Furthermore, it is also possible that the weight of the animals made them less inclined or less generally able to move, which again would disincentivise them to retrieve the pellet.

2.5.5 Parallel studies

A further limitation with all of these experiments is that it was not possible to directly compare data from the PFF-injected animals with the saline-injected control animals because of subtle differences in study design and timing – the animals were a slightly different age, were housed in a different manner (fewer rats per cage), and the timings of the post-mortem work carried out were different. In retrospect, incorporation of larger treated and untreated saline control arms in the original study design would have been useful to allow confirmation of whether degeneration has occurred in brain regions. Though this has been shown through timecourse comparison in the PFF groups, it would have also been beneficial to directly compare the numbers/density of neurons in non-PFF animals

2.5.6 The use of candesartan

In addition to its role as an AT2-receptor antagonist, which underlies its use as a treatment for hypertension, candesartan has been shown to inhibit TLRs 2 and 4 in human monocytes and *in vivo* in mice (Dasu et al., 2009). The relevance of its action as a TLR2/4 inhibitor is particularly key as these are the two TLRs most implicated in PD. Different researchers argue for one of either TLR2 or TLR4 to be most relevant to the development and/or progression of PD. An advantage of using candesartan in this project is that the reagent inhibits both TLRs 2 and 4, allowing me to look at the combined effect of inhibition. Furthermore, candesartan is a licensed drug for use in humans thus allowing the direct and rapid translation of this work to the human disease if the study findings were clearly positive, and the potential repurposing of this drug in a clinical trial in PD patients.

In my study, candesartan has successfully protected the survival of the cholinergic neurons in both the vagus nerve nucleus and the hypoglossal nucleus of PFF-injected animals. This appears to be disease-state specific due to the lack of any significant differences between the cholinergic neuron density in untreated and candesartan-treated saline-injected control animals. This specificity of the effect to the disease-state implies an α -synuclein-dependent mechanism. The mechanism of action of candesartan will be investigated in *Chapter 3*. If our hypothesis is correct, then the mechanism is through candesartan decreasing the expression levels and/or the activation state of TLRs 2 and 4, resulting in an attenuated pro-inflammatory response to aggregated α -synuclein by microglia, and so not producing the neurotoxic cellular environment resulting from chronic inflammation. The relevance of the protection of cholinergic neurons within the vagus nerve nucleus is strong as this is thought to be one of the first regions to degenerate within the human PD brain according to Braak's hypothesis. The lack of a significant effect of candesartan in other brain regions is not necessarily conclusive to candesartan not having a protective effect, because this was likely due to the failure of the model to develop pathology in some key regions. For example, the model failed to result in neurodegeneration in the substantia nigra, and as a result candesartan had no significant effect on the number of dopaminergic neurons within this area. However, this does not exclude the possibility that had this degeneration occurred, candesartan would have had a protective effect on the survival of these neurons. Unfortunately this would require a second study to be carried out where dopaminergic degeneration occurred, and this was not within the scope of this PhD to carry out. As such, my data indicates that candesartan may be a useful protective agent against the development of α -synucleinopathy but more work is required to confirm its impact on dopaminergic degeneration.

In terms of practicality/feasibility, candesartan was a good candidate for use in this study. It was easily obtainable from a pharmacy due to its already being a licensed drug. The production of the drug-chocolate spread mixture for dosing was easy to carry out, though fairly time-consuming as this had to be carried out on a daily basis for the treatment of a large number of animals. The dosing of the animals was very simple as the chocolate spread was very appealing to the rats and the vast majority consumed it without even having to be removed from the cage – that is to say, no gavage was needed, and each rat took less than a minute to dose. This was particularly advantageous as due to there being a large number of rats needing dosing on a daily basis for 6 months, having to gavage or even inject animals in this manner for this length of time would simply not have been feasible, and would potentially have been traumatic to the animals. In addition, the tolerability of this drug was clearly seen, as treatment was carried out for a large number of animals (46 drug-treated animals total across the two parallel studies) over 6 months with no adverse effects seen in any animal.

Though a clear protective effect of candesartan treatment was seen in the increased survival of cholinergic neurons in the vagus nerve nucleus as compared to the untreated animals, this difference was not extensive and did not encompass all animals. The rats used in this study are Sprague Dawleys which are an outbred strain, making them non-genetically identical. This will result in differential effects both of the development of pathology in the model, and in the effect of the drug on the development of pathology. This has potentially resulted in the range of densities of cholinergic neurons: the genetically different rats responding diversely to candesartan treatment. This range of response will also be present if applied to humans, to a much greater extent – pathology will be much more heterogeneous in humans and in addition they may have multiple comorbidity conditions, experience different environmental factors, and be taking other medication. The key question arising from this study is whether the protective effect of candesartan on cholinergic neuron survival provides sufficient evidence to support testing this agent in the human disease. It could be argued that the limits of the protective ability of candesartan on any other aspects of the model means that further work is required to support the repurposing of this drug in a clinical trial.

A further point of interest is whether candesartan dosed orally is actually successfully reaching the brain to carry out its inhibitory effects. This was proven using liquid chromatography mass spectrometry (LCMS) and will be discussed in *Chapter 3.4.1*.

Another important consideration regarding candesartan use is that its effects are not limited to the inhibition of TLRs 2 and 4, and indeed its licensed function in the treatment of hypertension is through its inhibition of angiotensin II type 1 receptors. This inhibition of angiotensin has already been suggested as a method through which candesartan could be effective in the treatment of PD. For example, Rodriguez-Perez et al., (2018) published a study soon after I began my project, involving similar work with candesartan and other angiotensin type 1 receptor antagonists in an α -synuclein overexpression model produced through viral vector injection directly in the brain. Candesartan showed a protective effect on the development of pathology in this model in terms of protection against dopaminergic degeneration in the substantia nigra, against a loss of dopaminergic striatal terminals, and against a decline in motor ability – as determined through behavioural tests including the rotarod test. This work suggests that candesartan might be having a protective effect in our model through blocking angiotensin type 1 receptors rather than TLRs. In the next chapter, I will investigate potential mechanisms by which candesartan is exerting a protective effect on cholinergic neurodegeneration and behaviour.

	WLK control	CBH control only	CBH control +drug	WLK PFF	CBH PFF only	CBH PFF +drug
75 –					[]	
50 –	_	-			-	
37 –						
25 –						

Supplementary Figure 2.1 TH protein in my study shows post-translational modification not seen in Kuan model. Western blot for TH protein using samples taken from the original Kuan study and from my own study.

Chapter 3. INVESTIGATING HOW CANDESARTAN IS HAVING A PROTECTIVE EFFECT ON THE PROGRESSION OF THE A-SYNUCLEINOPATHY RAT MODEL

DECLARATION

The single ion recording (SIR) method used in the liquid chromatography mass spectrometry (LCMS) experiments was produced by Dr Stephen Thompson (Department of Clinical Neurosciences, University of Cambridge). The samples were run through by me with the assistance of Dr Thompson.

Immunohistochemistry slide scanning was carried out by the Histopathology and ISH core facility at CRUK Cambridge.

3.1 Aims of chapter

The PFF rat model produced by Kuan et al., (2019) is novel in its recapitulation of early features of PD, and in its modelling the slow rate of progression of disease. This makes it ideal for the exploration of drug intervention and how this affects the early stages of disease development – which is highly relevant in terms of future therapies to delay disease onset – as presented in *Chapter 2*. An unexplored feature of this model thus far is inflammation. *In vitro* work and many animal studies have demonstrated an inflammatory response to α -synuclein. The presence of aggregated α -synuclein in the brain of the PFF-injected animals has been demonstrated in the previous chapter and in the original publication by Kuan et al., (2019), however it has not previously been investigated whether this correlates with an increase in inflammation.

In the previous chapter it was found that daily candesartan treatment had a protective effect on the development of pathology in the PFF rat model of PD; the aim of the work presented in this chapter was to investigate the mechanistic basis of this effect, and in particular to determine whether candesartan suppresses inflammation in the rat brain via a TLR-dependent mechanism.

3.1.1 Objectives

- 1. To confirm that candesartan is crossing the blood-brain barrier, and for how long it is detectable within the brain.
- 2. To determine whether the peripheral injection of α -synuclein PFFs results in inflammation in the brain and/or periphery.
- 3. To determine the effect of candesartan treatment on levels of TLRs 2 and 4, and on inflammation in the α -synuclein PFF rat model.

3.2 Introduction

3.2.1 Neuroinflammation in Parkinson's disease

As discussed in *Chapter 1*, neuroinflammation is thought to play an important role in the progression of Parkinson's disease. The immune response is initially beneficial for injured tissue through promoting the clearance of cell debris and the secretion of neurotrophic factors to enhance neuron survival (Gao and Hong, 2008; Kordower, 2003; Trapp et al., 2007). However, a persistent and uncontrolled inflammatory response will result in chronic inflammation and the extended release of pro-inflammatory cytokines, reactive oxygen species, and nitric oxide. These are neurotoxic and further exacerbate neuron damage (Gao and Hong, 2008). The proposed role of inflammation in the progression of PD provides a potential avenue for therapeutic intervention. I hypothesised that the protective effect of candesartan seen in *Chapter 2* was occurring through blocking the chronic α -synuclein-induced inflammation, specifically through TLRs. Candesartan is a candidate for drug repurposing and has been proposed to act via TLRs 2 and 4, although its mechanism of action as an anti-hypertensive is through AT₁ receptor blockade (as discussed in *Chapter 1.4.3.1*).

3.2.2 Investigating inflammation in α -synuclein PFF model

As discussed in *Chapter 1.3.1.3*, inflammation has been seen in various animal models of PD and it has been suggested that it plays a role in the development of pathology within these models, however the specific link between this inflammation and α -synuclein is not clear. It is thought that α -synuclein activates the immune system; this has been shown through *in vitro* work in addition to in α -synuclein overexpressing models. In these latter models high levels of inflammation are seen in response to α -synuclein is hard to determine. The PFF model works through PFFs acting as a template for the recruitment of soluble endogenous α synuclein into insoluble Lewy body-like inclusions (Volpicelli-Daley et al., 2011). For this reason, the PFF model provides a better model to study inflammation in response to endogenous α -synuclein levels, as would be experienced in the human idiopathic Parkinson's brain. The utilisation of the PFF model in understanding neuroinflammation in PD has been reviewed by Stoll and Sortwell., (2022).

The presence of inflammation in PFF models of PD is less well established than in more acute models, perhaps indicating that the inflammatory response to α -synuclein isn't as high when only endogenous levels of α -synuclein are present. An acute inflammatory reaction directly to the injected PFFs has been reported; this occurs in the brain at 1-3 days postinjection, prior to the induction of any phosphorylated α -synuclein inclusions (Harms et al., 2017; Karampetsou et al., 2017). It is possible that this early inflammation occurs due to surgical injections causing mild trauma to the site of injection, and not as a reaction to the foreign PFFs. The relevance of this early inflammation is unknown – it is possible that this initial inflammation immediately fades and plays no role in the further development of inflammation, alternatively, it may play a key role in model development and early inclusion formation. A potential mechanism may involve the priming of immune cells. Previous work has shown that pre-exposure of microglia to a-synuclein, or a different pro-inflammatory stimulant such as LPS, prior to a secondary stimulus such as a TLR agonist (Roodveldt et al., 2013), an endogenous factor such as neuromelanin (Wilms et al., 2003), a proinflammatory factor (Facci et al., 2014), an endotoxin (Purisai et al., 2007), or exposure to a pro-inflammatory stimulant in a pre-existing disease model (Cunningham et al., 2005; McColl et al., 2007) results in a higher pro-inflammatory response on secondary exposure. During the establishment of centrally-injected PFF animal models more chronic inflammation has also been identified. This includes an upregulation of MHCII in response to α -synuclein inclusions (Duffy et al., 2018; Harms et al., 2017; Thomsen et al., 2021) - which is not seen in PBS-injected or monomerinjected animals – and an increase in Iba1⁺ cells showing morphology indicative of microglial activation (Duffy et al., 2018).

3.2.3 Determining inflammation within the animal brain

Inflammation within the post-mortem animal brain can be determined by quantifying cell activation markers, peripheral immune cell infiltration, or measuring downstream products of immune cell activation, such as pro-inflammatory cytokine levels. (Summarised in **Figure 3.1**).

3.2.3.1 Morphological changes

Microglial activation can be determined through morphological changes to the cell in addition to alterations in the expression of cell surface markers. An activated microglial cell shows amoeboid morphology with a large cell body and few cell processes – in contrast, microglia in the surveying state show ramified morphology with a small cell body and many extended processes to allow their surveying of the environment (Nimmerjahn et al., 2005). These morphological changes can be identified using simple light microscopy after staining with Iba1.

3.2.3.2 Marker expression

Iba1 (ionised calcium binding adapter molecule 1) is a specific marker for microglia/macrophages (Imai et al., 1996). It is a pan-microglial marker expressed in both ramified and activated microglia, though on activation expression of Iba1 increases (Ito et al., 1998). As such, Iba1 may be used as a marker of inflammation, but the presence of Iba1⁺ microglia alone is not sufficient for identifying inflammation – this should be used alongside morphological changes of the Iba1⁺ cells, differences in intensity of staining, or microgliosis (an increased number of microglia at the area of inflammatory insult). Other markers have been suggested as being specific for microglial activation; ED1 (or its equivalent in humans – CD68) is frequently cited as being one such marker. ED1 is specific to phagocytic microglia; it is expressed in the lysosomal membranes of phagocytes and this expression increases during phagocytic activity (Bauer et al., 1994; Damoiseaux et al., 1994). However this does not make it entirely specific to activated microglia – ED1 is upregulated on microglial activation but is still expressed along the spectrum of non-activated microglia (Jurga et al., 2020). MHCII is a further marker that is expressed in all microglia, but significantly upregulated in activated microglia (Jurga et al., 2020). MHCII mediates the adaptive immune response involving the activation of T-cells via antigen presentation by MHCII (Harms et al., 2013). Neither of these markers is specific for activated microglia, however the level of their expression may be indicative of the activation state of the cell.

Another feature of inflammation is the infiltration of peripheral immune cells into the brain. An example of this is the movement of peripheral macrophages and monocytes to a site of insult. Within the brain, peripheral macrophages/monocytes are usually restricted to perivascular locations, however when an inflammatory insult occurs these cells can move towards the insult to areas lacking perivascular spaces. In rats these cells express Iba1 and

MHCII, similarly to microglia, but can be distinguished through their expression of CD163 (Harms et al., 2017; Polfliet et al., 2006). As such, the expression of CD163 on cells within the rat brain and away from vasculature can be assumed to be infiltrating macrophages/monocytes and thus be indicative of an inflammatory insult.

3.2.3.3 Downstream pro-inflammatory factors

A further method of determining inflammation in the brain may be through the identification of downstream pro-inflammatory factors. The end result of inflammatory pathways is the activation of pro-inflammatory cytokines and chemokines, the upregulation of nitric oxide synthase, and the release of reactive oxygen species. These systems work so as to coordinate the activation and recruitment of immune cells and to regulate the immune response as a whole. This is with the eventual aim of controlling and eradicating the invading pathogen or intrinsic damage. As such, the detection of these pro-inflammatory factors within the brain is indicative of inflammation.



Figure 3.1 Microglial activation spectrum. Spectrum between ramified and amoeboid microglia including morphology, function, secretory markers, and surface markers. M1 and M2 phenotypes form a continuum; M1 describe classically activated pro-inflammatory microglia whereas M2 describe alternately activated anti-inflammatory microglia.

3.3 Methods

3.3.1 Study design

The overall study design is as described in *Chapter 2.3.1*.

3.3.2 Liquid chromatography mass spectrometry

Liquid chromatography mass spectrometry (LCMS) was carried out to assess whether candesartan had successfully reached the brain. Six Sprague Dawley rats were obtained and housed, 3 per cage, in standard laboratory conditions: 12-hour light and dark cycle, 22°C room temperature, 55% relative humidity, food and water ad libitum. Each rat was given one dose of candesartan (10 mg/kg/day; mixed with chocolate spread; Amias) at either 1 hour, 3 hours, 8 hours, 24 hours, or 30 hours pre-culling, the sixth rat was not dosed. All rats were culled via overdose of CO₂. A cannula was inserted into the heart to obtain 2 ml of blood and then a major vein was cut. The rat was then decapitated and the brain removed; simultaneously the liver and spleen were also removed. These were all flash frozen and stored at -80°C. All tissues were homogenised the following day in the organic solvent acetonitrile at a ratio of 9 MeCN: 1 H₂O using a mechanical homogeniser. After homogenising, samples were spun at 15000 g and 4°C for 20 minutes, the supernatant was removed and put through a 0.2 µm filter. A positive control was created by dissolving pure candesartan in MeCN, this was used to develop an SIR (single ion recording) method; this was carried out by Dr Stephen Thompson (Department of Clinical Neurosciences, University of Cambridge). All samples, including the positive control, were then run through the LCMS machine by me.

3.3.3 qPCR

qPCR was carried out to measure the expression levels of TLRs 2 and 4 in addition to the pro-inflammatory factors TNF- α , IL-1 β , and IL-6. Brainstem flash frozen immediately prior to death was removed from the freezer, weighed, and the correct volume of QIAzol lysis reagent (Qiagen; #79306) added before being vortexed until no visible tissue could be seen. RNA was isolated using the RNeasy Plus Universal Mini kit (Qiagen; 73404). RNA

concentration and chloroform contamination was determined using a nanodrop. RNA to cDNA conversion was carried out according to instructions in SuperScriptTM III First-Strand Synthesis SuperMix for qRT-PCR kit (Thermo Fisher Scientific #11752050). Samples (5 µl cDNA) were loaded in triplicate into a MicroAmpTM Optical 384 Well reaction plate (Thermo Fisher Scientific #4309849) using an automatic liquid pipetting robot (Eppendorf epMotion® 5070). The reaction mixture was formed of 10 µl TaqManTM Genexpression Master Mix (Thermo Fisher Scientific 4369510), 1 µl TaqMan primer (see **Table 3.1**) and 4 µl UltraPureTM DNase free H₂O (Thermo Fisher Scientific #10977035). This formed a total volume of 20 µl in each well. For each primer a No Template Control (NTC) was also loaded onto the plate as a negative control; this consisted of the TaqManTM mastermix, a TaqMan primer, and DNase free H₂O. The qPCR was carried out on the QuantStudioTM 12K Real-Time PCR System. The loaded plate was incubated for 2 min at 50°C and 10 min at 95°C, followed by 40 cycles of 10 sec at 95°C and 1 min at 60°C. To compare between the candesartan treatment groups the C_T was normalised both to GAPDH, and to the average of the untreated group - i.e. the control+drug were normalised to the control only, and the PFF+drug were normalised to the PFF only. This gave both the control and PFF untreated groups normalised to 1, and the fold change of the candesartan-treated groups compared to the corresponding value.

Gene	TaqMan Assay ID	Reference sequence
TLR2	Rn02133647_s1	NM_198769.2
TLR4	Rn00569848_m1	NM_019178.1
TNF-α	Rn99999017_m1	NM_012675.3
IL-1β	Rn00580432_m1	NM_031512.2
IL-6	Rn01410330_m1	NM_012589.2
AT-1	Rn02758772_s1	NM_030985.4
GAPDH	Rn01775763_g1	NM_017008.4

Table 3.1 TaqMan primers used for qPCR.

3.3.4 Western blot

Western blots to quantify protein levels of TLRs and their downstream factors were carried out on animal midbrain and brainstem. Flash frozen brain regions were individually homogenised in RIPA lysis buffer with Complete protease inhibitors (Merck #11873580001), and a BCA Protein Assay Kit (Thermo Fisher #23227) was used to quantify protein concentration. Proteins (60 µg) were separated by electrophoresis on a 4-12% polyacrylamide gel (Invitrogen *NP0336BOX*), after which they were transferred to a polyvinylidene fluoride membrane (Cytiva #10600023) for 90 min at a constant voltage of 30 V. Effective transfer was checked using ponceau staining. Each membrane was blocked with either 5% milk-TBST or

5% BSA and then incubated overnight at 4°C with the specific primary antibody, as listed in **Table 3.2**. The membrane was washed and then incubated with the appropriate HRP-conjugated secondary antibody for 1 hour. Immunoreactivity was visualised with chemiluminescence (Cytiva *GERPN2232*).

Gene	Species	Dilution	Manufacturer/Catalogue No.
TLR2	Rabbit monoclonal	1:1000	Abcam <i>ab209217</i>
TLR4	Rabbit polyclonal	1:1000	Abcam <i>ab13867</i>
MyD88	Rabbit polyclonal	1:1000	Abcam <i>ab2064</i>
TNF-α	Rabbit polyclonal	1:1000	Abcam <i>ab6671</i>
TNF-α	Rabbit monoclonal	1:1000	Abcam <i>ab205587</i>
NFĸB	Rabbit monoclonal	1:1000	Cell signalling Technology #3033
B-actin HRP-	Mouse monoclonal	1:5000	Santa Cruz Biotech sc-47778HRP
conjugated			

Table 3.2 Antibodies used for western blotting.

3.3.5 Immunohistochemistry

Immunohistochemistry of various microglial and monocytic markers was carried out on a 1:12 series of post-mortem brain sections. Sections were first quenched with 10% methanol and 10% hydrogen peroxide before being blocked in TBS with 0.3% Triton X-100 and 5% serum for 1 hour. Sections were then incubated overnight at 4°C with primary antibodies diluted in TBS with 3% Triton X-100 and 1% serum, as listed in **Table 3.3**, followed by a 3 hour incubation with the relevant biotinylated secondary antibody (Invitrogen #A10042; Vector Laboratories #BA2001), and a 30 minute incubation with avidin-biotin complex (Vector #PK-6100). Antigen visualisation was carried out using 3,3'-Diaminobenzidine (DAB solution; Vector #SK-4100). DAB solution was left on the tissue from 1 minute to 4 minutes, dependent on tissue colour change. Sections were mounted on coverslips and dehydrated in sequential transfers in 70% EtOH, 90% EtOH, and 100% EtOH followed by immersion in 100% xylene for 30 minutes and then mounted with DPX mounting medium (Thermo Scienific).
The role of Toll-like receptors in Parkinson's disease

Gene	Species	Dilution	Manufacturer/Catalogue No.
Iba1	Rabbit polyclonal	1:500	Wako Chemicals #019-19741
MHCII	Mouse monoclonal	1:500	BD Biosciences #554926
ED1	Mouse monoclonal	1:500	BioRad MCA341GA
CD163	Mouse monoclonal	1:500	BioRad MCA342GA

Table 3.3 Antibodies used for immunohistochemistry on brain sections

Quantification of Iba1 staining was carried out using 2 different methods. The slides were scanned at 20X magnification at CRUK Cambridge along with ChAT-stained slides and TH-stained slides from the same animals. Iba1 and ChAT/TH staining had been carried out on adjacent 1:12 sections. Regions of interest on Iba1 stained sections were drawn through comparison with the adjacent ChAT/TH-stained section where the region of interest, i.e. the hypoglossal and vagus nerve nuclei, the substantia nigra, and the striatum could clearly be seen. (**Figure 3.2**).



Figure 3.2 Scanned slides for IHC quantification of Iba1 staining. (a) (i) ChAT stained slide with clear hypoglossal and vagus nerve nuclei shown in red box; (ii) Iba1 stained slide with region corresponding to (i) shown in red box. (b) Higher magnification of (i) ChAT staining; (ii) region of interest shown in green clearly corresponding to ChAT-positive staining in (i).

The first method used Aperio ImageScope software, specifically the Positive Pixel Count v9 algorithm which can be optimised to quantify brown-coloured staining (DAB staining). The algorithm works by analysing each pixel and allocating them into 4 colour

intensity ranges: background (blue), weak positive (yellow), medium positive (orange), and strong positive (red). The parameters were optimised so as to identify microglia as 'strong positive'. (**Figure 3.3**). The software produces outputs of number of strong positive pixels and intensity of strong positive pixels which were determined as the measure of microglial area and intensity. The ROIs showed a large variation in area and so to account for this the output is reported as number of strong positive pixels divided by the ROI area in mm². The parameters were kept the same across all of the samples. This method was problematic as background staining would be identified as a 'strong positive pixel' resulting in inaccurate detection of the specific staining of microglia (see comparison between **Figure 3.3b & c**). This method also only gave an output in terms of positive pixels, so not allowing a specific quantification of number of microglia.

The role of Toll-like receptors in Parkinson's disease



Figure 3.3 Imagescope software analysis. (a) ROI with optimised parameters applied; microglia identified in red. (b) Low background staining and (c) high background staining with corresponding overlay mask.

The second method used ImageJ software; this method allowed the counting of microglia through thresholding methods where only particles over a specified diameter were selected. 0.5 mm^2 square ROIs in each brain region were extracted for analysis. Briefly, the colour and size threshold were optimised for each box to select the cell bodies alone, an overlay mask was created (seen in red **Figure 3.4a**) and particle analysis was carried out to count the number of microglia in addition to the area of the selected cell bodies; this was averaged over the number of boxes per brain region. The parameters had to be adjusted for each box due to differing levels of background staining. To ensure non-bias when altering parameters the animal code and group number was blinded. This method was validated through the manual counting of a select number of boxes (N=17), which was shown to positively correlate with the automated count (Simple linear regression R^2 =0.6615; black line). (**Figure 3.4**). I decided that this was the optimal method and my reported results are based on this.





Figure 3.4 ImageJ software analysis. (a) Example of step-wise image processing to produce microglial cell count and area quantification. (b) Comparison of automated and manual microglial cell count shows a positive correlation as determined through a simple linear regression (black solid line). R^2 =0.6615.

3.3.6 Meso Scale Discovery assay

Meso Scale Discovery (MSD) assays were carried out on animal serum to assess levels of pro-inflammatory cytokine expression. MSD assays use proprietary technology integrating electrochemiluminescence and multiarray techniques to detect multiple proteins from one sample. The technology utilised allows for much higher sensitivity than a regular ELISA

(enzyme-linked immunosorbent assay). A blood sample was taken from each animal at baseline; this was then repeated every 2 months until the end of the study. The blood was allowed to clot at room temperature before being spun for 15 minutes at 2000 g. The serum was removed and stored at -80°C prior to later batch analysis. Mesoscale discovery V-PLEX Proinflammatory Panel 2 kits (MSD *K15059D*) were used to analyse the serum for the pro-inflammatory cytokines IFN- γ , IL-1 β , IL-4, IL-5, IL-6, KC/GRO, IL-10, IL-13, and TNF- α following the instructions in the kit. Briefly, the plates were blocked for 1 hour using the blocking agent H provided, the plates were then washed before samples and calibrators were added, these were then incubated for 2 hours with shaking. The plates were incubated with the detection antibodies for 2 hours before, finally, read buffer was applied to the washed plates and the plates immediately read using the MSD plate reader. Results were analysed using the MSD Discovery software.

3.3.7 Statistical analysis

Data analysis and graph generation was carried out using GraphPad Prism 8.3. Data normality was assessed using the Shapiro-Wilk test. Pairwise comparisons were carried out between untreated and candesartan-treated PFF or control groups using a Mann-Whitney test. For repeated measures analysis of groups over time a Mixed-effect analysis was carried out followed by Šídák's multiple comparisons test.

3.4 Results

3.4.1 Does candesartan cross the blood brain barrier?

The premise of my study design requires that candesartan is getting into the brain to have its hypothesised action on the TLRs on neuroimmune cells. Liquid chromatography mass spectrometry (LCMS) was carried out on the brains, livers, and spleens of 6 animals each dosed with candesartan at a different timepoint prior to culling (1 hour, 3 hours, 8 hours, 24 hours, and 30 hours pre-culling). A clear peak at the correct molecular weight indicating the presence of candesartan in the brain, liver, and spleen was present in all of the animals dosed 24 hours or less prior to culling. No significant peaks above background noise were seen in the untreated negative control animal, or in the animal dosed 30 hours prior to culling. The highest peak seen at 1.91 does not have the correct molecular weight to be candesartan, and does not correspond to the clear peak seen in the pure candesartan positive control sample. (**Figure 3.5**). This experiment shows that candesartan is able to cross the blood-brain barrier of the rats within an hour of their dosing, and is still present in the brain at 24 hours post-dosing, but has been metabolised by 30 hours post-dosing. Similar peaks were also observed in the liver (see **Figure 3.6**).



Figure 3.5 Candesartan is able to enter the brain, and persists for 24 hours. LCMS plots of (a) homogenised brain from untreated rat; (b) pure candesartan suspended in HPLC-grade acetonitrile; and homogenised brain from rat dosed with candesartan (c) 1 hour, (d) 24 hours, and (e) 30 hours prior to culling.



Figure 3.6 Candesartan persists in the liver. LCMS plots of (top to bottom) homogenised liver from untreated rat; and from rat dosed with candesartan 1 hour, 24 hours, and 30 hours prior to culling.

3.4.2 Is candesartan successfully reducing the expression of TLR2/4?

The hypothesis underlying this study is that candesartan will have a protective effect on the development of pathology in the α -synuclein animal model through its blocking of TLRs, and subsequent pro-inflammatory effects. To test whether candesartan was having this effect, I determined the levels of TLR expression, and TLR protein levels in the animal brains. qPCR of TLRs 2 and 4 showed no fold-change in the expression of either TLR2 or TLR4 in rat brainstem on candesartan treatment compared to untreated; all normalised to GAPDH and to either the untreated control or untreated PFF groups. (**Figure 3.7**)



Figure 3.7 Candesartan treatment does not have an effect on the expression of TLRs in the rat brainstem. Comparison between the fold change in (a) TLR2 and (b) TLR4 mRNA levels of drug-treated compared to untreated control (left) and PFF-injected (right) animals. Normalised to GAPDH.

Western blots using tissue from the midbrain showed a trend towards decreased TLR protein levels in both drug-treated PFF-injected animals and drug-treated control animals compared to untreated PFF-injected and control animals respectively. These trends did not reach significance for either PFF-injected or control animals. (**Figure 3.8**). (Mann-Whitney test; **Figure 3.8b** control p=0.1049, PFF p=0.1381; **Figure 3.8c** control p=0.2786, PFF p=0.1017). There also appears to be a trend towards increased TLR protein levels in the PFF-injected animals compared to the controls, though with the caveat that these cannot be directly compared due to their being conducted as parallel studies at different times.



Figure 3.8 Candesartan treatment results in a trend towards decreased protein levels of TLRs 2 and 4 in the rat midbrain. (a) Representative western blot of TLR2 and TLR4 with β -actin as a loading control; quantification of (b) TLR2; and (c) TLR4 protein levels of untreated and drug-treated saline-injected control (left) and PFF-injected (right) animals. Bands normalised to β -actin and then to the mean of 'control only' bands to allow comparison between different blots. Standard deviation shown in error bars. Mann-Whitney test; p>0.05 for all comparisons).

In the brainstem, TLR2 was significantly decreased in the candesartan-treated control animals (Mann-Whitney test; p<0.05), however this was not seen in the brainstem of PFF-injected animals. (**Figure 3.9**). No significant changes in TLR4 levels were observed between treated and untreated groups (Mann-Whitney test; p>0.05 for all other pairwise comparisons).





Figure 3.9 Candesartan treatment results in a trend towards decreased protein levels of TLR2 but not of TLR4 in the rat brainstem. (a) Representative western blot of TLR2 and TLR4 with β -actin as a loading control; quantification of (b) TLR2; and (c) TLR4 protein levels of untreated and drug-treated saline-injected control (left) and PFF-injected (right) animals. Mann-Whitney test; $p^*<0.05$. Bands normalised to β -actin and then to the mean of 'control only' bands to allow comparison between different blots. Standard deviation shown through error bars. Mann-Whitney test; p>0.05 for all comparisons).

Further work to identify whether candesartan was blocking TLRs 2 and 4 used western blots of MyD88, the universal adapter protein of TLRs, in animal midbrain. The antibody used claimed to show bands at 35 kDa, however, though these bands were present in the blots, they were not able to be quantified due to the non-specific bands surrounding the correct MW. Instead the bands at 70 kDa were used for quantification as they were assumed to be MyD88 dimers (Loiarro et al., 2005). A trend towards decreased MyD88 was seen in the candesartantreated compared to untreated control animals, however this did not reach significance (Mann-Whitney test). No such trend was seen in the PFF-injected animals. (**Figure 3.10**).





Figure 3.10 Candesartan treatment does not alter protein levels of MyD88. (a) Representative western blot of MyD88 with β -actin as a loading control; (b) quantification of MyD88 protein levels of untreated and drug-treated saline-injected control (left) and PFF-injected (right) animals. Bands normalised to β -actin and then to the mean of 'control only' bands to allow comparison between different blots. Mann-Whitney test; p>0.05 for all comparisons).

The primary function of candesartan is as an AT_1 blocker, however studies have also shown a decrease in $AT1_1$ expression on candesartan treatment (Rodriguez-Perez et al., 2018). Therefore the expression of AT_1 in the animal brain was determined using qPCR. No fold change was seen in the expression of AT_1 on candesartan treatment compared to untreated; all normalised to GAPDH and to either the untreated control or untreated PFF groups. (**Figure 3.11**).



Figure 3.11 Candesartan treatment does not alter AT_1 *expression. Comparison between the fold change in* AT_1 *mRNA levels of drug-treated compared to untreated control (left) and PFF-injected (right) animals. Normalised to GAPDH.*

3.4.3 Does candesartan treatment impact microglial activation?

I looked at the effect of candesartan treatment on microglial activation using Ibal immunohistochemistry. This was carried out in 3 areas of interest – the vagus and hypoglossal nuclei (grouped together for analysis), the substantia nigra, and the striatum. At 6 months, Candesartan was shown to have no effect on microglial activation in either PFF-injected animals or saline-injected control animals as determined by the number and area of the microglia. This was the case in the vagus/hypoglossal nerve nuclei (**Figure 3.12**), the substantia nigra (**Figure 3.13**), and the striatum (**Figure 3.14**). In addition, although direct statistical comparison between control and PFF groups was not made due to the fact that the studies were conducted separately (as described in *Chapter 2.4.2*), the shape and size of the microglia look very similar between the controls and the PFF-injected animals. It is possible therefore that the

lack of response to candesartan is due to a lack of activation of microglia as produced by this α -synuclein model.



Figure 3.12 Candesartan does not have any effect on microglial number or morphology in the hypoglossal and vagal nerve nuclei. Representative images of Iba1 staining in the outlined vagal nerve/hypoglossal nuclei in (a) control animals and (b) PFF-injected animals. Quantification of (c) number of Iba1⁺ cells and (d) area of Iba1⁺ cells in control (left) and PFF-injected (right) animals. Quantification carried out on 0.5 mm² boxes placed within the region of interest.



Figure 3.13 Candesartan does not have any effect on microglial number or morphology in the substantia nigra. Representative images of Iba1 staining in the outlined substantia nigra in (a) control animals and (b) PFF-injected animals. Quantification of (c) number of Iba1⁺ cells and (d) area of Iba1⁺ cells in control (left) and PFF-injected (right) animals. Quantification carried out on 0.5 mm² boxes placed within the region of interest.



Figure 3.14 Candesartan does not have any effect on microglial number or morphology in the striatum. Quantification of (a) number of Iba1⁺ cells and (b) area of Iba1⁺ cells in control (left) and *PFF-injected (right) animals. Quantification carried out on 0.5 mm*² boxes placed within the region of interest.

I used a selection of other antibodies aimed specifically at markers of microglial activation. MHCII is expressed on all microglia; the antibody used here has been seen in various studies (including Harms et al., 2017 and Rodriguez-Perez et al., 2018) to identify activated microglia due to their expressing higher levels of MHCII. There were no apparent differences between treatment groups at 6 months, and indeed a very low number of cells throughout the brain expressing MHCII – as detected by this antibody – at all. (**Figure 3.15a-b**) An animal showing a clear inflammatory insult in the neck of the cerebellum was used as a positive control. (**Figure 3.15c**) Through comparison with this, it can be seen that the pro-inflammatory effects of the PFF-injection are negligible, and thus treatment with candesartan

has no clear effect on the activation of microglia. (Figure 3.15). Staining of MHCII was also carried out at the 2 and 4 month timepoints; no differences were seen between any treatment groups. (Figure 3.16). In addition, MHC staining was compared between the PFF animals 4 months post PFF injection, and the control animals 6 months post saline injection in order to allow comparison of animals at a similar age (PFF-injected animals were 14 weeks old at the point of injection whereas the saline injected animals were 8 weeks old). This was done because age may have an effect on neuroinflammation. No differences were seen between the 4-month PFF animals and the 6-month control animals indicating that inflammation is not occurring in this model. (Figure 3.17). Due to very low levels of staining, the IHC was not quantified, however further investigation was carried out through western blots to quantify levels of MHCII. Western blots also did not show any significant differences in levels of MHCII between treatment group in either the brainstem or the midbrain (Mann-Whitney test; p>0.05). (Figure 3.18).



Figure 3.15 Candesartan has no effect on MHCII expression at 6 months. Representative images of MHCII staining in the vagal nerve/hypoglossal nuclei in (a) control animals, (b) PFF-injected animals, and (c) an animal experiencing an inflammatory insult in the cerebellum. All images are at same magnification and have similar levels of background staining.



Figure 3.16 Candesartan has no effect on MHCII expression at different timepoints. MHCII staining in untreated (left; blue) and candesartan-treated (right; green) hypoglossal/vagal nerve nuclei at (a) 2 month and (b) 4 month timepoints.



Figure 3.17 No differences in inflammation (as assessed using MHCII) are seen between PFFinjected and saline-injected control animals when at comparable ages. MHC staining in (a) untreated or (c) candesartan-treated PFF-injected and (b) untreated or (d) candesartan-treated control salineinjected animals at a similar age. PFF-injected animals at 30 weeks old; saline-injected control animals at 32 weeks old.

Chapter 3



Figure 3.18 Candesartan is not associated with changes in MHCII protein levels. Representative western blots of MHCII and quantification of blots in the (a-b) midbrain and (c-d) brainstem of control and PFF-injected animals. The well in (c) – well 9 in PFF+drug – which can be seen to have run poorly was removed from all analysis. Mann-Whitney test; p>0.05.

A further marker of microglial activation is ED1, expressed by amoeboid microglia. This staining did not show any apparent differences between treatment groups on visual inspection. However, it was very difficult to tell whether the staining was successful. Even in the animal with an inflammatory insult in the cerebellum which is being used as a positive control there was very little obvious staining. (**Figure 3.19**). Only when viewed at the highest magnification was there some noticeable staining in the positive control, but this was not seen at all in any of the other animals (data not shown). Finally, as a further indicator of inflammation (though not specifically of microglial activation), a marker of infiltrating monocytes, CD163, was used. This is indicative of whether peripheral cells are becoming activated and infiltrating the brain. There were no apparent differences on visual inspection of stained sections between treatment groups at 6 months. (**Figure 3.20**).



Figure 3.19 ED1 staining is difficult to identify in an animal with an inflammatory insult in the cerebellum being used as a positive control. (a) Iba1 staining and MHCII staining in the whole brain clearly showing an increase in expression at the neck of the cerebellum. (b) ED1 staining is not clear anywhere in the brain, including in the neck of the cerebellum where inflammation is occurring. At 20X magnification (right) staining can be seen as small dots at the area of inflammation, however this is not seen elsewhere in the brain.



Figure 3.20 CD163 is not differentially expressed in different treatment groups. CD163 staining in untreated (left) and candesartan-treated (right) (a) PFF-injected and (b) saline-injected control animals. Magnification 20X.

3.4.4 Is candesartan treatment associated with changes in downstream inflammatory markers?

Activation of TLRs was investigated in the brainstem – the area where neurodegeneration was seen in this model – through measuring the expression of downstream inflammatory molecules. TNF α , IL-1 β , and IL-6 expression levels were determined using qPCR. No differences were seen in either the control or PFF-injected groups on candesartan treatment. (**Figure 3.21**).



Figure 3.21 Candesartan treatment is not associated with alterations in the expression of downstream pro-inflammatory cytokines in the brainstem. Expression of (a) TNF- α , (b) IL-1 β , and (c) IL-6 in untreated and candesartan-treated control (left) and PFF-injected (right) animals. Normalised to GAPDH. Untreated normalized to 1 and drug-treated displayed as fold change of the untreated animals in each respective group.

I also attempted to measure the downstream inflammatory protein levels of TNF- α (**Figure 3.22**) and NF- κ B (**Figure 3.23**) using western blotting techniques, but none of the antibodies were successful at identifying a band at the correct MW so protein expression levels of these factors were not determined.



Figure 3.22 Western blots of TNF-a were unsuccessful at identifying a band at the correct molecular weight for TNF-a (26 kDa). Example western blots using (a) Abcam ab6671 or (b) Abcam ab205587.



Figure 3.23 Western blots were unsuccessful at identifying band at the correct molecular weight for $NF-\kappa B$ (65 kDa). Western blot using $NF\kappa B$ Cell Signalling Technology #3033 antibody blocked and diluted in (a) 5% milk or (b) 5% BSA. Spleen used as positive control.

3.4.5 Does candesartan treatment influence serum peripheral immune factors?

Peripheral inflammatory markers were measured using an MSD assay. This was carried out on animal serum taken from all animals at baseline and then each 2 month timepoint. The majority of the cytokines in the multiplex assay were below the limit of detection or had such low values that the mean coefficient of variation (CVs) were too high and so deemed unacceptable for use. The three cytokines with acceptable CVs and so used for investigation were TNF- α , IL-10, and KC/GRO (a chemoattractant for monocytes and neutrophils known as CXCL1 in humans). These three assays had mean CVs of below 40% - 30.8%, 32.2%, and 3.7% respectively. These cytokines also contained many samples below the limit of detection – TNF- α PFF 31.6%, control 26.3%; IL-10 PFF 57.4%, control 7.9%; KC/GRO PFF 5.1%, control 0%. For analysis, these samples below the limit of detection were allocated a value between 0 and the lower limit of detection, as is standard practice.

Investigating changes in cytokines over time does not show any significant differences in the levels of TNF- α or IL-10 over time. (**Figure 3.24a-b**). In addition, the levels of IL-10 at the 2 and 6 month timepoints were all below the level of detection, possibly indicating an issue with this cytokine assay. However, KC/GRO shows a clear peak at the 2 month timepoint in the PFF-injected animals which is not seen in the control animals which show a significant decrease in KC/GRO over time; this could be indicative of an early peripheral immune response in the PFF-injected animals. (**Figure 3.24c**). (Mixed-effects analysis followed by Šídák's multiple comparisons test; p<0.05; p<0.01; p<0.0001). Comparison between treatment groups at different timepoints showed significant decreases in KC/GRO on candesartan treatment at 2 months, indicating that candesartan may be attenuating the early peripheral immune response. Candesartan is also seen to significantly decrease KC/GRO in the control study at the 6 month timepoint (**Figure 3.24c**). (Mixed-effects analysis followed by Šídák's multiple comparisons test; p<0.05). No other significant differences were seen between treated and untreated groups.



Figure 3.24 Peripheral serum cytokine levels over time in candesartan treated versus untreated animals. Calculated concentration mean of (a) TNF- α , (b) IL-10, and (c) KC/GRO in candesartantreated and untreated control (left) and PFF-injected (right) animals. Mixed-effects analysis followed by Šídák's multiple comparisons test; $p^*<0.05$; $p^{**}<0.01$; $p^{****}<0.0001$. Mixed-effects analysis followed by Šídák's multiple comparisons test; $p^{\#}<0.05$.

3.5 Discussion

3.5.1 Summary

Neuroinflammation has been implicated in the development and progression of Parkinson's disease. The hypothesis being investigated in this thesis is whether this neuroinflammation is mediated though the binding of aggregated α -synuclein to Toll-like receptors found on microglia and innate immune cells, leading to an exacerbation of neurodegenerative pathology. To test this, the drug candesartan, a putative TLR2 and 4 blocker, was used in an α -synuclein animal model. The use and effects of candesartan treatment on behaviour and neurodegeneration in this animal model is described in the previous chapter.

In this chapter I sought to identify whether the mechanism through which candesartan was having a protective effect on the development of cholinergic degeneration was through the inhibition of TLR activation, and subsequent neuroinflammation. The results show that candesartan is present in the brain, and thus potentially able to cross the blood brain barrier. However my data do not clearly show a significant impact of candesartan on the levels of TLR expression or protein levels in the brains of PFF-injected animals – though a significant decrease in TLR2 protein levels in the brainstem of candesartan-treated compared to untreated control animals was seen. There also does not appear to be any effect of candesartan on downstream neuroinflammation as shown by a lack of microglial activation, and by the lack of detectible changes to the expression of downstream inflammatory cytokines in the brain including TNF- α and IL-1 β . Finally, the PFF model appears to show early activation of the peripheral immune system as seen through a significant increase in the chemokine CXCL1 (referred to as KC/GRO in the MSD assay). This is attenuated on candesartan treatment, indicating that candesartan could be having a protective effect through reducing peripheral inflammation.

These results suggest that candesartan is having a minimal effect on TLR activation and inflammation in the brain of this α -synuclein PFF animal model. However, the data indicate that very limited inflammation is occurring within the PFF-injected animals' brains, thus making it difficult to measure any impact of candesartan on brain inflammatory processes. Whilst it is possible that early reduction of peripheral inflammation in candesartan treated PFF animals could have some impact on reducing neurotoxicity, it seems likely that the action of candesartan to attenuate degeneration of cholinergic neurons in the brainstem is occurring through mechanisms other than anti-inflammatory ones.

3.5.2 Lack of neuroinflammation

The main limitation of this study was that significant brain inflammation was not induced by the peripheral injection of the α -synuclein PFFs, making it difficult to evaluate any anti-inflammatory effect of candesartan. This lack of inflammation was shown through comparison of PFF-injected and saline-injected control animals at the same experimental timepoint and through a comparison of PFF-injected and saline-injected and saline-injecte animals at the same age but different experimental timepoint (required due to the two parallel studies having animals at slighty different ages at the start of experimentation).

Inflammation is found in centrally-induced PFF models – including an upregulation of MHCII (Duffy et al., 2018; Harms et al., 2017; Thomsen et al., 2021) and an increase in activated Iba1⁺ microglia (Duffy et al., 2018). The markers of inflammation reported in centrally-induced PFF models were investigated here in the Kuan et al., (2019) model but neither a change in microglial morphology nor an increase in MHCII expression was seen at any of the recorded timepoints. The peripheral site of injection and the conjugation of the PFFs to RVG9R (the rabies glycoprotein allowing the passage of the PFFs across the blood-brain) are the two main differences between the model used in this study and other models showing PFF-induced inflammation.

As described in *Chapter 2.2.1.4*, one of the advantages of the Kuan et al., (2019) model is that the injection occurs peripherally. This decreases the probability that surgical injection-specific inflammation will be seen in the brain, potentially removing a source of immune cell priming in the brain. The PFFs still enter the brain, but through less invasive mechanisms. If true that PFFs are responsible for priming immune cells, as discussed in the chapter introduction, it is unclear why this would not occur in the model used here unless the RVG9R conjugation prevented the immunogenicity of the PFFs without affecting their ability to induce endogenous α -synuclein aggregation. In the original study using RVG9R, Kumar et al., (2007) showed that repeated administration of RVG9R did not induce inflammatory

cytokines or antibodies against the peptide. This negates the possibility of the RVG9R itself resulting in the initial activation or priming of the immune system in the model used here. Other peripherally-induced models, such as that produced through the injection of PFFs into the duodenum (Challis et al., 2020), have shown local inflammation induced early post-injection. Unfortunately they have not investigated whether this inflammation persists post the initial injection insult, or whether there is inflammation in the brain in addition to the α -synuclein pathology which spreads to the brains of these mice when aged (Challis et al., 2020). Investigation into this would be interesting to determine whether later-onset neuroinflammation, independent to the initial acute response to a foreign substance (i.e. the PFFs), occurs without CNS injection.

The immunogenic α -synuclein species in PD has not been fully established. It is thought that pro-inflammatory effects arise from the aggregation of the endogenous α synuclein into soluble oligomeric species followed by arrangement into larger fibrils and ultimately Lewy body-like inclusions (Cremades et al., 2012). Multiple studies have shown the formation of these inclusions on CNS-injection of PFFs. Some have suggested fibrils to have the greatest pro-inflammatory effect (Grozdanov et al., 2019; Hoffmann et al., 2016). Conversely, others suggest that the oligomeric form has the greatest pro-inflammatory effect and is therefore the most toxic (Emin et al., 2022). In the animal model used in my work, aggregation into high molecular weight (HMW) species has been identified in Chapter 2.4.2 using immunoprecipitation and immunoblotting techniques. However, the specific presence of Lewy body-like pathology was not established in this work despite attempts made to replicate that seen in Kuan et al., (2019). This unfortunately made it impossible to determine specifically where α -synuclein pathology was occurring and as a result to determine whether there was any colocalisation with activated immune cells. Antibodies claiming to distinguish different α synuclein species have been produced, however their specificity is doubted by many in the field. As a result, it is very difficult to identify forms of α -synuclein other than large insoluble fibrils, and so the presence of oligomeric species within the animal model used here remain elusive. Investigation into the immunogenic species of α -synuclein will be carried out in Chapter 4.

It has been suggested that inflammation in human PD occurs very early in the disease course (Iannaccone et al., 2013; Kim et al., 2022; Ouchi et al., 2005). This could similarly be occurring in this rat model of PD, however the earliest timepoint at which data was gathered –

including blood for peripheral inflammation determination and post-mortem brains for central inflammation determination – was at 2 months post PFF injection. No neuroinflammation is occurring at this earliest timepoint, however in the periphery there is a significant increase in the chemokine KC/GRO (CXCL1). CXCL1 is a chemoattractant for multiple immune cells including monocytes, T-cells, and neutrophils (Serdar et al., 2020). This could be indicative of the peripheral immune system being activated by PFFs, either as the injected PFFs themselves (though it is unlikely that these will still be present in the animal at 2 months post-injection), or in response to the aggregation of endogenous α -synuclein. Monocytes can be activated directly by α -synuclein, and α -synuclein may then be presented to T-cells and B-cells resulting in their activation and further peripheral inflammation. It is possible that the KC/GRO response seen at 2 months is the tail-end or the initiation of a more intense peripheral immune activation which has not been detected at earlier timepoints through lack of data, or has already faded by the 4 month timepoint. Peripheral inflammation can have neurotoxic effects, however these commonly occur through the induction and exacerbation of central inflammation, leading to a neurotoxic environment (Filipov, 2019) which is not seen in my work. Candesartan is having an attenuating effect on this peripheral inflammation at 2 months, and so however peripheral inflammation is having a neurotoxic effect, the action of candesartan in reducing this may be a contributory factor to the neuroprotective effects observed in *Chapter 2*. One limitation of the MSD assays was that different timepoints were carried out on different plates and so inter-plate variability may contribute to differences seen between timepoints. This could be rectified through the use of a positive control within each plate, however unfortunately this was not carried out within the assays completed here.

Future investigation into whether there is an early acute inflammatory effect specifically to the injected PFFs could be carried out. It would be interesting to observe whether there is an initial inflammatory response to the foreign substance of PFFs, as if not this could aid the discussion of whether this priming of the immune system is crucial to the development of inflammation further along the disease progression. It would also be interesting to determine whether there is an inflammatory response within these first 2 months leading to the induction of KC/GRO, and whether in my study I have simply missed a larger peripheral response due to not collecting data at the correct timepoint. This work warrants further exploration into peripheral inflammation, and could involve peripheral immunophenotyping studies in this model.

3.5.3 Alternative mechanism of action of candesartan

Candesartan is a licensed drug for the treatment of hypertension. It achieves this through its action as an angiotensin II receptor blocker (AT₁ blocker). Candesartan binds specifically to the AT_1 receptor and inhibits the binding of angiotensin II through a noncompetitive mechanism. It has been suggested that through this blockade candesartan may be able to have an effect on the progression of Parkinson's disease. Some work has shown that candesartan inhibits the expression of AT_1 receptors (Rodriguez-Perez et al., 2018), however this is not universal and others have shown no effect on mRNA expression of AT₁ (Nishimura et al., 2000). My work similarly showed no differences in mRNA expression of AT_1 receptors on candesartan treatment. LCMS was carried out to ensure that the candesartan was reaching the brain and clear peaks corresponding to candesartan were seen in the brain for up to 24 hours (Figure 3.5). One consideration in this particular experiment is that the animal brains were not perfused and for that reason there would still have been blood in the cerebral blood vessels. As a result, it cannot be claimed for certain that any candesartan detected in the homogenised animal brains is from candesartan in the brain parenchyma rather than from residual blood in these cerebral blood vessels. This could explain why AT₁ mRNA levels are not decreased. Further indicators of AT_1 blockade were not assessed. Though likely that candesartan is carrying out its blockade of AT₁, as supported by a wealth of evidence in other studies, it has not been specifically proven in my study.

Angiotensin II binds to angiotensin II type 1 receptors (AT₁), and through this has proinflammatory actions (Suzuki et al., 2003). AT₁ receptors are expressed in the substantia nigra and the striatum (Allen et al., 1992). Angiotensin II acts to stimulate NADPH oxidase to produce reactive oxygen species (ROS) (Griendling and Ushio-Fukai, 2000). Dysregulation of ROS results in oxidative stress, the production of toxic species including 4-hydroxynonenal (HNE), the oxidation of proteins, lipids, and DNA, and an environment generally toxic to cells (Jenner, 2003; Rodriguez-Pallares et al., 2008). It is thought that oxidative stress may play a role in Parkinson's disease whereby it results in an environment toxic to dopaminergic neurons, resulting in their death and the development of PD. Though oxidative stress is mainly linked to mitochondrial dysfunction, it may also be associated with dysfunction in the reninangiotensin system. An increase in AT₁ expression has been seen in response to α -synuclein as seen in the AAV9- α -synuclein animal model (Rodriguez-Perez et al., 2018), however this has not been identified in human PD. AT₁ serum autoantibodies (which act as agonists to AT₁ receptors) were seen to be increased in a group of people with PD compared to controls
(Labandeira et al., 2022). Therefore, the action of candesartan in blocking AT₁ receptors and reducing oxidative stress may explain how candesartan is having a protective effect on the survival of ChAT⁺ neurons in *Chapter 2.4.5* despite not altering levels of inflammation. A study into how candesartan and other AT₁ blockers affected the progression of pathology in an α -synuclein overexpression animal model showed that the blockage of AT₁ had a protective effect on the development of pathology (Rodriguez-Perez et al., 2018). This work looked into the anti-inflammatory action of candesartan – showing a decrease in MHC II-positive microglia and in pro-inflammatory cytokine expression in treated vs untreated α -synuclein overexpressing animals – however this was not investigated in regards to the involvement of TLRs. Furthermore, it was shown that the AT₁ blockers decreased the oxidative stress induced by α -synuclein overexpression.

3.5.3.1 Alternative TLR-mediated mechanism

In addition to their presence on immune cells, TLRs are also found on neurons (Barajon et al., 2009; Lafon et al., 2006; Ma et al., 2006; Tang et al., 2007; Zhou et al., 2009). In these cells the activation of TLRs results in different downstream effects; one such effect is in the process of autophagic flux. As discussed in Chapter 1.2.1.2, dysfunction of the autophagy-lysosome pathway is thought to aid α -synuclein propagation between cells through resulting in the increased exocytosis of α-synuclein (Alvarez-Erviti et al., 2011; Danzer et al., 2012; Lee et al., 2013; Poehler et al., 2014). Multiple studies have shown that the activation of neuronal TLRs results in the inhibition of autophagy. The activation of neuronal TLR2 (Chedid et al., 2022; Kim et al., 2021) and TLR4 (Chung et al., 2022) has been shown in vitro in SH-SY5Y neurons. When co-cultured with LPS-activated microglia (Chung et al., 2022), treated with a TLR2 agonist (Chedid et al., 2022), or with α -synuclein fibrils (Kim et al., 2021) an increase in the autophagic flux marker p62 was observed, indicative of a dampening of flux (Liu et al., 2016). The use of TLR2/4 inhibitors ameliorated this accumulation of p62 in addition to the build-up of phosphorylated α -synuclein, demonstrating a role for activated TLRs in the inhibition of autophagy. This inhibition of autophagy could also explain the increase in neuron-to-neuron transmission of α -synuclein as seen in work by Kim et al., (2015). Alterations to autophagic flux were similarly seen in microglia in a PFF-injected mouse brain. However, in this work TLR4 was crucial for autophago-lysosomal activity to occur and TLR4 deficient mice showed deficits in this activity (Venezia et al., 2017). These studies show that the suppression of TLRs can have effects independent of inflammation. It is possible that the protective effect of candesartan described in Chapter 2 is occurring through its action in

Chapter 3

suppressing the expression of neuronal TLRs, leading to increased autophagic flux. This would result in a decrease in α -synuclein accumulation, a decrease in the exocytosis of α -synuclein, and therefore a decrease in the pathological spread of α -synuclein through the brain. If true, this could explain the protective effect of candesartan treatment on the survival of the ChAT⁺ neurons. In my study, levels of TLRs 2 and 4 have only been quantified through western blotting. This does not allow the determination of cell-specific expression, and may be weighted towards microglial TLR expression due to their expressing higher levels of TLRs than neurons. This could be achieved through immunohistochemistry, however this was not possible in this study due to a failure to find any convincing antibodies staining specifically for TLR expression.

The work described in this chapter has conclusively shown that inflammation is not present within the brain in this animal model, and indicates that candesartan is having a minimal effect on brain inflammation, though whether this is due to the lack of PFF-induced neuroinflammation is unclear. Further work is needed to determine the protective mechanism by which candesartan is slowing the progression of disease pathology as shown in *Chapter 2*.

Chapter 4.

INVESTIGATING WHETHER BLOCKING TLRS ALTERS A-SYNUCLEIN-INDUCED INFLAMMATION IN PERIPHERAL IMMUNE CELLS OF PEOPLE WITH PARKINSON'S DISEASE

DECLARATION

Chapter 4.5.4 **"TLR blockers"** contains some text modified from the published review article **"Toll-like receptors and their therapeutic potential in Parkinson's disease and \alpha-synucleinopathies"**. This article was published in Brain, Behaviour, and Immunity: <u>https://doi.org/10.1016/j.bbi.2019.06.042</u> under Green open access with a 12-month embargo which has now lifted. The accepted manuscript is freely available to view within Apollo – the University of Cambridge repository at:

https://www.repository.cam.ac.uk/items/2abbd448-b6b5-4aa5-a22c-22279f75ecd9

The author contributions are as follows: I, Catherine Kennedy (née Horne), and Dr Antonina Kouli are joint first authors and equally contributed to the writing of the text. The figures were produced by Dr Antonina Kouli. Critical feedback was provided by Dr Caroline Williams-Gray in addition to the editing of the final manuscript.

The extraction of peripheral blood mononuclear cells (PBMCs) from whole blood was carried out by a combination of myself, Dr Antonina Kouli, Dr Jonathan Holbrook, and Lakmini Kahanawita. The blood was obtained from participants (both patients and controls) attending the Cambridge Parkinson's Disease Research Clinic at the John Van Geest Centre for Brain Repair in Cambridge. Participants had provided written consent for donation of their blood samples and use in this research (REC number 03/303). The clinic is run by Dr Caroline Williams-Gray.

The production of α-synuclein pre-formed fibrils was carried out by Dr Wei-Li Kuan.

Endotoxin testing was carried out by the LonzaTM Rapid Endotest service.

Super-resolution microscopy and analysis of the images taken was carried out by Dr Evgenia Lobanova.

4.1 Aims of chapter

Given that I was not able to demonstrate a clear mechanism by which candesartan was having a neuroprotective effect in the α -synuclein PFF animals, a key aim of this chapter was to determine whether candesartan blocks the inflammatory response to α -synuclein *in vitro*, using human peripheral immune cells.

Candesartan is reported to reduce the expression of both TLR2 and TLR4, and both receptors have been implicated in mediating the inflammatory response of immune cells to α -synuclein in model systems. However, a key question is whether the response in human PD is predominantly driven by the activation of TLR2 or TLR4. This has important implications for future therapeutic strategies targeting TLRs in PD. Though TLR-blockers have been used in various *in vivo* and *in vitro* studies relevant to PD (as reviewed in *Chapter 1.4*) there is very limited work investigating how they impact on immune responses using human cells, and in particular no studies have specifically investigated the effect of TLR4-specific blockers on human immune cell responses. In this chapter I aimed to optimise a protocol to study the contribution of both TLRs 2 and 4 to the α -synuclein-induced pro-inflammatory response of immune cells in PD, and to collect pilot data to inform future studies using cells from larger cohorts of PD patients and controls.

4.1.1 Objectives

- 1. To determine whether α -synuclein produces an inflammatory response in human PBMCs, and which species of α -synuclein produces the most robust response.
- 2. To investigate whether candesartan is able to block the inflammatory response to α -synuclein in ex vivo human cells
- 3. To investigate whether blockers specifically targeting TLR2, TLR4, or both receptors, are able to block the inflammatory response to α-synuclein.
- 4. To investigate the optimal experimental conditions for measuring the α -synuclein induced inflammatory response in human immune cells, and the effects of TLR blockade.

4.2 Introduction

In this chapter, work is being transitioned from animal models to models using human cells. Animal models are useful in researching organisms as a whole and allow the effects of an intervention to be monitored in a whole system, however they are limited in the extent to which they replicate the human disease. For this reason, in this chapter human peripheral blood mononuclear cells (PBMCs) were used to investigate TLR responses.

4.2.1 Monocytes

The use of PBMCs in this chapter is in part due to the growing interest in the role of the peripheral immune system in PD. PBMCs include monocyte cells; these are key cells of the innate immune system. There are three main subgroups of monocytes: classical, intermediate, and non-classical. Classical monocytes are enriched in the blood of people with PD compared to healthy controls (Grozdanov et al., 2014; Wijeyekoon et al., 2020). This has been associated with an increase in TREM2 which is thought to be involved in TLR stimulation, in TLRs 2 and 4 expression (Drouin-Ouellet et al., 2015; Wijeyekoon et al., 2020), and in LRRK2 levels which is thought to contribute to monocyte dysregulation (Bliederhaeuser et al., 2016). Further weight to monocytes playing a role in PD is provided by evidence for their having a distinct gene expression pattern in early stage PD compared to controls. Genes involved in immune activation including TNF-α and MyD88 (the TLR central adapter protein) were found to be differentially expressed (Schlachetzki et al., 2018). This increase in monocyte sub-populations and further dysregulation of monocytes and their expression profiles could result in an increased inflammatory response and could play an important role in the pathology of PD. In addition, monocytes are the predominant cell type expressing TLRs in the peripheral immune system; they express all TLRs with TLRs 1, 2, and 4 at the highest levels (Hornung et al., 2002). This makes them particularly useful in the study of PD where the mostly highly implicated TLRs are TLRs 2 and 4.

The use of PBMCs in this study is also in part due to their forming an easily accessible human cell model. The immune cells of potentially highest interest in PD are microglia,

Chapter 4

however these cells are difficult to obtain from humans to study *in vitro*. The only methods of observing microglia in situ are through neuroimaging techniques or through post-mortem studies. Neuroimaging allows observation of microglia at different stages of PD, and sequential neuroimaging allows the monitoring of changes in microglial activation over time. (Gerhard et al., 2006; Ouchi et al., 2005). However, these studies do not allow the investigation of individual microglia, or provide methods of studying mechanisms behind microglial activation. Post-mortem work allows the investigation of individual microglia, and some investigation into mechanisms of action. However, in these works microglia are static, and are often capturing late-stage PD. In vitro methods of investigating human microglia can involve extracting 1° human microglia, or through the use of iPSCs (induced pluripotent stem cells) produced from patient cells. These in vitro methods are very useful, however they can be complicated. Peripheral monocytes have some similarities with microglia, though they are ontogenetically distinct. These two cell types are indistinguishable in regards to morphology, can both form inflammatory macrophages, and express many of the same factors – such as Iba1. (Baufeld et al., 2018). I am utilising the similarities between these cell types so as to use monocytes as an easily accessible cell model of microglia. There are differences between the cell types, but it is being assumed that the main TLR-mediated inflammatory responses are similar, and may give us some insight into how microglia may respond to α -synuclein. Another method of modelling microglial responses is through the use of microglial cell lines such as the BV2 mouse microglia cell line. These have been used extensively by others, and, though useful, do not fulfil the remit of this chapter aiming to move towards the human disease.

For the experiments in this chapter, I aimed to stimulate the TLR pathway in human PBMCs/monocytes and used TNF- α as the key read-out due to its production by monocytes downstream of the TLR/NF- κ B activation pathway. The role of TNF- α has been particularly implicated in the aetiology and progression of Parkinson's. A study of patients with inflammatory bowel disease (IBD; a disease associated with increased risk of the development of PD) found that those IBD patients who had received anti-TNF therapy showed reduced PD incidence (Peter et al., 2018). In addition, anti-TNF treatments in animal models of PD have shown attenuation of the loss of dopaminergic neurons (Barnum et al., 2014; Harms et al., 2011; McCoy et al., 2008, 2006). From these studies it can be seen that the levels of cytokines, including TNF- α , in patients is important. This TNF- α production may occur due to the activation of TLRs as this cytokine is downstream of their NF- κ B activation pathway. The peripheral activation of TLRs 2 and 4 has been shown by Wijeyekoon et al., (2020) and Drouin-

Ouellet et al., (2015). Further research is required to determine how the inhibition of these TLRs may affect the release of TNF- α in human peripheral immune cells, and how this may differ in people with PD or healthy controls.

4.2.2 Toxic species of α-synuclein

Pathogenic forms of α -synuclein are thought to play a key role in activating innate immune cells via TLR 2 and 4, as reviewed in *Chapter 1.4*, but there is some uncertainty regarding which form is the most inflammatory.

The specific species of aggregated α -synuclein which exerts the highest proinflammatory effect is unknown. Recent work by Emin et al., (2022) showed that small soluble α -synuclein aggregates resulted in a higher pro-inflammatory response in BV2 microglia than larger aggregates. One major criticism of many studies is the lack of characterisation of aggregates. This study used a density centrifugation method to separate a heterogeneous mixture of lab-generated aggregate species, and then characterised the species using high resolution microscopy, providing an accurate characterisation of exactly what was being added to the cultured BV2 cells. Additionally, in this study soluble α -synuclein aggregates from soaked PD and healthy control brains were characterized. These aggregates were extracted utilising a technique produced by Hong et al., (2018) whereby post-mortem brain is soaked to extract the soluble oligometric species. PD brains had a higher proportion of small aggregates compared to the controls, and these small aggregates were more inflammatory to BV2 microglia than the larger aggregates extracted from control brains. Hughes et al., (2019) also showed that oligometric α -synuclein treatment resulted in higher pro-inflammatory effects than fibrils in the BV2 microglial cell line. These oligomers were produced using simple incubation with shaking methods. In contradictory work, also using BV2 microglia, Hoffmann et al., (2016) showed that fibrillar α -synuclein had an increased pro-inflammatory effect on these microglia compared to monomeric or oligomeric species. The production of oligomers in this study was carried out using incubation with 4-hydroxy-2-nonenal (HNE). HNE induces oligometisation through covalent modification of α -synuclein, and these oligometis differ in conformational properties to fibrils (Bae et al., 2013). Furthermore, a comprehensive study by Grozdanov et al., (2019) likewise showed that fibrils induce increased inflammation compared to other α -synuclein species. This study used a series of well-categorised recombinant α synuclein species including multiple types of oligomeric species (both small homogenous and larger heterogeneous species), fibrillar species (fibrils, ribbons, and those produced at differing

pHs), and mutated monomeric species. These were used to stimulate healthy control peripheral blood monocytes and BV2 microglia.

The pro-inflammatory nature of α -synuclein has thus been well-established, however further work is required to determine the most pro-inflammatory species.

4.2.3 TLR blockers

Candesartan has been repurposed as a TLR blocker through its reduction of the expression of both TLRs 2 and 4 (Dasu et al., 2009). The blockade of the α -synuclein-induced inflammatory response by candesartan has been investigated by Daniele et al., (2015), however this was carried out in primary mouse microglia. The effectiveness of candesartan at blocking the response of human immune cells to α -synuclein has not been established. This work also used the TLR2 specific blocker CU-CPT22. CU-CPT22 is a small molecule inhibitor, which acts to inhibit the TLR1/TLR2 heterodimer through interaction with the extracellular domain of the TLR (Cheng et al., 2012). Through this they showed that the α -synuclein-induced inflammatory response was dependent on TLR2 signalling. However, TLR4 was not investigated in this work, and so although a TLR2-dependent response is seen, the contribution of TLR4 is unknown. Others, including Hughes et al., (2019) have shown that the inflammatory response of microglia (in this case the BV2 mouse microglial cell line) is TLR4-dependent. This work was carried out using TAK-242, another small molecule inhibitor. It works to suppress TLR4 activation via its action on the intracellular domains of the TLR - specifically the TIR domain – through which it inhibits signal conduction (Ii et al., 2006; Kawamoto et al., 2008). These two studies, due to their only investigating either TLR2 or TLR4, have not fully established which TLR is the most essential to the α -synuclein-induced inflammatory response, and neither have observed this in humans. The use of candesartan in this chapter will allow the inhibition of expression of both TLRs 2 and 4. The use of CU-CPT22 and TAK-242 will allow the determination of which TLR blockade is most effective at attenuating an α-synucleininduced response.

4.3 Methods

4.3.1 PBMC extraction

Venous blood was collected from participants (both patients and controls) attending the Cambridge Parkinson's Disease Research Clinic at the John Van Geest Centre for Brain Repair in Cambridge. Participants had provided written consent for donation of their blood samples and use in this research (REC number 03/303). Blood was collected in EDTA⁺ tubes to prevent clotting. The blood was mixed with sterile PBS 1:1 and added to SepMateTM tubes containing Ficoll. These were then spun at 120000 g for 10 min to separate the PBMCs. The PBMCs were washed in PBS and spun twice before being resuspended in 10 ml PBS and counted. The cells were then centrifuged and resuspended in 90 % FBS/10% DMSO before being quickly transferred to the -80°C freezer initially, and then to long-term storage in liquid nitrogen. Cells were extracted either by myself or by a research associate within the lab group.

4.3.2 PBMC thawing

Cryovials were thawed and immediately diluted in warmed RPMI media supplemented with foetal bovine serum and pen-strep. The cells were then spun at 400 g for 5 min, the supernatant discarded, and the cells resuspended in warm culture medium. This process was carried out very quickly to ensure that the cells were left in toxic DMSO for as little time as possible. The cells were counted before being centrifuged again and resuspended to make a solution of 1 million cells/ml.

4.3.3 Monocyte separation

Monocyte separation was carried out using a kit from Miltenyi Biotec (*Cat. No. 130-050-201*). The kit protocol was used. Briefly, after thawing, cells were counted and spun at 300 g and 4°C for 10 min before being resuspended in MACS buffer. The correct volume of CD14 beads according to PBMC number (20 μ l per 10⁷ cells) was added and the solution incubated for 15 min at 4°C. The column was placed into the magnetic field of a MACS separator and washed before the cell suspension was added. Unlabelled cells passed through the column and

the column was washed to ensure only labelled cells remained. The unlabelled cells were discarded. The column was removed from the magnetic field and the magnetically labelled cells eluted using a plunger. The cells were then counted before use in the assay.

4.3.4 Stimulation assays

PBMCs or monocytes were plated in flat-bottomed 96-well plates with 100,000 cells per well and left to rest for 2 hours in an incubator at 5% CO₂ and 37° C.

4.3.4.1 Stimulation experiments

The cells were stimulated with either LPS (10 ng/ml), α -synuclein monomers (concentration dependent on oligomeric concentration in specific assay; rPeptide), α -synuclein oligomers (concentration between 0.5 μ M and 5 μ M and optimised as described later; see below for method of production), or α -synuclein PFFs (4 μ M; see below for method of production). These were then left for 24 hours, 48 hours, or 72 hours in a 37°C incubator. The supernatant was removed into a v-bottomed 96-well plate. This was spun at 300g for 5 minutes to pellet the cells. The supernatant was then removed, aliquotted, and frozen at -80°C for future ELISA experiments.

4.3.4.2 TLR blockade experiments

After the cells were plated and left to rest for 2 hours, either Candesartan (concentration between 5 μ M and 30 μ M and optimised as later described; ThermoFisher *Cat. No.* 466190010), TAK-242 (1 μ M; Tocris Biosciences *Cat. No.* 6587), or Cu-CUPT22 (concentration between 5 μ M and 30 μ M and optimised as later described; Tocris Biosciences *Cat. No.* 4884) was added, or the cells were left untreated. All drugs were resuspended in DMSO according to the manufacturer's instructions. The cells were treated with blocking agents alone (or left untreated) for 24 hours before the addition of stimulants. The stimulation and supernatant collection were carried out as described above.

4.3.5 Preparation of α-synuclein pre-formed fibrils

A-synuclein pre-formed fibrils were prepared as previously described in the literature (Volpicelli-Daley et al., 2014). A-synuclein monomers (rPeptide) were resuspended in 10 mM Tris-HCl, 50 mM NaCl, pH 7.6 at a concentration of 5 mg/ml. A 37°C thermomixer was used to shake them for 7 days at 1000 r.p.m., they were then aliquoted and transferred to a -80°C

freezer. Prior to use the PFFs were thawed and diluted to 0.1 μ g/ μ l and sonicated using a Soniprep 150 Plus ultrasonic disintegrator (MSE) with 60 pulses at 10% power with a pause every 10 pulses to allow the solution to cool.

4.3.6 Preparation of α-synuclein oligomers

A-synuclein oligomers were prepared using a technique developed and characterised by the Klenerman group (Department of Chemistry, University of Cambridge). A-synuclein monomers (rPeptide) were resuspended in PBS at a concentration of 70 μ M. A 37°C thermomixer was used to shake them for 6 hours at 200 r.p.m., they were then either kept at 4°C for up to 24 hours before use or were aliquoted and transferred to a -80°C freezer. For use in stimulation assays, the oligomers were thawed (if frozen) and diluted to the required concentration directly in the well containing the media-PBMC solution.

4.3.7 Endotoxin testing

The α -synuclein monomers obtained from rPeptide were tested for endotoxin using the LonzaTM Rapid Endotest service. This service used a Kinetic-QCLTM LAL Assay. Briefly, the chromogenic substrate (consisting of the LAL reagent – Limulus amebocyte lysate; reacts to bacterial endotoxin and LPS – and a chromogenic reagent) is added to the test sample. This will turn yellow after an amount of time inversely proportional to the amount of endotoxin in the sample. The colour change is monitored using an incubating absorbance plate reader measuring absorbance at 405 nm. The amount of endotoxin can be measured through comparison to a standard curve. (Methods obtained from the Lonza webpage for the Kinetic-QCLTM Kinetic Chromogenic LAL Assay; all aspects of the testing process were carried out by Lonza and not by me).

4.3.8 Super-resolution microscopy

All super resolution microscopy was carried out by Dr Evgenia Lobanova. For full methods see Emin et al., (2022). Briefly, the anti- α -synuclein capture antibody (MJFR-14-6-4-2; Abcam; *ab227047*) diluted to 10 nM was added to the prepared coverslip and incubated for 10 min. The coverslip was washed and then the samples incubated for an hour. The detection antibody (MJFR-14-6-4-2-647; Abcam; *ab216309*) was incubated for up to 30 min. The coverslip was washed and then imaged. STORM microscopy was used for super-resolution microscopy; 8000 frames of 15 ms each.

4.3.9 Enzyme-linked immunosorbent assays (ELISA)

The levels of TNF- α and IL-1 β in cell culture supernatants were quantified using ELISAs. Two different kits were used due the failure of a kit partway through experimental process, these were the Invitrogen kit *Cat No. CHC1753* (TNF- α) or *CHC1213* (IL-1 β) and the R&D Systems kit *Cat No. DY210* (TNF- α) or *DY201* (IL-1 β). Optical density was measured at 450 nm using the FLUOstar Omega plate reader. Each standard and sample were measured in 2 adjacent replicates and an average value calculated. The measured optical density was plotted against the known concentration of the standards to generate a 4 parameter curve logistic curve fit using the *MARS* data analysis software.

4.3.9.1 ELISA kit failure

Having successfully used the Invitrogen kit for many experiments, as indicated above, over several months, it stopped worked partway through the optimisation of the oligomer concentration gradients. The nature of the failure was unknown. Despite many repeats using new kits from the same supplier, and direct comparison of old and new standards, the standard curves were not developing. A subset of images of these plates is included in **Figure 4.1**. After multiple plate failures, wastage of samples, and communication with the manufacturer it was decided to try a new manufacturer – R&D systems.



Figure 4.1 Failed ELISA plates using the Invitrogen kit. Standard curve in duplicate is in the first two columns on the left of each plate, remainder of wells are samples in duplicate. The plate on the bottom right was photographed after adding 2M sulphuric acid to stop the colour change reaction, hence the yellow colour, the other plates were photographed before the addition of the sulphuric acid.

4.3.9.2 Dilutions in ELISA

The most effective dilution of supernatant to use in the assay was dependent on the kit used and on the individual being tested. When using the Invitrogen kit a dilution of 1:8 was used. This dilution was recommended by the manufacturer and had been tested by others in the lab prior to the work carried out here. On the transition from the Invitrogen kit to the new kit from R&D systems, it was found that a 1:8 dilution caused plate saturation (**Figure 4.2aI**). Samples were diluted to 1:16 and also resulted in complete plate saturation (**Figure 4.2aII**). A concentration gradient was then carried out from 1:32 down to 1:128 (**Figure 4.2aIII**). These dilutions resulted in measurements of TNF- α well-within the standard curve. In later experiments, with various individual's cells, the results were below the level of detection. It was decided that the dilution gradient needed repeating to ensure that results were not being missed through samples being over-diluted. This was repeated first with two subjects' cells and surprisingly it was found that TNF- α produced in response to all stimulants was well below the

top of the standard curve but still not at an acceptable level even in response to LPS which would be expected to result in a much higher response (**Figure 4.2bI**). This was then repeated with more samples to ensure that this wasn't solely the response of one outlying sample. In all 4 of these samples the response was substantially below the top of the standard curve even at a 1:8 dilution (**Figure 4.2bII**). These experiments showed unexpected results where the monomers/oligomers barely showed a response above the unstimulated, and even the LPS did not have as high a response as had been seen before. For this reason, a final dilution gradient was carried out with new batches of monomers, oligomers, and LPS. This showed a response similar to that seen before, and even with these high responses they remained sufficiently below the standard curve at a 1:8 dilution (**Figure 4.2bIII**). It was decided to use the 1:8 dilution from this point forward as there was the concern that, though the responses were well below the standard curve, diluting even less may result in the complete saturation of some individuals due to the wide range of individual responses.



Figure 4.2 Dilution gradients of various subject's cells. (a) Plate saturation at (I) 1:8 dilution (N=2); (II) 1;16 dilution (N=2); (III) a partial dilution gradient of 1:32-1:128 (N=4). (b) Full dilution gradient of 1:8-1:128. (I) N=2; (II) N=4; (III) N=4. Dotted line represents top of standard curve for each assay.

4.4 Results

4.4.1 Monocytes show a high basal level of inflammation

I originally planned to use MACS separated monocytes in my stimulation experiments. However my preliminary experiments showed a high level of basal inflammation in unstimulated cultured monocytes, which might mask any subtle stimulatory effects induced by α-synuclein. The reason for this basal inflammation was not clear so work was carried out to establish whether this was occurring due to stress induced by the cell thawing process, or due to the monocyte cell separation process. A comparison between both fresh and frozen/thawed monocytes and PBMCs was carried out (Figure 4.3a-b). These cells were all taken from the same individual and processed at the same time. Fresh monocytes showed increased levels of basal TNF- α production compared to fresh PBMCs, potentially indicating that the cell separation process induces cell stress and cytokine production. The frozen/thawed monocytes showed the highest basal levels of TNF- α (Figure 4.3c). This indicates that the thawing process followed by the cell separation process stresses the cells despite resting periods. In contrast, basal TNF-a production in PBMCs was low, and did not differ when the PBMCs had been frozen and thawed before culture compared to fresh cells. The responses to stimulation with LPS, a-synuclein, and PFFs was also compared between fresh and frozen/thawed monocytes and PBMCs and showed a TNF- α response in all cell types. The response of the monocytes to α -synuclein stimulation does not appear to exceed the unstimulated response, whereas it does in whole PBMCs. From these experiments, I decided to use whole PBMCs for further stimulation experiments. Frozen PBMCs were used given that it was not practical to obtain fresh blood from patients every time an assay was carried out, and these preliminary experiments did not suggest an inferior response in frozen/thawed PBMCs.



Figure 4.3 TNF- α concentration in supernatant of fresh or frozen cells in response to stimulation. (a) Fresh and (b) frozen monocyte (left) and PBMC (right) response to stimulation with LPS and different forms of α -synuclein after having been stimulated for 24 hours, 48 hours, or 72 hours. (c) Conflated unstimulated cell comparison. (N=1). Average CV=3.80%.

4.4.2 The α-synuclein used is endotoxin-free

A sample of the α -synuclein being used in this study in addition to a sample of media was sent off to an external company (*Lonza*) for high sensitivity endotoxin testing. The samples being endotoxin free is very important as previous work from our group has shown that even very low levels of endotoxin (1.6 EU/ml) induced a similar pro-inflammatory effect to α - synuclein (White et al., 2018). Measurements carried out by *Lonza* showed that levels in both the monomeric α -synuclein and media alone contained <0.05 EU/ml endotoxin (the lower level of detection of their assay). The monomeric α -synuclein was the same product used to create the further aggregated species of α -synuclein (both oligomers and PFFs). This was carried out under sterile conditions – i.e. using sterile equipment and in an extraction hood – so as to avoid endotoxin contamination. Due to the unfeasibility of sending off a sample every time a new batch was made, it was assumed that the aggregated products would have similar levels of endotoxin to the monomeric species tested.

4.4.3 PBMCs are responsive to monomeric and oligomeric α-synuclein

I investigated the response of PBMCs to different forms of α-synuclein compared to LPS as a positive control, and compared different durations of stimulation. The main species of interest in this study were oligomers due to a prior hypothesis by our collaborator Dr Derya Emin in Professor David Klenerman's lab where she showed that oligomers are the most inflammatory species (Emin et al., 2022). For this reason oligomers were used at different concentrations in these stimulation assays and resulted in a dose-dependent TNF-α response. Oligomer concentration was measured as the starting concentration of monomers in the preparation of the oligomers. At the highest concentration the oligomers resulted in an inflammatory response comparable to LPS. However even at the high concentration of 5 µM (concentration measured as starting concentration of monomers in preparation) the PFFs did not cause a significant inflammatory response. Surprisingly, the monomeric α -synuclein (4) μ M) also caused an inflammatory response comparable in magnitude to LPS and oligomers. (Figure 4.4a). Different durations of stimulation were assessed to determine the optimal timecourse of a-synuclein-induced inflammation. A robust inflammatory response to asynuclein was seen at 24 hours post-stimulation, this was maintained at 48 hours, but had started to decline after 72 hours. (Figure 4.4a). 24 hours was chosen for use in further experiments due to observing an optimal α -synuclein response, and due to it being the most practical of the timepoints. This experiment was carried out using samples from both PD and controls to investigate whether there were any differences in the cells presumed to have had prior exposure to pathological forms of α -synuclein (i.e. those from people with PD) and cells presumed not to have had this exposure (i.e. those from controls), and the data is presented stratified according to group in Figure 4.4b. No significant differences were seen between patient and control groups (Two-way ANOVA followed by Bonferroni's multiple comparison test; p>0.99), though there was a consistent trend towards a decreased response in patients with each stimulant and at each timepoint. (**Figure 4.4b**).



Figure 4.4 TNF- α concentration in supernatant of PBMCs in response to stimulation. (a) PBMC response to stimulation with LPS and different forms of α -synuclein after having been stimulated for 24 hours, 48 hours, or 72 hours. (N=8) (b) Stratification of (a) into PD (N=4) and control (N=4) responses at 24, 48, and 72 hours of stimulation. Average CV=3.73%.

The high response of monomeric α -synuclein was surprising though not entirely unprecedented as an inflammatory response to monomers has been seen by Nissen et al., (2019) and White et al., (2018), however several previous studies carrying out similar work (using cultured PBMCs/monocytes/BV2 microglial cell line/THP-1 monocytic cell line) show a monomer-induced response similar to baseline unstimulated levels (Bearoff et al., 2023; Codolo et al., 2013; Grozdanov et al., 2019; Hughes et al., 2019; Pike et al., 2021; Trudler et al., 2021). One hypothesis was that the monomers were aggregating in the culture medium to form oligomers which were more inflammatory. To investigate this, I sent the supernatant from unstimulated cells, from monomer-stimulated cells (at 24 hours and 48 hours of stimulation), and from PFF-stimulated cells to our collaborator Dr Evgenia Lobanova in Professor David Klenerman's lab for single molecule imaging using super resolution microscopy (N=1; 4 images were analysed from each sample) (Figure 4.5). This work (carried out and images produced by Dr Lobanova) showed that the culture medium from PBMCs stimulated with monomer for 24 hours showed an increased proportion, and an increased total number, of larger aggregates (length >150 nm) than the supernatant from unstimulated cells (Figure 4.5b). Furthermore, when the culture is prolonged for an extra 24 hours (48 hours total) there is an even higher proportion, and total number, of larger aggregates in the medium. As expected, there is a much higher fraction of these larger aggregates in PFFs which have been deliberately aggregated. (Figure 4.5b). These results suggest that monomers are aggregating in solution, and this could explain why the inflammatory reaction of cells to this incubation with supposed monomers is higher than expected and comparable to incubation with oligomers. On the basis of these experiments, the optimal form of α -synuclein for future experiments is oligomeric α -synuclein and not PFFs. Monomeric α -synuclein will still be used but not as a negative control as was originally intended. The optimal timecourse is 24 hours as the inflammatory response is at a high level and has not begun to decline.

Chapter 4



Figure 4.5 Super resolution microscopy and SiMPull of a-synuclein aggregates in supernatant from stimulated cells. (a) Super resolution microscopy images of a-synuclein aggregates in supernatant from (I) unstimulated cells; (II) cells stimulated with PFFs for 24 hours; (III); cells stimulated with monomer for 24 hours; (IV) cells stimulated with monomer for 48 hours. (b) Quantification of (left) the percentage of total aggregates with length >150nm and (right) the total number of aggregates with length >150nm. N=1; 4 images were analysed for each sample.

4.4.4 PBMCs respond to oligomeric α-synuclein in a dose-dependent manner

Further experiments were carried out to establish a dose-response relationship for oligomeric α -synuclein and to confirm whether this was consistent across subjects in a larger group. All participant PBMCs responded in a dose-dependent manner, though some had a much stronger response than others. Both patients and controls responded in a similar manner. The optimal concentration was determined to be 4 μ M of oligomers; at this concentration all participant cells produced an inflammatory response. This was seen for both TNF- α (**Figure 4.6a**) and IL-1 β (**Figure 4.6b**). These experiments showed a consistently robust TNF- α response, however the IL-1 β response was less robust and less consistent, and even at a concentration of 5 μ M the response was not high. For this reason, it was decided to focus on the TNF- α inflammatory response in the following experiments.



Figure 4.6 Inflammatory response on stimulation with oligomer concentration gradient. (a) $TNF-\alpha$. (PD are labelled PDPxxxx, N=6; control are labelled Cxxx, N=4). (b) IL-1 β (PD N=3; control N=2). Average TNF- α CV=2.83%; IL-1 β CV=2.15%.

4.4.5 The inflammatory response is TLR4 but not TLR2- dependent

Treatment of PBMCs with the specific TLR4 blocker TAK-242 for 24 hours prior to stimulation resulted in a significant decrease in TNF α release on cell stimulation with LPS, α -synuclein monomers, and α -synuclein oligomers (N=4 participant samples; two-way ANOVA followed by Tukey's posthoc test; p<0.001). (**Figure 4.7a**). The concentration of TAK-242 used (1 μ M) was chosen based on work by Hughes et al., (2019). In a second experiment (N=3 participant samples), I compared the PBMC response to stimulation following treatment for 24 hours with the TLR2-specific blocker Cu-cpt22, and with candesartan which, as discussed in

Chapter 2.2.2, has been reported to decreases the expression of both TLRs 2 and 4. Both TAK-242 and Candesartan fully abrogated the TNF- α release in response to cell stimulation with LPS (Two-way ANOVA followed by Tukey's posthoc test; p<0.0001). Cu-cpt22 did not have a significant effect on LPS-induced TNF- α release. (Figure 4.7b). The monomers and oligomers used in this experiment did not have the expected effect of inducing TNF- α release, so it was not possible to assess the effects of the drug treatments on α -synuclein-induced inflammation in this particular experiment.



Figure 4.7 Effect of TLR blocking treatment on TNF- α release from stimulated cells. (a) Untreated and TLR4-blocker TAK-242-treated (N=4; Average CV=6.13%) and (b) untreated, TAK-242, TLR2/4-blocker candesartan, and TLR2-blocker Cu-cpt22-treated (N=3; Average CV=3.55%)) PBMC's response to LPS and different forms of α -synuclein. Two-way ANOVA followed by Tukey's posthoc test; $p^{***}<0.001$, $p^{****}<0.0001$.

The above experiment (**Figure 4.7b**) was repeated again with a new batch of monomers and oligomers which triggered an immune response. TAK-242 consistently supressed the immune response to LPS, and to monomeric and oligomeric α -synuclein. (**Figure 4.8a**). A candesartan concentration gradient was carried out to assess the lowest concentration at which it was effective at blocking the inflammatory response. This was carried out from both an assay optimisation standpoint, and also so that concentrations used here could be translated into use in a drug repurposing clinical trial at a later date. The gradient was chosen according to concentrations used successfully in the literature. Multiple studies used candesartan at 1 μ M (Dasu et al., 2009; Qie et al., 2020), however a preliminary experiment using this concentration did not result in any inflammatory effects (data not shown). Daniele et al., (2015) used candesartan up to 30 μ M and so a similar range was used here. A clear dose-dependent inhibitory effect on TNF- α was seen. TNF- α levels were significantly decreased from those

released from untreated LPS-stimulated cells when treated with a concentration of 10 μ M candesartan, and this response was fully abrogated at a candesartan dose of 20 μ M.(Two-way ANOVA followed by Tukey's post-hoc test; p<0.05; p<0.001). (**Figure 4.8b**). Further work was carried out to assess whether the failure of action of Cu-cpt22 was due to it being used at a suboptimal concentration – the original concentration of 10 μ M having been used by Daniele et al., (2015) on primary microglia. A concentration gradient revealed that treatment with Cu-cpt22 across a range of concentrations did not result in a decrease in TNF- α release on cell stimulation with either LPS or α -synuclein. (**Figure 4.8c**). The LPS -stimulated Cu-cpt22-treated cells produced increased TNF- α compared to LPS stimulated but untreated cells, this difference reached significance at a concentration of 10 μ M Cu-cpt22 (p>0.001) but there did not appear to be a dose-dependent effect above this concentration. Though LPS is thought to have some stimulatory action on TLR2 in addition to its main receptor, TLR4, in these experiments it was clear that the TLR2 blocker Cu-cpt22 did not block the TNF- α response to LPS and may in fact have caused a compensatory upregulation of TLR4 to make up for the loss of TLR2, resulting in increased TNF- α release.



Figure 4.8 Effect of TLR blocking treatment on TNF-a release from stimulated cells. (a) Comparison between untreated and TAK-242-treated stimulated PBMCs. (N=4). (b) Comparison between untreated and a concentration gradient of candesartan-treated stimulated PBMCs. (N=4). (c) Comparison between untreated and a concentration gradient of Cu-cpt22-treated stimulated PBMCs. (N=4). Two-way ANOVA followed by Tukey's posthoc test; p<0.05, p***<0.001, p****<0.0001. Average CV=2.87%.

4.5 Discussion

4.5.1 Summary

In this chapter I sought to identify whether human peripheral immune cells responded to different forms of α -synuclein, and whether blocking TLRs would inhibit inflammatory responses to α -synuclein in human immune cells. This question has obvious relevance to Parkinson's irrespective of whether the cells are derived from patients or controls given that α synuclein is the key pathogenic protein in PD. However, I had also hoped to extend this work into a larger study comparing the responses of PBMCs from a cohort of PD patients versus age-matched healthy controls. Due to a variety of unavoidable circumstances, including kit failures, Covid and Brexit-related reagent supply issues, and a large variance in individual response to stimulants, I was not able to progress to a patient versus control study within the time available for my PhD. However I have made a number of valuable observations which will inform future work. These include:

- 1. High basal levels of inflammation occur in human monocyte preparations and may confound the response to exogenous stimulation hence PBMCs may be a better choice of cell type for this work.
- 2. Using frozen/thawed PBMCs does not significantly affect the inflammatory (TNF- α) response to stimulation.
- 3. Careful preparation of solutions allows the α -synuclein response to not be confounded by endotoxin, and so unexpected results can be trusted to be legitimate.
- 4. Oligomers are the optimal stimulant, and show consistent robust response at a concentration of 4 μ M.
- 5. The optimal timecourse for stimulation is 24 hours.
- The inflammatory response to LPS and to α-synuclein monomers and oligomers is TLR4-dependent.
- 7. Candesartan effectively blocks the inflammatory response to α -synuclein, this is seen in human immune cells despite the difficulty in showing this occurring in the α -synuclein animal model in *Chapter 3*.

This work was carried out on PBMCs from individuals with PD, as well as controls. The results suggest that PBMCs from both controls and PD patients respond in a proinflammatory fashion to stimulation with both monomeric and oligomeric α -synuclein as shown through their increased production of TNF- α . Whilst the work was underpowered to compare PD versus control responses, this preliminary work suggests that the observed peripheral inflammation measured in PD cohorts might relate to extrinsic factors (e.g. the presence of pathogenic α -synuclein) rather than inherent differences in the function of innate immune cells in the disease state.

In summary, these results show a positive impact of both candesartan and TAK242 on controlling the α -synuclein induced inflammation in human peripheral immune cells. This work has optimised conditions for stimulating and blocking TLR pathways in these cells. As a result, this paves the way for a larger project investigating TLR-dependent innate immune responses to α -synuclein in a large cohort of PD and controls.

4.5.2 Assay failures

4.5.2.1 Kit failure

The initial work carried out in this chapter was using a Thermofisher ELISA kit, however partway through experimentation this kit stopped working – specifically the standards consistently failed to develop. Though initially it was thought that there was a problem with the standards themselves, repetition of the experiment multiple times using a new kit sent by Thermofisher (with a different lot number) continually produced no or very faint standards despite the protocol being followed to the letter, and being carried out using the same reagents and equipment that had previously worked well. This problem was not resolved despite discussion with the manufacturer. It was decided that an aspect within the kit had been changed (unconfirmed by the manufacturer). This was most likely the capture antibody as in part of the troubleshooting the standards and detection antibodies from previous working kits were used and still failed to result in standard development, unfortunately there was no remaining capture antibody from a previous kit to test. Alternatively this could be due to a change in the plate being used – to the best of my knowledge the supplier of these was consistent, however it was noted that between the working plates and the failed plates a new box of plates had been opened. It is possible that a change to the plate resulted in a failure of the capture antibody binding, and so resulting in a failure of all standards – even those which had worked before.

This problem was eventually surmounted through the use of an ELISA kit from a different manufacturer - R&D systems. This unfortunately resulted in a number of experiments being aborted, wastage of samples and reagents, and the repetition of optimisation steps due to having to use a new kit, and consequently to large delays to the project.

4.5.2.2 Variability

In addition to the failure of kits, there was very large variability between the responses observed using different subject's cells. A degree of variability was expected, however the extent to which this occurred was not foreseen. This made it difficult to determine the optimal dilution to use to allow for the detection of very low levels of TNF- α , and the prevention of the saturation of very high levels of TNF- α . A dilution of 1:8 was settled on, however it cannot be guaranteed that in any further work saturation will not occur with some particularly reactive subject cells. Other studies, including Nissen et al., (2019) and White et al., (2018) have similarly seen large variability in the inflammatory responses of human PBMCs. Grozdanov et al., (2019) appeared to show less variability in levels of cytokine production in response to different forms of α -synuclein, this could be due to the presentation of data as fold increase over LPS. This may have been carried out to remove the large inter-subject variability and could be considered in any future work continuing my project.

The variability seen by myself, White, and Nissen may result from the variability in the ages of individuals included in the studies. White et al., (2018) used a cohort comprised of early stage PD patients with an age range of 55-80 years old. The Nissen et al., (2019) PD cohort comprised an age range of 43-79 years old. My own work also used a large range of ages (44-85 years old) as, due to the preliminary nature of the work carried out here, a specific cohort had not been chosen. Ageing is known to result in immunosenescence leading to immune deficiency, a decline in the immune response, and increased susceptibility to infection (Castle, 2000). This has been investigated using whole PBMCs and monocytes derived from elderly people. An impaired immune response of peripheral monocytes in aged subjects is seen by Bruunsgaard et al., (1999), Delpedro et al., (1998), and Nyugen et al., (2010) where a deficit in the induction of pro-inflammatory cytokines is seen in response to LPS and other TLR agonists. This impaired immune response in elderly subjects was similarly seen by Metcalf et al., (2015) in whole PBMCs. Contradictorily, others have shown an increased production of cytokines from aged compared to young subjects. (Fagiolo et al., 1993; Pinke et al., 2013; Roubenoff et al., 1998). Though a conclusive result has not been found, it is clear that ageing

Chapter 4

affects the immune reaction of cells, and that this results in differing levels of cytokine production. This could explain the large variation in TNF- α seen in my subjects on stimulation, and emphasises the importance of a carefully selected and age-matched cohort in future work.

The disease duration of people with PD may also have an effect on the inflammatory response of cells. This is controlled for in the White et al., (2018) paper (with a mean disease duration of 4.3 years), however in the Nissen et al., (2019) cohort the disease duration ranges from 3-13 years. In my work there is also a spread in disease duration (0.21-5.84 years). Studies are indicative of alterations in immune function as disease progresses. Mixed data is published on whether correlations between the production of pro-inflammatory cytokines and Hoehn-Yahr stage/UPDRS III score are positive or negative (Hasegawa et al., 2000; Reale et al., 2009). However, progression through H&Y or UPDRS III does not occur at the same rate in all people with PD and so is not directly linked to disease duration. Work by Devos et al., (2013) looked specifically at disease duration and found a negative correlation with colon mRNA levels of pro-inflammatory cytokines. This evidently does not directly correspond to the behaviour of PBMCs, however does provide more evidence that peripheral inflammation is altered in PD, and may be dependent specifically on disease duration. Further indications of alterations in immune function through disease duration arise from studies identifying an increase in autoantibodies to α -synuclein in early PD which declines as the disease progresses (Scott et al., 2018). Differences in the immune response, and specifically in the PBMC response, correlating to disease duration are less conclusive than changes seen with age. However, there is evidence that this altered immune response may occur, and as a result disease duration is a further factor that should be considered when selecting a cohort for future work.

A further factor which it is becoming increasingly clear must be controlled for is sex. Sex differences in both innate and adaptive immune response are well established, with females mounting stronger immune responses than males. (Klein and Flanagan, 2016). These immune differences can also be seen in a disease state. Nissen et al., (2019) found that immune responses within PD were altered dependent on sex, whereby some differences in immune function observed between controls and PD were seen only in males.

There are many other factors which may result in high variability in the response of PBMCs to stimulation. These include the health of the subject when the blood was taken, the vaccination status of the subject, any comorbidities, and the method of freezing down and then thawing the PBMCs – this can cause cell stress. These factors should all be taken into account

in an effort to control variability. However, even controlling for all of these factors, heterogeneity in human cell responses should be expected and many intrinsic factors will contribute to the response of PBMCs. The aim with this work is to explore whether in spite of the heterogeneity in human cell responses that when comparing between factors such as patient groups and cell treatment groups there is sufficient signal to rise above any of the noise from the heterogeneity.

4.5.2.3 Failure in oligomer production

Throughout the period of experimentation there were many issues with the assays failing to produce consistent responses to monomers/oligomers. Initial work showed a consistently high inflammatory response to both monomers and oligomers, this occurred in all individuals tested, though a large variability in the degree of response was seen – as discussed above. The oligomers used in this work were produced in accordance with the protocol developed by the Klenerman group (Department of Chemistry, University of Cambridge; collaborators of the Williams-Gray group). This was an updated protocol of that used by Hughes et al., (2019). It is a very simple protocol and initially produced consistent results between multiple separate monomer aggregations. Unfortunately, this consistency did not continue and stimulation experiments were repeated with PBMCs from the same individual/batch and I found that subjects who had previously produced a strong response to oligomers now produced little/no response to the new batch of oligomers, even though all other experimental conditions were identical (**Supplementary Figure 4.1**). This showed that these later failures were oligomer dependent.

The production of oligomers, though carried out in a regulated fashion, does not produce exactly the same combination of sizes of oligomers every time due to the random nature of aggregation. This could result in one batch producing more pro-inflammatory species than another. This may explain some of the differences in response to oligomers between experiments. A further reason for differences may be due to freezing/thawing of the oligomers. Based on initial variability in oligomer response it was decided to produce a large batch of oligomers and freeze them in aliquots so that the differences in oligomer size between batches could be controlled for. However, it was found that the inflammatory response to oligomers was attenuated if not entirely abrogated after the freeze/thaw process (**Figure 4.7b**). When the oligomers did not undergo freezing/thawing the PBMC inflammatory response to them was

175

retained (**Figure 4.8**). Future studies should carefully consider how to control for differences in oligomer species between batches.

4.5.3 The inflammatory response to monomers versus oligomers

There is some controversy over a-synuclein monomer-induced PBMC/monocyte inflammatory responses with much of the literature suggesting that oligomers are the more inflammatory species. Conflicting studies arise around this topic. Papers such as White et al., (2018) and Nissen et al., (2019) show an inflammatory response to monomers from PBMCs or monocytes, similar to that seen with aggregated α -synuclein. Conversely, (Bearoff et al., 2023; Codolo et al., 2013; Grozdanov et al., 2019; Hughes et al., 2019; Pike et al., 2021; Trudler et al., 2021) show no, or limited, inflammatory response to monomer treatment as compared to aggregated α -synuclein treatment. These studies differ on the type of cells stimulated, the method of production of aggregated α -synuclein, and the chosen concentration of monomers used to stimulate. The inflammatory effect of monomers seen in this chapter was thought to be due to the high concentration of monomer chosen (4 µM). This concentration was chosen to correspond to the optimal oligomer concentration $-4 \mu M$ as according to the concentration gradient seen in **Figure 4.6**. It could be speculated that the high inflammatory response seen in response to this monomer concentration could be due to the sheer amount of protein being added to the culture medium. However, Pike et al., (2021) used concentrations of monomer up to 20 μ M and did not see any inflammatory effects – though this was in 1° human microglia, and they were only interested in NLRP3 inflammasome-mediated IL-1ß secretion – no other cytokine production was monitored. As shown in Figure 4.5, a different effect of using such a high concentration of monomers is that this makes them more prone to aggregation into higher molecular weight species. Though this has not occurred extensively in our culture medium, it can be seen that a proportion of monomers have formed oligomeric species >150 nm and that this proportion increases further when left in culture for an extra 24 hours. It is possible that these aggregated species within the culture medium of the monomer-stimulated cells are what is actually responsible for the inflammatory effects seen. This is very difficult to control for; other studies have used techniques such as filtration of monomer solutions before addition to cultures (Codolo et al., 2013; Pike et al., 2021), and vigorous trituration (to remove clumps) (Bearoff et al., 2023) to ensure that no high weight molecular species were found to contaminate the monomeric solution. This should be considered for future work using this assay. A further consideration is whether to control for the starting concentration of monomers,

or whether to control for the number of molecules at the end of the aggregation process. Within this chapter the former control was used – so the concentration of oligomers specified as being used was actually the concentration of the monomers which had been aggregated. This means that when monomers and oligomers at a corresponding concentration are added to the cultured cells, the total number of oligomers present is fewer than the total number of monomers present – as the oligomers are formed from, multiple monomers. This could explain why the oligomers, though suggested to be the more immunogenic species, produce a very similar response to the presumed less immunogenic but more abundant monomeric species.

4.5.4 TLR blockers

The use of candesartan abrogated the inflammatory response of the cultured immune cells in response to both LPS (as a positive control) and α -synuclein species. This occurred partially when at a candesartan dose of 10 µM, and fully when at 20 µM and above. The method through which candesartan achieved this is not conclusive due to candesartan having varied effects. Candesartan works to block AT₁ receptors in its role as an anti-hypertensive; AT₁ receptors can enhance the activation of NF-KB (Skultetyova et al., 2007) and thus the blockage of these AT_1 receptors by candesartan may block this NF- κ B activation and the subsequent production of pro-inflammatory cytokines. It is possible that candesartan is having antiinflammatory effects in this manner in this assay, however this would not explain the blockage of a response to LPS. LPS is known to exert its inflammatory effects through binding to and activating TLR4, resulting in the activation of downstream pro-inflammatory factors. In this way candesartan appears to be having an anti-inflammatory effect - at least in the LPSstimulated cells - through reducing the expression of TLR4 (and potentially also TLR2) as shown in work by Dasu et al., (2009). Further work is required to determine whether candesartan's action in these α -synuclein stimulation experiments is through reduction of TLR4 gene expression. A simple suggested experiment to provide supporting evidence for this could be the use of flow cytometry to measure TLR expression on stimulated PBMCs with and without candesartan treatment. Regardless of mechanism, candesartan is having antiinflammatory effects. This could be conducive to its having a neuroprotective effect, as was seen in *Chapter 2* (though through an undetermined mechanism). The tolerability of long-term treatment with candesartan is well characterised through its use as an anti-hypertensive. Therefore its anti-inflammatory properties to stimulation with α -synuclein combined with its

already being a licensed drug make it an ideal candidate for further study and a potential clinical trial.

Similarly to candesartan, TAK-242 abrogated the inflammatory response of the cultured immune cells to both LPS and α -synuclein species. This was expected for LPS, as it is a TLR4 ligand, but shows that TLR4 is also essential for the α -synuclein-mediated inflammatory response, and so provides further evidence for α -synuclein species acting as a DAMP to activate TLR4. TAK-242 has been used in a clinical trial for the treatment of sepsis (Rice et al., 2010). Though unsuccessful in its primary outcomes, this trial demonstrated that TAK-242 is well tolerated – with the rate of adverse events being only slightly higher in those treated with TAK-242 than placebo, and the most frequent adverse event being methemoglobinemia. However, this study only administered TAK-242 for a short period of time – a 30 minute loading dose followed by a 96 hour continuous infusion – for a very acute disease. The treatment of PD – a slowly progressing rather than acute disease – would require long-term treatment with TAK-242 which has not been assessed in any human trials. For this reason, candesartan makes a more promising candidate for use in treating PD in humans. However, TAK-242 is useful in helping to determine the mechanism of the protective effect of candesartan.

The blockade of TLR4 by TAK-242 clearly prevented α -synuclein-induced inflammation, however TLR2 has also been suggested to recognise α -synuclein as a DAMP. Candesartan has been shown to decrease the expression of both TLRs 2 and 4 (Dasu et al., 2009), so it is possible that the anti-inflammatory effects seen on candesartan treatment could be due to a combination of decreased TLR2 and 4 expression. The TLR2-specific blocker Cucpt22 was chosen to test this. Cu-cpt22 failed to suppress α -synuclein-induced inflammation in this assay, this indicates that the α -synuclein may be acting specifically via TLR4 to activate the PBMC immune response. However, further work is required to justify this claim as no positive control was used in this study to confirm that Cu-cpt22 was blocking TLR2 – though this has been established in the literature by Cheng et al., (2012), and has been used by Daniele et al., (2015) against α -synuclein-induced TLR2 activation. LPS predominantly activates TLR4, so explaining why the suggested blockage of TLR2 by Cu-cpt22 does not significantly affect the inflammation resulting from this. A TLR2 agonist, such as Pam-3-csk could be used in tandem with Cu-cpt22 to prove that Cu-cpt22 is indeed blocking TLR2 activation.

This work is demonstrating a potential therapeutic mechanism whereby treatment with a TLR blocker decreases the α -synuclein-induced inflammatory response of human immune cells, which could have a protective effect on the progression of PD. Further work is required to demonstrate the applicability of this in humans; this will be discussed in *Chapter 5*.

Chapter 4



Supplementary Figure 4.1 Effect of frozen oligomers on oligomer-stimulated TNF-a production. Experiment 1 used a batch of oligomers produced and used within 24 hours (full data from this experiment can be seen in Figure 4). Experiment 2 uses batches of oligomers produced and then frozen prior to use in an assay (full data for PDP 1426 can be seen in Figure 2bI; full data for PDP1544 can be seen in Figure 2bII; PDP1426 and PDP1544 were stimulated with different batches of oligomers, both were frozen before use).
Chapter 5. DISCUSSION

5.1 Introduction

In this thesis I explored the theory that TLRs play a key role in the progression of Parkinson's disease. I tested this by using TLR-blockers in an α -synuclein animal model, as well as in cultured human peripheral immune cells, to assess their effect on the inflammatory response of these systems. I aimed to address the key hypotheses that:

- TLRs play an important role in the development and progression of PD through mediating the pro-inflammatory response of microglia and peripheral monocytes to the disease state.
- TLR blockade will have a protective effect on the development of pathology in an αsynuclein animal model through the suppression of inflammation.
- TLR blockade will reduce the inflammatory response of human peripheral immune cells to α-synuclein.

5.2 Key findings

The key findings from this thesis are summarised as follows:

Candesartan shows a protective effect on the development of cholinergic neurodegeneration in an α-synuclein animal model (*Chapter 2*). An animal model of PD was produced through the injection of α-synuclein PFFs into the tail vein of rats, replicating the model produced and published by Kuan et al., (2019). Animals were dosed daily with candesartan to test how blocking TLRs may alter disease progression. A protective effect was seen in terms of reduced cholinergic degeneration in the vagus nerve nucleus in candesartan-treated as opposed to untreated PFF-injected animals. No differences in cholinergic neuron density were seen between candesartan-treated and untreated control animals, suggesting a disease-specific effect. The area of the brainstem where cholinergic degeneration occurred is a region where the earliest PD pathology is thought to occur in the human disease (Braak et al., 2003), though this α-synuclein pathology was not demonstrated in this area in my work. This could implicate

the blockage of TLRs as a key therapeutic target in early PD. A small protective effect of candesartan was also seen in terms of olfactory deficits (another early feature of human PD) with the untreated animals showing a trend towards a higher proportion of failure in the behavioural test compared to the treated animals. Furthermore, a protective effect of candesartan treatment was seen on the development of motor dysfunction, however this may be occurring due to confounding weight effects. No effect of candesartan was seen on dopaminergic degeneration, however this may have been due to this being an early stage model, and therefore it had not yet developed α synuclein pathology in the dopaminergic nigrostriatal system.

- The mechanism by which candesartan is exerting a neuroprotective effect is **unclear** (*Chapter 3*). The α -synuclein PFF animal model does not result in measurable neuroinflammation, and so candesartan does not appear to be exerting the observed protective effect through reducing brain inflammation. It also did not reduce the expression of AT₁ receptors, but it was not determined whether blockage of these receptors was occurring. Though established as decreasing the expression of TLRs 2 and 4 by Dasu et al., (2009), in my work, candesartan treatment showed only a small trend towards a decrease in the expression and in the protein levels of TLRs 2 and 4 in the rat brain. The expression of downstream factors in the brain, including the proinflammatory cytokines TNF- α , IL-1 β , and IL-6, did not show a decrease on candesartan treatment. Similarly, the majority of peripheral cytokines, including TNF- α , showed no changes on candesartan-treatment, and in fact many consistently remained below the limit of detection of the MSD assay regardless of treatment group. KC/GRO (CXCL1) was the exception to this, showing significant upregulation at 2 months. This upregulation was attenuated by candesartan treatment, indicative of a potential protective effect. Finally, no differences were seen in microglial activation on candesartan treatment as shown through assessing microglial morphological changes, microgliosis, and MHCII expression. There does not appear to be any induction of neuroinflammation in the animal model when measured at 6 months, however this may be due to a failure to assess brain inflammation at an earlier timepoint, particularly given that peripheral inflammation – in the form of upregulation of KC/GRO – was identified at 2 months.
- A-synuclein species are pro-inflammatory to human peripheral immune cells, and this inflammation appears to be TLR4-dependent (*Chapter 4*). Different forms of

 α -synuclein, including oligomers and monomers, resulted in a pro-inflammatory TNF- α response in peripheral immune cells from both people with PD and healthy controls. Optimal experimental timecourses and concentrations of α -synuclein were established to lay the ground for future work to explore further whether there are differences in inflammatory responses between controls and PD. Three different TLR blockers were used to distinguish whether TLR 2 or 4 is the most important in mediating the α synuclein-induced inflammatory response, and also to assess whether the licensed drug candesartan – used in the previous chapters in an animal model – was capable of suppressing this inflammation. Optimal concentrations were established for these TLR blockers to allow this work to be extended to a large cohort of controls and PD. This work showed suppression of α -synuclein-induced inflammation by higher concentrations of candesartan, and by the specific TLR4-blocker TAK-242. The TLR2blocker Cu-cpt22 did not suppress this inflammation.

5.3 General discussion

Interest in the role of TLRs in PD, and indeed TLRs in neurodegenerative diseases in general, has grown over the past 15 years. Many studies have provided evidence to support a role for TLRs in Parkinson's disease, particularly TLRs 2 and 4 –see **Table 1.1, Table 1.2, Table 1.3, Table 1.4** – however the use of a TLR blocker to treat PD in a clinical trial has yet to be carried out. In this thesis I aimed to gain pre-clinical evidence to support the use of the licensed drug candesartan in a clinical trial, by investigating whether candesartan could attenuate α -synuclein related pathology in an animal model. There is disagreement in the literature over whether TLR2 or TLR4 is the most important in relaying the inflammatory response to α -synuclein, but candesartan has been reported to decrease the expression of both TLRs 2 and 4 (Dasu et al., 2009), thus providing a way to target both TLRs with one drug. Candesartan is an ideal candidate for repurposing in a clinical trial due to it already being a licensed drug, and its well-established chronic use to treat hypertension, with minimal adverse effects.

My work in *Chapter 2* shows that candesartan is having a protective effect against the development of neurodegeneration within the animal model, providing evidence for a positive effect if used therapeutically. Due to the model not developing as expected, the main pathological deficits seen were cholinergic degeneration in the hypoglossal and vagal nerve

nuclei. These areas are the first areas which degenerate within the human PD brain (Braak et al., 2003). In keeping with this, the animals exhibited olfactory deficits, which represent one of the first features of human PD, occurring over 20 years prior to the development of motor symptoms according to some studies (Fereshtehnejad et al., 2019). As such the model produced by Kuan et al., (2019) and replicated within my thesis could be described as a prodromal model of PD whereby neurodegeneration has begun in specific tracts of the brain but is yet to spread to the dopaminergic tracts, and the beginnings of behavioural deficits are being seen but these are yet to extend to motor deficits. Further study by Kuan et al., (2022) showed that when left for up to 18 months post PFF-injection degeneration of TH neurons occurred to an even greater extent. This suggests that if my animals were left to progress for longer than 6 months then pathology may have been seen within the dopaminergic system. Prodromal animal models of PD are not common (Taguchi et al., 2020). Many models do display some features seen in prodromal PD – such as olfactory dysfunction and GI dysfunction. However, these frequently occur alongside largescale neurodegeneration (Höglinger et al., 2015; Prediger et al., 2006; Rodrigues et al., 2014; Tian et al., 2008), or are induced through non-physiological means such as acute environmental toxin exposure without Lewy pathology or injection directly into the olfactory bulb or GI tract (Anderson et al., 2007; Drolet et al., 2009; Kim et al., 2019; Rey et al., 2016). As a result, these models are not particularly accurate models of idiopathic PD, and furthermore they frequently develop too quickly to allow long-term drug intervention to be assessed. The model that I utilised therefore provides a novel tool for investigating the initial stages of α -synuclein induced pathology, including prodromal features, and provides a solid basis for my work investigating the effect of TLR suppression, using candesartan treatment, on pathology development. Furthermore, there is increased interest in clinical trials being carried out in prodromal PD, such as in people with genetic mutations which predispose them to developing PD, or people experiencing REM sleep behaviour disorder (RBD) who have a high chance of going on to develop PD (Roguski et al., (2020) estimate there to be around 70% of RBD patients who go on to develop PD). My work is particularly relevant for these prodromal trials as candesartan is having a protective effect on prodromal symptoms in the α -synuclein animal model. However, in this model candesartan was given prior to the induction of disease which is not possible to achieve in humans as even during the prodromal phase pathology has already started.

Many studies investigating the role of TLRs in mediating neurodegenerative pathology use TLR-knockout models (see **Table 1.2** and **Table 1.4** for specific studies).

Though effective as a proof of concept for TLR-related protective mechanisms, TLR knockout is clearly not a viable solution for use in humans. Other studies have investigated the blockage of TLRs using anti-TLR antibodies (Codolo et al., 2013; Gustot et al., 2015; Kim et al., 2018; La Vitola et al., 2018; Stefanova et al., 2011), or TLR inhibitors (Chung et al., 2022; Daniele et al., 2015; Dutta et al., 2021). None of these therapies have been used in clinical trials to date. Antibody therapies have been suggested in PD but the focus so far has been on anti- α -synuclein antibodies. These have shown some progress in animal models and early clinical development (Schneeberger et al., 2016), but there are risks associated with such therapies, exemplified by those associated with the lecanemab and aducanumab Alzheimer's trials (Mahase, 2022; Walsh et al., 2021). Kim et al., (2018) suggest that antibody therapies aimed at neuroinflammation should also be explored. However more readily available and less experimental treatments may be a more successful way forward. TLR inhibitors such as Cu-cpt22 (Daniele et al., 2015), C29, CLI-095 (Chung et al., 2022), wtTIDM (Dutta et al., 2021), and TAK-242 (Hughes et al., 2019) have all been explored in relation to in vitro cell cultures and PD animal models. C29 (TLR2 small molecule inhibitor), CLI-095 (TLR4 small molecule inhibitor), and wtTIDM (peptide blocking TLR2-interaction domain of MyD88) have only been used in a limited number of studies relating to PD animal or *in vitro* models. Though suggested to be protective against α -synuclein-induced inflammation in these PD models, they have only contributed to mechanistic knowledge and following on from this the authors have not suggested their therapeutic use. TAK-242 (TLR4 small molecule inhibitor) has shown neuroprotective effects in animal models for conditions including amyotrophic lateral sclerosis (ALS) and traumatic brain injury (TBI) through its blockade of TLR4 and subsequent reduction in neuroinflammation. (Fellner et al., 2017; Feng et al., 2017). Hughes et al., (2019) also showed this may be relevant to PD with reduction in α -synuclein-induced oxidative stress, neuronal cell death, and inflammation on TAK-242 treatment of BV2 microglial cells. TAK-242 has previously been used in a Phase I randomised, double-blind, placebo-controlled clinical trial for the treatment of sepsis which, although negative for its primary outcome, demonstrated the safety and short-term tolerability of this compound (Rice et al., 2010). Although these studies were in disorders other than a-synucleinopathies, they indicate that TAK-242 could have beneficial effects in PD through the blocking of TLR4 and subsequent decrease in the neuroinflammatory effects of α -synuclein. However, there is little research on the safety of this compound for long-term treatment which would be essential for its use in the treatment of PD. CU-CPT22 (TLR2 small molecule inhibitor) has been shown to reduce a-synuclein-induced TNF- α production in primary microglia (Daniele et al., 2015). A protective effect against

Chapter 5

pathological α -synuclein accumulation and motor and sensory dysfunction was similarly seen in an MPTP animal model. CU-CPT22 has been used in a type 2 diabetes mellitus mouse model and suppressed the expression of pro-inflammatory cytokines – including TNF- α , IL-1 β , and IL-6. Though not directly relevant to PD, this work does show that CU-CPT22 can be tolerated and has anti-inflammatory effects *in vivo* (Tian et al., 2021). No work has been carried out investigating the safety and tolerability of CU-CPT22 in humans.

My work described in this thesis has provided further evidence for a potential protective role of TAK-242 in α -synuclein-induced inflammation and for the first time demonstrated that this protective effect occurs in human immune cells. This effect was not seen for CU-CPT22, in contrast to the literature, though further work needs to be done to ensure that TLR2 was successfully blocked in my study. However, the main focus of this thesis in terms of TLR blockers was candesartan. This drug has a number of benefits over blockers such as TAK-242 and CU-CPT22. Candesartan is a licensed drug and used in the long term treatment of hypertension meaning that its safety and tolerability over a long timescale has already been assessed. My work has shown that candesartan has protective effects against the development of Parkinson's-like pathology in an animal model. I could not clearly demonstrate the mechanism through which this occurred but candesartan has multiple potential mechanisms of action. Firstly, the blockage of AT₁ receptors (Rodriguez-Perez et al., 2018). Secondly, the inhibition of autophagy through decreasing the expression of TLRs on neurons (Chedid et al., 2022; Chung et al., 2022; Kim et al., 2021). Thirdly, anti-inflammatory effects, as demonstrated in human peripheral immune cells stimulated with α -synuclein. These effects may be occurring through decreasing the expression of TLRs 2 and 4. Though not proven in my animal study, there is abundant evidence from the literature that inflammation plays a role in the development and progression of PD-like pathology in animal models and in the risk for, and progression of human PD. Hence the anti-inflammatory effect of candesartan may be relevant to its neuroprotective action.

5.4 Future directions

5.4.1 Further exploration of candesartan

The work I have carried out using the Kuan et al., (2019) animal model has shown promising results in terms of demonstrating that the licensed drug candesartan can slow the

development of neurodegeneration in a model of prodromal/early PD. My work provides additional support for repurposing candesartan in a clinical trial in PD, however further work is need to clarify the mechanism of action, and to determine whether candesartan also impacts the progression of more established disease.

In this study, candesartan was used in a model of PD specifically focussing on the development and prodromal stages of PD, including cholinergic neuron degeneration in the brainstem and olfactory deficits. This study therefore provides evidence suggesting candesartan might be beneficial to modify the course of prodromal or early PD. However, due to this model not displaying some key features of established PD - namely the presence of pathological α -synuclein in key PD-related brain areas, degeneration of dopaminergic neurons in the substantia nigra, the degeneration of dopaminergic fibres in the striatum, and motor deficits – it is unknown whether candesartan also protects against these features which develop later in disease (according to Braak et al., 2003). Further work should be carried out to investigate the effect of candesartan treatment on these key features. A more acute animal model of Parkinson's could be used, such as a double hit α -synuclein overexpressing model injected with LPS (Gao et al., 2011). This model produces substantial dopaminergic degeneration in the striatum and substantia nigra over a period of 5 months post injection, in addition to aggregated α -synuclein inclusions and neuroinflammation. Though not the best model in regards to emulating idiopathic PD because of the non-physiological levels of α synuclein in addition to the injection of a foreign inflammatory agent which clearly does not occur in human PD, this model would allow determination of the effects of candesartan on dopaminergic degeneration and on inflammation. In addition, similarly to the Kuan model, this model develops over a relatively long time period (5 months) so allowing the assessment of relatively long-term drug intervention, though evidently not a time period as long as that over which human PD develops. This would build on the data from my work focusing on early cholinergic degeneration in the Kuan et al., (2019) model, and will provide further rationale for the use of candesartan in a clinical trial.

The investigation of the protective effects of candesartan on the development of pathology in a model of PD was carried out using a protocol where treatment was started prior to the establishment of pathology in the model. This is a commonly used approach, done so as to maximise the effect of the drug on pathology, however this is clearly not possible to achieve when treating the human disease. The pathology of PD is thought to begin many years prior to

Chapter 5

the development of motor symptoms and diagnosis of disease. Even in people with prodromal features of PD such as hyposmia and RBD, the pathological process has already started. Therefore, any follow-on studies to the α -synuclein animal work carried out in this thesis should carefully consider ways in which it would be possible also to test the effect of candesartan treatment on the progression of pathology when pathology has already been established. This could be carried out using the Kuan et al., (2019) model where candesartan treatment could be started at the 6 month timepoint where pathology is known to have developed, and these animals could then be studied for a further 12-months as a follow up study by Kuan et al., (2022) showed the further development of pathology up to 18 months post PFF injection.

Though my work has suggested a protective effect of candesartan treatment, the mechanism through which candesartan is exerting this effect has not been successfully established. My original hypothesis was that candesartan would act as a TLR blocker, and that this blockage of inflammation would result in protective effects on pathology. Unfortunately, neuroinflammation was not detectable within the animal model used in my studies, and as result I could not measure the impact of candesartan on inflammation – as shown in *Chapter 3*. Furthermore, the expected inhibition of expression of AT₁ receptors on candesartan use was not seen, and so the action of candesartan was not determined within this work. Mechanism of action is important to establish before use in a clinical trial so as to allow proof of target engagement and the monitoring of relevant mechanistic biomarkers. Further work to establish whether candesartan is exerting an anti-inflammatory effect in vivo could include looking at an earlier timepoint in the PFF model, or looking at a timepoint in between those used in my work; this could expand on the KC/GRO upregulation seen in the PFF model at 2 months. Alternatively, an LPS double-hit model could be used due to LPS being well-established as resulting in inflammation. These works should carefully assess the peripheral immune response in addition to the central immune response so as to build on the candesartan-induced reduction in KC/GRO seen in my work. Moreover, TLRs are expressed on innate immune cells in the periphery as well as in the brain; the peripheral levels of TLRs have not been determined in my work and may be playing an important role.

As discussed in *Chapter 3.5.3*, these protective effects may arise from candesartan's action on the renin-angiotensin system and its potential regulation of oxidative and inflammatory stresses through this mechanism. Further work is needed to assess the effect of

candesartan on the renin-angiotensin system in my work. Though the expression of AT_1 was measured, this did not show any differences in AT_1 mRNA levels on candesartan treatment. Techniques such as quantitative autoradiography may be used as a more accurate alternative to assess the ability of angiotensin II to bind to the AT_1 receptors on candesartan treatment (as carried out in work including Armando et al., (2001) and Nishimura et al., (2000)).

Alternatively, candesartan may still be working through neuronal TLRs, however instead of regulating inflammation it may be having an effect on the autophagy-lysosome system to remove aggregated α -synuclein. The assessment of this in an animal model would require a method of measuring TLR expression specifically on different cell types – potentially though immunofluorescence double-labelling with a neuronal marker and an anti-TLR antibody. In addition, post-mortem work could be carried out to assess the levels of key readouts of autophagy in neurons such as p62 expression. Finally, TLRs expressed by innate peripheral immune cells may also play a role in the development of pathology; these could be assessed using flow cytometry techniques.

5.4.2 Human PBMC studies

The work carried out in *Chapter 4* has optimised a protocol for further studies on human PBMC inflammatory responses to α -synuclein. This work was planned with the ultimate goal of carrying out a much larger study comparing the responses of PBMCs from a cohort of PD patients versus age-matched healthy controls. Unfortunately, due to time restraints, this was not possible to achieve within the timeframe of my PhD, however this will be an important future experiment. This work could utilise existing PBMC samples from the NET-PDD (Neuroinflammation and Tau aggregation in Parkinson's Disease Dementia) study carried out by Dr Antonina Kouli and Dr Caroline Williams-Gray. This study recruited newlydiagnosed patients – within 2 years of diagnosis – stratified into two prognostic groups according to their risk of developing dementia during disease progression, and age-matched controls. As the primary goal of the study was to monitor neuroinflammation longitudinally, the cohort was recruited with strict exclusion criteria in regards to any confounding inflammation – i.e. no chronic inflammatory or autoimmune disorders and no use of antiinflammatory or immune modulating medications. This makes the samples ideal for use in this study to determine the inflammatory reaction of PBMCs to α -synuclein.

Chapter 5

Stored PBMCs from participants from each of the aforementioned stratified prognostic groups in addition to age and sex-matched healthy controls could be used. These could be cultured, stimulated, and treated as optimised in *Chapter 4*. This would allow an investigation of whether the TLR-mediated inflammatory response is related to clinical heterogeneity and the progression of disease. It would also allow further investigation into whether the activation of peripheral immune cells observed in patient-derived PBMCs (Drouin-Ouellet et al., 2015; Grozdanov et al., 2014; Wijeyekoon et al., 2020) is due to inherent differences in the function of immune cells in the disease state, or whether it is due to extrinsic factors such as exposure to pathogenic α -synuclein aggregates when in the disease state. This work could then be extended to see how immune cells interact within an environment more physiologically similar to that which would be experienced within the PD brain. Work by Emin et al., (2022) has utilised a technique produced by Hong et al., (2018) whereby post-mortem brain is soaked to extract the soluble oligomeric species. Emin showed that soaked PD brains contained a larger proportion of smaller α -synuclein aggregates than control brain, and that these aggregates produced a greater inflammatory response when added to cultured BV2 microglia. The use of soluble extracts from soaked PD or control brains to stimulate PBMCs from PD or control individuals would allow further analysis of whether the immune response of PBMCs is inherently different when the immune cells are derived from people with PD, or whether the extrinsic environment (i.e. the nature of α -synuclein species from the soaked diseased brain) is a greater factor in the differing levels of inflammation seen in people with PD. These studies using soaked brain could also allow further analysis of whether TLR blockers, in particular candesartan, are sufficient to control an immune response triggered by pathological α -synuclein species.

The work completed in *Chapter 4* was carried out using whole PBMCs due to my early work demonstrating that cultured monocytes show high basal levels of inflammation, potentially due to stress from the combination of the freeze/thaw and monocyte isolation processes. It was also thought that PBMCs may more accurately demonstrate physiological conditions whereby interactions between different types of immune cells are occurring. Further work could be carried out to determine which immune cell type primarily underlies the PBMC response to α -synuclein shown in my work. This could be achieved using post-culture flow cytometry to investigate specific immune cell activation markers, and could be further extended to intracellular flow to measure the cytokine production – particularly focussing on TNF- α – from these different cell types. Flow cytometry could also be carried out to assess TLR levels on different cell types, and to investigate whether the candesartan treatment is decreasing the TLR expression, or whether it is having its anti-inflammatory effects through different mechanisms.

The question of whether inflammation is key to the neurodegeneration seen in the PD brain is a contentious one. The *in vitro* assays used in *Chapter 4* could be extended to explore the link between neuroinflammation and neurodegeneration. The treatment of cultured neurons – in the form of patient iPSC-derived neurons, a neuron cell line such as SH-SY5Y, or rat primary neurons – with the supernatant from the cultured and α -synuclein stimulated human PBMCs used in this work could be carried out. This would provide an insight into how the inflammatory factors released from α -synuclein-stimulated cells affect the survival and health of neurons. This could then be extended to investigate the protective effects of TLR blockers and whether their action on immune cells is sufficient to decrease the hypothesised toxicity on neurons. Furthermore, TLR blockers could be used specifically on cultured neurons to investigate the suggestion previously discussed that the protective action of candesartan in the animal model is due to candesartan abrogating the neuronal TLR-induced inhibition of autophagy. This work would contribute to an understanding of the protective effects.

These further projects would extend the work I have carried out and provide further evidence to support the use of TLR blockers, such as candesartan, in a clinical trial for PD. Ultimately this may lead to a new therapeutic strategy through which the progression of PD may be slowed, with implications for improving the quality of life of many around the world experiencing PD.

REFERENCES

- Abdelmotilib, H., Maltbie, T., Delic, V., Liu, Z., Hu, X., Fraser, K.B., Moehle, M.S., Stoyka, L., Anabtawi, N., Krendelchtchikova, V., Volpicelli-Daley, L.A., West, A., 2017. α-Synuclein Fibril-induced Inclusion Spread in Rats and Mice Correlates with Dopaminergic Neurodegeneration. Neurobiol. Dis. 105, 84–98. https://doi.org/10.1016/j.nbd.2017.05.014
- Adams-Carr, K.L., Bestwick, J.P., Shribman, S., Lees, A., Schrag, A., Noyce, A.J., 2016. Constipation preceding Parkinson's disease: a systematic review and meta-analysis. J. Neurol. Neurosurg. Psychiatry 87, 710–716. https://doi.org/10.1136/jnnp-2015-311680
- Adamson, A., Buck, S.A., Freyberg, Z., De Miranda, B.R., 2022. Sex Differences in Dopaminergic Vulnerability to Environmental Toxicants — Implications for Parkinson's Disease. Curr. Environ. Health Rep. https://doi.org/10.1007/s40572-022-00380-6
- Akira, S., Uematsu, S., Takeuchi, O., 2006. Pathogen Recognition and Innate Immunity. Cell 124, 783–801. https://doi.org/10.1016/j.cell.2006.02.015
- Allen, A.M., MacGregor, D.P., Chai, S.Y., Donnan, G.A., Kaczmarczyk, S., Richardson, K., Kalnins, R., Ireton, J., Mendelsohn, F.A.O., 1992. Angiotensin II receptor binding associated with nigrostriatal dopaminergic neurons in human basal ganglia. Ann. Neurol. 32, 339–344. https://doi.org/10.1002/ana.410320306
- Alvarez-Erviti, L., Couch, Y., Richardson, J., Cooper, J.M., Wood, M.J.A., 2011. Alphasynuclein release by neurons activates the inflammatory response in a microglial cell line. Neurosci. Res. 69, 337–342. https://doi.org/10.1016/j.neures.2010.12.020
- Alvarez-Erviti, L., Rodriguez-Oroz, M.C., Cooper, J.M., Caballero, C., Ferrer, I., Obeso, J.A., Schapira, A.H.V., 2010. Chaperone-mediated autophagy markers in Parkinson disease brains. Arch. Neurol. 67, 1464–1472. https://doi.org/10.1001/archneurol.2010.198
- Anderson, G., Noorian, A.R., Taylor, G., Anitha, M., Bernhard, D., Srinivasan, S., Greene, J.G., 2007. Loss of enteric dopaminergic neurons and associated changes in colon motility in an MPTP mouse model of Parkinson's disease. Exp. Neurol. 207, 4–12. https://doi.org/10.1016/j.expneurol.2007.05.010
- Anglade, P., Vyas, S., Javoy-Agid, F., Herrero, M.T., Michel, P.P., Marquez, J., Mouatt-Prigent, A., Ruberg, M., Hirsch, E.C., Agid, Y., 1997. Apoptosis and autophagy in nigral neurons of patients with Parkinson's disease. Histol. Histopathol. 12, 25–31.
- Armando, I., Carranza, A., Nishimura, Y., Barontini, M., Ito, T., Saavedra, J.M., 2001. Candesartan decreases the sympatho-adrenal and hormonal response to isolation stress.
 J. Renin-Angiotensin-Aldosterone Syst. JRAAS 2, S130–S135. https://doi.org/10.1177/14703203010020012301
- Asenbaum, S., Brücke, T., Pirker, W., Podreka, I., Angelberger, P., Wenger, S., Wöber, C., Müller, C., Deecke, L., 1997. Imaging of Dopamine Transporters with Iodine-123-β-CIT and SPECT in Parkinson's Disease. J. Nucl. Med. 38, 1–6.
- Athauda, D., Foltynie, T., 2016. Insulin resistance and Parkinson's disease: A new target for disease modification? Prog. Neurobiol. 145–146, 98–120. https://doi.org/10.1016/j.pneurobio.2016.10.001

- Bae, E.-J., Ho, D.-H., Park, E., Jung, J.W., Cho, K., Hong, J.H., Lee, H.-J., Kim, K.P., Lee, S.-J., 2013. Lipid Peroxidation Product 4-Hydroxy-2-Nonenal Promotes Seeding-Capable Oligomer Formation and Cell-to-Cell Transfer of α-Synuclein. Antioxid. Redox Signal. 18, 770–783. https://doi.org/10.1089/ars.2011.4429
- Barajon, I., Serrao, G., Arnaboldi, F., Opizzi, E., Ripamonti, G., Balsari, A., Rumio, C., 2009. Toll-like Receptors 3, 4, and 7 Are Expressed in the Enteric Nervous System and Dorsal Root Ganglia. J. Histochem. Cytochem. 57, 1013–1023. https://doi.org/10.1369/jhc.2009.953539
- Barakat, W., Safwet, N., El-Maraghy, N.N., Zakaria, M.N.M., 2014. Candesartan and glycyrrhizin ameliorate ischemic brain damage through downregulation of the TLR signaling cascade. Eur. J. Pharmacol. 724, 43–50. https://doi.org/10.1016/j.ejphar.2013.12.032
- Barnum, C.J., Chen, X., Chung, J., Chang, J., Williams, M., Grigoryan, N., Tesi, R.J., Tansey, M.G., 2014. Peripheral Administration of the Selective Inhibitor of Soluble Tumor Necrosis Factor (TNF) XPro ®1595 Attenuates Nigral Cell Loss and Glial Activation in 6-OHDA Hemiparkinsonian Rats. J. Park. Dis. 4, 349–360. https://doi.org/10.3233/JPD-140410
- Bauer, J., Sminia, T., Wouterlood, F.G., Dijkstra, C.D., 1994. Phagocytic activity of macrophages and microglial cells during the course of acute and chronic relapsing experimental autoimmune encephalomyelitis. J. Neurosci. Res. 38, 365–375. https://doi.org/10.1002/jnr.490380402
- Baufeld, C., O'Loughlin, E., Calcagno, N., Madore, C., Butovsky, O., 2018. Differential contribution of microglia and monocytes in neurodegenerative diseases. J. Neural Transm. 125, 809–826. https://doi.org/10.1007/s00702-017-1795-7
- Bearoff, F., Dhavale, D., Kotzbauer, P., Kortagere, S., 2023. Aggregated alpha-synuclein transcriptionally activates pro-inflammatory canonical and non-canonical NF-κB signaling pathways in peripheral monocytic cells. Mol. Immunol. 154, 1–10. https://doi.org/10.1016/j.molimm.2022.12.006
- Benner, E.J., Banerjee, R., Reynolds, A.D., Sherman, S., Pisarev, V.M., Tsiperson, V., Nemachek, C., Ciborowski, P., Przedborski, S., Mosley, R.L., Gendelman, H.E., 2008. Nitrated alpha-synuclein immunity accelerates degeneration of nigral dopaminergic neurons. PloS One 3, e1376. https://doi.org/10.1371/journal.pone.0001376
- Beraud, D., Twomey, M., Bloom, B., Mittereder, A., Neitzke, K., Ton, V., Chasovskikh, S., Mhyre, T., Maguire-Zeiss, K., 2011. α-Synuclein Alters Toll-Like Receptor Expression. Front. Neurosci. 5.
- Bernheimer, H., Birkmayer, W., Hornykiewicz, O., Jellinger, K., Seitelberger, F., 1973. Brain dopamine and the syndromes of Parkinson and Huntington Clinical, morphological and neurochemical correlations. J. Neurol. Sci. 20, 415–455. https://doi.org/10.1016/0022-510X(73)90175-5

- Betarbet, R., Sherer, T.B., MacKenzie, G., Garcia-Osuna, M., Panov, A.V., Greenamyre, J.T., 2000. Chronic systemic pesticide exposure reproduces features of Parkinson's disease. Nat. Neurosci. 3, 1301–1306. https://doi.org/10.1038/81834
- Bianchi, M.E., 2007. DAMPs, PAMPs and alarmins: all we need to know about danger. J. Leukoc. Biol. 81, 1–5. https://doi.org/10.1189/jlb.0306164
- Bliederhaeuser, C., Zondler, L., Grozdanov, V., Ruf, W.P., Brenner, D., Melrose, H.L., Bauer, P., Ludolph, A.C., Gillardon, F., Kassubek, J., Weishaupt, J.H., Danzer, K.M., 2016. LRRK2 contributes to monocyte dysregulation in Parkinson's disease. Acta Neuropathol. Commun. 4, 123. https://doi.org/10.1186/s40478-016-0396-2
- Borghammer, P., Just, M.K., Horsager, J., Skjærbæk, C., Raunio, A., Kok, E.H., Savola, S., Murayama, S., Saito, Y., Myllykangas, L., Van Den Berge, N., 2022. A postmortem study suggests a revision of the dual-hit hypothesis of Parkinson's disease. Npj Park. Dis. 8, 1–11. https://doi.org/10.1038/s41531-022-00436-2
- Braak, H., de Vos, R.A.I., Bohl, J., Del Tredici, K., 2006. Gastric α-synuclein immunoreactive inclusions in Meissner's and Auerbach's plexuses in cases staged for Parkinson's disease-related brain pathology. Neurosci. Lett. 396, 67–72. https://doi.org/10.1016/j.neulet.2005.11.012
- Braak, H., Rüb, U., Gai, W.P., Del Tredici, K., 2003. Idiopathic Parkinson's disease: possible routes by which vulnerable neuronal types may be subject to neuroinvasion by an unknown pathogen. J. Neural Transm. 110, 517–536. https://doi.org/10.1007/s00702-002-0808-2
- Braak, H., Sastre, M., Bohl, J.R.E., de Vos, R.A.I., Del Tredici, K., 2007. Parkinson's disease: lesions in dorsal horn layer I, involvement of parasympathetic and sympathetic pre- and postganglionic neurons. Acta Neuropathol. (Berl.) 113, 421–429. https://doi.org/10.1007/s00401-007-0193-x
- Braak, Tredici, K.D., Rüb, U., de Vos, R.A.I., Jansen Steur, E.N.H., Braak, E., 2003. Staging of brain pathology related to sporadic Parkinson's disease. Neurobiol. Aging 24, 197–211. https://doi.org/10.1016/S0197-4580(02)00065-9
- Brochard, V., Combadière, B., Prigent, A., Laouar, Y., Perrin, A., Beray-Berthat, V., Bonduelle, O., Alvarez-Fischer, D., Callebert, J., Launay, J.-M., Duyckaerts, C., Flavell, R.A., Hirsch, E.C., Hunot, S., 2009. Infiltration of CD4+ lymphocytes into the brain contributes to neurodegeneration in a mouse model of Parkinson disease. J. Clin. Invest. 119, 182–192. https://doi.org/10.1172/JCI36470
- Brodacki, B., Staszewski, J., Toczyłowska, B., Kozłowska, E., Drela, N., Chalimoniuk, M., Stępien, A., 2008. Serum interleukin (IL-2, IL-10, IL-6, IL-4), TNFα, and INFγ concentrations are elevated in patients with atypical and idiopathic parkinsonism. Neurosci. Lett. 441, 158–162. https://doi.org/10.1016/j.neulet.2008.06.040
- Bruunsgaard, H., Pedersen, A.N., Schroll, M., Skinhøj, P., Pedersen, B.K., 1999. Impaired production of proinflammatory cytokines in response to lipopolysaccharide (LPS) stimulation in elderly humans. Clin. Exp. Immunol. 118, 235–241. https://doi.org/10.1046/j.1365-2249.1999.01045.x

- Bsibsi, M., Ravid, R., Gveric, D., van Noort, J.M., 2002. Broad Expression of Toll-Like Receptors in the Human Central Nervous System. J. Neuropathol. Exp. Neurol. 61, 1013–1021. https://doi.org/10.1093/jnen/61.11.1013
- Burns, R.S., LeWitt, P.A., Ebert, M.H., Pakkenberg, H., Kopin, I.J., 1985. The clinical syndrome of striatal dopamine deficiency. Parkinsonism induced by 1-methyl-4phenyl-1,2,3,6-tetrahydropyridine (MPTP). N. Engl. J. Med. 312, 1418–1421. https://doi.org/10.1056/NEJM198505303122203
- Cabin, D.E., Shimazu, K., Murphy, D., Cole, N.B., Gottschalk, W., McIlwain, K.L., Orrison, B., Chen, A., Ellis, C.E., Paylor, R., Lu, B., Nussbaum, R.L., 2002. Synaptic Vesicle Depletion Correlates with Attenuated Synaptic Responses to Prolonged Repetitive Stimulation in Mice Lacking α-Synuclein. J. Neurosci. 22, 8797–8807. https://doi.org/10.1523/JNEUROSCI.22-20-08797.2002
- Camacho, M., Macleod, A.D., Maple-Grødem, J., Evans, J.R., Breen, D.P., Cummins, G., Wijeyekoon, R.S., Greenland, J.C., Alves, G., Tysnes, O.B., Lawson, R.A., Barker, R.A., Williams-Gray, C.H., 2021. Early constipation predicts faster dementia onset in Parkinson's disease. Npj Park. Dis. 7, 1–7. https://doi.org/10.1038/s41531-021-00191w
- Campolo, M., Paterniti, I., Siracusa, R., Filippone, A., Esposito, E., Cuzzocrea, S., 2019. TLR4 absence reduces neuroinflammation and inflammasome activation in Parkinson's diseases in vivo model. Brain. Behav. Immun. 76, 236–247. https://doi.org/10.1016/j.bbi.2018.12.003
- Castaño, A., Herrera, A.J., Cano, J., Machado, A., 1998. Lipopolysaccharide Intranigral Injection Induces Inflammatory Reaction and Damage in Nigrostriatal Dopaminergic System. J. Neurochem. 70, 1584–1592. https://doi.org/10.1046/j.1471-4159.1998.70041584.x
- Castle, S.C., 2000. Clinical Relevance of Age-Related Immune Dysfunction. Clin. Infect. Dis. 31, 578–585. https://doi.org/10.1086/313947
- Cereda, E., Barichella, M., Cassani, E., Caccialanza, R., Pezzoli, G., 2012. Clinical features of Parkinson disease when onset of diabetes came first: A case-control study. Neurology 78, 1507–1511. https://doi.org/10.1212/WNL.0b013e3182553cc9
- Cerri, S., Mus, L., Blandini, F., 2019. Parkinson's Disease in Women and Men: What's the Difference? J. Park. Dis. 9, 501–515. https://doi.org/10.3233/JPD-191683
- Challis, C., Hori, A., Sampson, T.R., Yoo, B.B., Challis, R.C., Hamilton, A.M., Mazmanian, S.K., Volpicelli-Daley, L.A., Gradinaru, V., 2020. Gut-seeded α-synuclein fibrils promote gut dysfunction and brain pathology specifically in aged mice. Nat. Neurosci. 23, 327–336. https://doi.org/10.1038/s41593-020-0589-7
- Chao, P., Lee, H., Yin, M., 2016. Asiatic acid attenuated apoptotic and inflammatory stress in the striatum of MPTP-treated mice. Food Funct. 7, 1999–2005. https://doi.org/10.1039/C6FO00041J
- Chedid, J., Labrador-Garrido, A., Zhong, S., Gao, J., Zhao, Y., Perera, G., Kim, W.S., Halliday, G.M., Dzamko, N., 2022. A small molecule toll-like receptor antagonist rescues α-

synuclein fibril pathology. J. Biol. Chem. 298. https://doi.org/10.1016/j.jbc.2022.102260

- Cheng, K., Wang, X., Zhang, S., Yin, H., 2012. Discovery of Small-Molecule Inhibitors of the TLR1/TLR2 Complex. Angew. Chem. 124, 12412–12415. https://doi.org/10.1002/ange.201204910
- Cheng, Y., Tong, Q., Yuan, Y., Song, X., Jiang, W., Wang, Y., Li, W., Li, Y., Zhang, K., 2023. α-Synuclein induces prodromal symptoms of Parkinson's disease via activating TLR2/MyD88/NF-κB pathway in Schwann cells of vagus nerve in a rat model. J. Neuroinflammation 20, 36. https://doi.org/10.1186/s12974-023-02720-1
- Chesselet, M.-F., 2008. In vivo alpha-synuclein overexpression in rodents: A useful model of Parkinson's disease? Exp. Neurol., The Role of α-synuclein in the Pathogenesis of Parkinson's Disease / Gene Therapy for Parkinson's 209, 22–27. https://doi.org/10.1016/j.expneurol.2007.08.006
- Choi, I., Zhang, Y., Seegobin, S.P., Pruvost, M., Wang, Q., Purtell, K., Zhang, B., Yue, Z., 2020. Microglia clear neuron-released α-synuclein via selective autophagy and prevent neurodegeneration. Nat. Commun. 11, 1386. https://doi.org/10.1038/s41467-020-15119-w
- Choi, J.-Y., Jang, E.-H., Park, C.-S., Kang, J.-H., 2005. Enhanced susceptibility to 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine neurotoxicity in high-fat diet-induced obesity. Free Radic. Biol. Med. 38, 806–816. https://doi.org/10.1016/j.freeradbiomed.2004.12.008
- Chu, Y., Dodiya, H., Aebischer, P., Olanow, C.W., Kordower, J.H., 2009. Alterations in lysosomal and proteasomal markers in Parkinson's disease: Relationship to alphasynuclein inclusions. Neurobiol. Dis. 35, 385–398. https://doi.org/10.1016/j.nbd.2009.05.023
- Chung, L.Y.-R., Lin, Y.-T., Liu, C., Tai, Y.-C., Lin, H.-Y., Lin, C.-H., Chen, C.-C., 2022. Neuroinflammation Upregulated Neuronal Toll-Like Receptors 2 and 4 to Drive Synucleinopathy in Neurodegeneration. Front. Pharmacol. 13, 845930. https://doi.org/10.3389/fphar.2022.845930
- Codolo, G., Plotegher, N., Pozzobon, T., Brucale, M., Tessari, I., Bubacco, L., Bernard, M. de, 2013. Triggering of Inflammasome by Aggregated α–Synuclein, an Inflammatory Response in Synucleinopathies. PLOS ONE 8, e55375. https://doi.org/10.1371/journal.pone.0055375
- Colton, C.A., 2009. Heterogeneity of Microglial Activation in the Innate Immune Response in the Brain. J. Neuroimmune Pharmacol. 4, 399–418. https://doi.org/10.1007/s11481-009-9164-4
- Conte, C., Roscini, L., Sardella, R., Mariucci, G., Scorzoni, S., Beccari, T., Corte, L., 2017. Toll Like Receptor 4 Affects the Cerebral Biochemical Changes Induced by MPTP Treatment. Neurochem. Res. 42, 493–500. https://doi.org/10.1007/s11064-016-2095-6

- Côté, M., Drouin-Ouellet, J., Cicchetti, F., Soulet, D., 2011. The critical role of the MyD88dependent pathway in non-CNS MPTP-mediated toxicity. Brain. Behav. Immun. 25, 1143–1152. https://doi.org/10.1016/j.bbi.2011.02.017
- Cremades, N., Cohen, S.I.A., Deas, E., Abramov, A.Y., Chen, A.Y., Orte, A., Sandal, M., Clarke, R.W., Dunne, P., Aprile, F.A., Bertoncini, C.W., Wood, N.W., Knowles, T.P.J., Dobson, C.M., Klenerman, D., 2012. Direct Observation of the Interconversion of Normal and Toxic Forms of α-Synuclein. Cell 149, 1048–1059. https://doi.org/10.1016/j.cell.2012.03.037
- Cuervo, A.M., Stefanis, L., Fredenburg, R., Lansbury, P.T., Sulzer, D., 2004. Impaired Degradation of Mutant α-Synuclein by Chaperone-Mediated Autophagy. Science 305, 1292–1295. https://doi.org/10.1126/science.1101738
- Cunningham, C., Wilcockson, D.C., Campion, S., Lunnon, K., Perry, V.H., 2005. Central and Systemic Endotoxin Challenges Exacerbate the Local Inflammatory Response and Increase Neuronal Death during Chronic Neurodegeneration. J. Neurosci. 25, 9275– 9284. https://doi.org/10.1523/JNEUROSCI.2614-05.2005
- Członkowska, A., Kohutnicka, M., Kurkowska-Jastrzębska, I., Członkowski, A., 1996. Microglial Reaction in MPTP (1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine) Induced Parkinson's Disease Mice Model. Neurodegeneration 5, 137–143. https://doi.org/10.1006/neur.1996.0020
- da Rocha Sobrinho, H.M., Saar Gomes, R., da Silva, D.J., Quixabeira, V.B.L., Joosten, L.A.B., Ribeiro de Barros Cardoso, C., Ribeiro-Dias, F., 2021. Toll-like receptor 10 controls TLR2-induced cytokine production in monocytes from patients with Parkinson's disease. J. Neurosci. Res. 99, 2511–2524. https://doi.org/10.1002/jnr.24916
- Damoiseaux, J.G., Döpp, E.A., Calame, W., Chao, D., MacPherson, G.G., Dijkstra, C.D., 1994. Rat macrophage lysosomal membrane antigen recognized by monoclonal antibody ED1. Immunology 83, 140–147.
- Daniele, S.G., Béraud, D., Davenport, C., Cheng, K., Yin, H., Maguire-Zeiss, K.A., 2015. Activation of MyD88-dependent TLR1/2 signaling by misfolded α-synuclein, a protein linked to neurodegenerative disorders. Sci Signal 8, ra45–ra45. https://doi.org/10.1126/scisignal.2005965
- Danzer, K.M., Kranich, L.R., Ruf, W.P., Cagsal-Getkin, O., Winslow, A.R., Zhu, L., Vanderburg, C.R., McLean, P.J., 2012. Exosomal cell-to-cell transmission of alpha synuclein oligomers. Mol. Neurodegener. 7, 42. https://doi.org/10.1186/1750-1326-7-42
- Dasu, M.R., Riosvelasco, A.C., Jialal, I., 2009. Candesartan inhibits Toll-like receptor expression and activity both in vitro and in vivo. Atherosclerosis 202, 76–83. https://doi.org/10.1016/j.atherosclerosis.2008.04.010
- De Lella Ezcurra, A.L., Chertoff, M., Ferrari, C., Graciarena, M., Pitossi, F., 2010. Chronic expression of low levels of tumor necrosis factor-α in the substantia nigra elicits progressive neurodegeneration, delayed motor symptoms and microglia/macrophage

activation. Neurobiol. Dis., Inflammation in Neuropsychiatric Disease 37, 630–640. https://doi.org/10.1016/j.nbd.2009.11.018

- Decressac, M., Mattsson, B., Lundblad, M., Weikop, P., Björklund, A., 2012. Progressive neurodegenerative and behavioural changes induced by AAV-mediated overexpression of α-synuclein in midbrain dopamine neurons. Neurobiol. Dis. 45, 939–953. https://doi.org/10.1016/j.nbd.2011.12.013
- Deguine, J., Barton, G.M., 2014. MyD88: a central player in innate immune signaling. F1000Prime Rep. 6. https://doi.org/10.12703/P6-97
- Dehay, B., Bové, J., Rodríguez-Muela, N., Perier, C., Recasens, A., Boya, P., Vila, M., 2010. Pathogenic lysosomal depletion in Parkinson's disease. J. Neurosci. Off. J. Soc. Neurosci. 30, 12535–12544. https://doi.org/10.1523/JNEUROSCI.1920-10.2010
- Del Tredici, K., Hawkes, C.H., Ghebremedhin, E., Braak, H., 2010. Lewy pathology in the submandibular gland of individuals with incidental Lewy body disease and sporadic Parkinson's disease. Acta Neuropathol. (Berl.) 119, 703–713. https://doi.org/10.1007/s00401-010-0665-2
- Delpedro, A.D., Barjavel, M.J., Mamdouh, Z., Faure, S., Bakouche, O., 1998. Signal Transduction in LPS-Activated Aged and Young Monocytes. J. Interferon Cytokine Res. 18, 429–437. https://doi.org/10.1089/jir.1998.18.429
- Desplats, P., Lee, H.-J., Bae, E.-J., Patrick, C., Rockenstein, E., Crews, L., Spencer, B., Masliah, E., Lee, S.-J., 2009. Inclusion formation and neuronal cell death through neuron-to-neuron transmission of alpha-synuclein. Proc. Natl. Acad. Sci. U. S. A. 106, 13010–13015. https://doi.org/10.1073/pnas.0903691106
- Deumens, R., Blokland, A., Prickaerts, J., 2002. Modeling Parkinson's Disease in Rats: An Evaluation of 6-OHDA Lesions of the Nigrostriatal Pathway. Exp. Neurol. 175, 303– 317. https://doi.org/10.1006/exnr.2002.7891
- Devi, L., Raghavendran, V., Prabhu, B.M., Avadhani, N.G., Anandatheerthavarada, H.K., 2008. Mitochondrial Import and Accumulation of α-Synuclein Impair Complex I in Human Dopaminergic Neuronal Cultures and Parkinson Disease Brain *. J. Biol. Chem. 283, 9089–9100. https://doi.org/10.1074/jbc.M710012200
- Devos, D., Lebouvier, T., Lardeux, B., Biraud, M., Rouaud, T., Pouclet, H., Coron, E., Bruley des Varannes, S., Naveilhan, P., Nguyen, J.-M., Neunlist, M., Derkinderen, P., 2013. Colonic inflammation in Parkinson's disease. Neurobiol. Dis. 50, 42–48. https://doi.org/10.1016/j.nbd.2012.09.007
- Doherty, K.M., Hardy, J., 2013. Parkin Disease and the Lewy Body Conundrum. Mov. Disord. 28, 702–704. https://doi.org/10.1002/mds.25486
- Doorn, K.J., Moors, T., Drukarch, B., van de Berg, W.D., Lucassen, P.J., van Dam, A.-M., 2014. Microglial phenotypes and toll-like receptor 2 in the substantia nigra and hippocampus of incidental Lewy body disease cases and Parkinson's disease patients. Acta Neuropathol. Commun. 2, 90. https://doi.org/10.1186/s40478-014-0090-1

- Dorsey, E.R., Sherer, T., Okun, M.S., Bloem, B.R., n.d. The Emerging Evidence of the Parkinson Pandemic. J. Park. Dis. 8, S3–S8. https://doi.org/10.3233/JPD-181474
- Doty, R.L., Deems, D.A., Stellar, S., 1988. Olfactory dysfunction in parkinsonism: A general deficit unrelated to neurologic signs, disease stage, or disease duration. Neurology 38, 1237–1237. https://doi.org/10.1212/WNL.38.8.1237
- Drolet, R.E., Cannon, J.R., Montero, L., Greenamyre, J.T., 2009. Chronic rotenone exposure reproduces Parkinson's disease gastrointestinal neuropathology. Neurobiol. Dis. 36, 96–102. https://doi.org/10.1016/j.nbd.2009.06.017
- Drouin-Ouellet, J., Cicchetti, F., 2012. Inflammation and neurodegeneration: the story "retolled." Trends Pharmacol. Sci. 33, 542–551. https://doi.org/10.1016/j.tips.2012.07.002
- Drouin-Ouellet, J., Gibrat, C., Bousquet, M., Calon, F., Kriz, J., Cicchetti, F., 2011. The role of the MYD88-dependent pathway in MPTP-induced brain dopaminergic degeneration. J. Neuroinflammation 8, 137. https://doi.org/10.1186/1742-2094-8-137
- Drouin-Ouellet, J., St-Amour, I., Saint-Pierre, M., Lamontagne-Proulx, J., Kriz, J., Barker, R.A., Cicchetti, F., 2015. Toll-Like Receptor Expression in the Blood and Brain of Patients and a Mouse Model of Parkinson's Disease. Int. J. Neuropsychopharmacol. 18. https://doi.org/10.1093/ijnp/pyu103
- Duffy, M.F., Collier, T.J., Patterson, J.R., Kemp, C.J., Luk, K.C., Tansey, M.G., Paumier, K.L., Kanaan, N.M., Fischer, D.L., Polinski, N.K., Barth, O.L., Howe, J.W., Vaikath, N.N., Majbour, N.K., El-Agnaf, O.M.A., Sortwell, C.E., 2018. Lewy body-like alphasynuclein inclusions trigger reactive microgliosis prior to nigral degeneration. J. Neuroinflammation 15, 129. https://doi.org/10.1186/s12974-018-1171-z
- Dutta, D., Jana, M., Majumder, M., Mondal, S., Roy, A., Pahan, K., 2021. Selective targeting of the TLR2/MyD88/NF-κB pathway reduces α-synuclein spreading in vitro and in vivo. Nat. Commun. 12, 5382. https://doi.org/10.1038/s41467-021-25767-1
- Dzamko, N., Gysbers, A., Perera, G., Bahar, A., Shankar, A., Gao, J., Fu, Y., Halliday, G.M., 2017. Toll-like receptor 2 is increased in neurons in Parkinson's disease brain and may contribute to alpha-synuclein pathology. Acta Neuropathol. (Berl.) 133, 303–319. https://doi.org/10.1007/s00401-016-1648-8
- Elkahloun, A.G., Rodriguez, Y., Alaiyed, S., Wenzel, E., Saavedra, J.M., 2019. Telmisartan Protects a Microglia Cell Line from LPS Injury Beyond AT1 Receptor Blockade or PPARγ Activation. Mol. Neurobiol. 56, 3193–3210. https://doi.org/10.1007/s12035-018-1300-9
- Emin, D., Zhang, Y.P., Lobanova, E., Miller, A., Li, X., Xia, Z., Dakin, H., Sideris, D.I., Lam, J.Y.L., Ranasinghe, R.T., Kouli, A., Zhao, Y., De, S., Knowles, T.P.J., Vendruscolo, M., Ruggeri, F.S., Aigbirhio, F.I., Williams-Gray, C.H., Klenerman, D., 2022. Small soluble α-synuclein aggregates are the toxic species in Parkinson's disease. Nat. Commun. 13, 5512. https://doi.org/10.1038/s41467-022-33252-6
- Evans, J.R., Mason, S.L., Williams-Gray, C.H., Foltynie, T., Brayne, C., Robbins, T.W., Barker, R.A., 2011. The natural history of treated Parkinson's disease in an incident,

community based cohort. J. Neurol. Neurosurg. Psychiatry 82, 1112–1118. https://doi.org/10.1136/jnnp.2011.240366

- Facci, L., Barbierato, M., Marinelli, C., Argentini, C., Skaper, S.D., Giusti, P., 2014. Toll-Like Receptors 2, -3 and -4 Prime Microglia but not Astrocytes Across Central Nervous System Regions for ATP-Dependent Interleukin-1β Release. Sci. Rep. 4, 6824. https://doi.org/10.1038/srep06824
- Fagiolo, U., Cossarizza, A., Scala, E., Fanales-Belasio, E., Ortolani, C., Cozzi, E., Monti, D., Franceschi, C., Paganelli, R., 1993. Increased cytokine production in mononuclear cells of healthy elderly people. Eur. J. Immunol. 23, 2375–2378. https://doi.org/10.1002/eji.1830230950
- FEARNLEY, J.M., LEES, A.J., 1991. AGEING AND PARKINSON'S DISEASE: SUBSTANTIA NIGRA REGIONAL SELECTIVITY. Brain 114, 2283–2301. https://doi.org/10.1093/brain/114.5.2283
- Fei, Q., McCormack, A.L., Di Monte, D.A., Ethell, D.W., 2008. Paraquat neurotoxicity is mediated by a Bak-dependent mechanism. J. Biol. Chem. 283, 3357–3364. https://doi.org/10.1074/jbc.M708451200
- Fellner, A., Barhum, Y., Angel, A., Perets, N., Steiner, I., Offen, D., Lev, N., 2017. Toll-Like Receptor-4 Inhibitor TAK-242 Attenuates Motor Dysfunction and Spinal Cord Pathology in an Amyotrophic Lateral Sclerosis Mouse Model. Int. J. Mol. Sci. 18, 1666. https://doi.org/10.3390/ijms18081666
- Fellner, L., Irschick, R., Schanda, K., Reindl, M., Klimaschewski, L., Poewe, W., Wenning, G.K., Stefanova, N., 2013. Toll-like receptor 4 is required for α-synuclein dependent activation of microglia and astroglia. Glia 61, 349–360. https://doi.org/10.1002/glia.22437
- Fénelon, G., 2008. Psychosis in Parkinson's Disease: Phenomenology, Frequency, Risk Factors, and Current Understanding of Pathophysiologic Mechanisms. CNS Spectr. 13, 18–25. https://doi.org/10.1017/S1092852900017284
- Feng, Y., Gao, J., Cui, Y., Li, M., Li, R., Cui, C., Cui, J., 2017. Neuroprotective Effects of Resatorvid Against Traumatic Brain Injury in Rat: Involvement of Neuronal Autophagy and TLR4 Signaling Pathway. Cell. Mol. Neurobiol. 37, 155–168. https://doi.org/10.1007/s10571-016-0356-1
- Fereshtehnejad, S.-M., Yao, C., Pelletier, A., Montplaisir, J.Y., Gagnon, J.-F., Postuma, R.B., 2019. Evolution of prodromal Parkinson's disease and dementia with Lewy bodies: a prospective study. Brain J. Neurol. 142, 2051–2067. https://doi.org/10.1093/brain/awz111
- Filipov, N.M., 2019. Chapter Six Overview of peripheral and central inflammatory responses and their contribution to neurotoxicity, in: Aschner, M., Costa, L.G. (Eds.), Advances in Neurotoxicology, Role of Inflammation in Environmental Neurotoxicity. Academic Press, pp. 169–193. https://doi.org/10.1016/bs.ant.2018.10.001

- Fitzner, N., Clauberg, S., Essmann, F., Liebmann, J., Kolb-Bachofen, V., 2008. Human Skin Endothelial Cells Can Express All 10 TLR Genes and Respond to Respective Ligands. Clin. Vaccine Immunol. CVI 15, 138–146. https://doi.org/10.1128/CVI.00257-07
- Frank-Cannon, T.C., Alto, L.T., McAlpine, F.E., Tansey, M.G., 2009. Does neuroinflammation fan the flame in neurodegenerative diseases? Mol. Neurodegener. 4, 47. https://doi.org/10.1186/1750-1326-4-47
- Fu, H., Hosomi, N., Pelisch, N., Nakano, D., Liu, G., Ueno, M., Miki, T., Masugata, H., Sueda, Y., Itano, T., Matsumoto, M., Nishiyama, A., Kohno, M., 2011. Therapeutic effects of postischemic treatment with hypotensive doses of an angiotensin II receptor blocker on transient focal cerebral ischemia. J. Hypertens. 29, 2210–2219. https://doi.org/10.1097/HJH.0b013e32834bbb30
- Fuchs, J., Nilsson, C., Kachergus, J., Munz, M., Larsson, E.-M., Schüle, B., Langston, J.W., Middleton, F.A., Ross, O.A., Hulihan, M., Gasser, T., Farrer, M.J., 2007. Phenotypic variation in a large Swedish pedigree due to SNCA duplication and triplication. Neurology 68, 916–922. https://doi.org/10.1212/01.wnl.0000254458.17630.c5
- Gagne, J.J., Power, M.C., 2010. Anti-inflammatory drugs and risk of Parkinson disease: a meta-analysis. Neurology 74, 995–1002. https://doi.org/10.1212/WNL.0b013e3181d5a4a3
- Gao, H.-M., Hong, J.-S., 2008. Why neurodegenerative diseases are progressive: uncontrolled inflammation drives disease progression. Trends Immunol. 29, 357–365. https://doi.org/10.1016/j.it.2008.05.002
- Gao, H.-M., Zhang, F., Zhou, H., Kam, W., Wilson, B., Hong, J.-S., 2011. Neuroinflammation and α-Synuclein Dysfunction Potentiate Each Other, Driving Chronic Progression of Neurodegeneration in a Mouse Model of Parkinson's Disease. Environ. Health Perspect. 119, 807–814. https://doi.org/10.1289/ehp.1003013
- Gardet, A., Benita, Y., Li, C., Sands, B.E., Ballester, I., Stevens, C., Korzenik, J.R., Rioux, J.D., Daly, M.J., Xavier, R.J., Podolsky, D.K., 2010. LRRK2 Is Involved in the IFN-γ Response and Host Response to Pathogens. J. Immunol. 185, 5577–5585. https://doi.org/10.4049/jimmunol.1000548
- Gautier, C.A., Kitada, T., Shen, J., 2008. Loss of PINK1 causes mitochondrial functional defects and increased sensitivity to oxidative stress. Proc. Natl. Acad. Sci. 105, 11364–11369. https://doi.org/10.1073/pnas.0802076105
- George, J.M., 2002. The synucleins. Genome Biol. 3, reviews3002.1-reviews3002.6.
- Gerhard, A., Pavese, N., Hotton, G., Turkheimer, F., Es, M., Hammers, A., Eggert, K., Oertel, W., Banati, R.B., Brooks, D.J., 2006. In vivo imaging of microglial activation with [11C](R)-PK11195 PET in idiopathic Parkinson's disease. Neurobiol. Dis. 21, 404– 412. https://doi.org/10.1016/j.nbd.2005.08.002
- Giasson, B.I., Duda, J.E., Murray, I.V.J., Chen, Q., Souza, J.M., Hurtig, H.I., Ischiropoulos, H., Trojanowski, J.Q., -Y. Lee, V.M., 2000. Oxidative Damage Linked to Neurodegeneration by Selective α-Synuclein Nitration in Synucleinopathy Lesions. Science 290, 985–989. https://doi.org/10.1126/science.290.5493.985

- Gleiter, C.H., Jägle, C., Gresser, U., Mörike, K., 2004. Candesartan. Cardiovasc. Drug Rev. 22, 263–284. https://doi.org/10.1111/j.1527-3466.2004.tb00146.x
- Glinka, Y., Gassen, M., Youdim, M.B.H., 1997. Mechanism of 6-hydroxydopamine neurotoxicity, in: Riederer, P., Calne, D.B., Horowski, R., Mizuno, Y., Poewe, W., Youdim, M.B.H. (Eds.), Advances in Research on Neurodegeneration, Journal of Neural Transmission. Supplementa. Springer, Vienna, pp. 55–66. https://doi.org/10.1007/978-3-7091-6842-4_7
- Greenamyre, J.T., Sherer, T.B., Betarbet, R., Panov, A.V., 2001. Complex I and Parkinson's Disease. IUBMB Life 52, 135–141. https://doi.org/10.1080/15216540152845939
- Greenland, J.C., Cutting, E., Kadyan, S., Bond, S., Chhabra, A., Williams-Gray, C.H., 2020. Azathioprine immunosuppression and disease modification in Parkinson's disease (AZA-PD): a randomised double-blind placebo-controlled phase II trial protocol. BMJ Open 10, e040527. https://doi.org/10.1136/bmjopen-2020-040527
- Griendling, K.K., Ushio-Fukai, M., 2000. Reactive oxygen species as mediators of angiotensin II signaling. Regul. Pept. 91, 21–27. https://doi.org/10.1016/S0167-0115(00)00136-1
- Grozdanov, V., Bliederhaeuser, C., Ruf, W.P., Roth, V., Fundel-Clemens, K., Zondler, L., Brenner, D., Martin-Villalba, A., Hengerer, B., Kassubek, J., Ludolph, A.C., Weishaupt, J.H., Danzer, K.M., 2014. Inflammatory dysregulation of blood monocytes in Parkinson's disease patients. Acta Neuropathol. (Berl.) 128, 651–663. https://doi.org/10.1007/s00401-014-1345-4
- Grozdanov, V., Bousset, L., Hoffmeister, M., Bliederhaeuser, C., Meier, C., Madiona, K., Pieri, L., Kiechle, M., McLean, P.J., Kassubek, J., Behrends, C., Ludolph, A.C., Weishaupt, J.H., Melki, R., Danzer, K.M., 2019. Increased Immune Activation by Pathologic α-Synuclein in Parkinson's Disease. Ann. Neurol. 86, 593–606. https://doi.org/10.1002/ana.25557
- Gründemann, J., Schlaudraff, F., Haeckel, O., Liss, B., 2008. Elevated α-synuclein mRNA levels in individual UV-laser-microdissected dopaminergic substantia nigra neurons in idiopathic Parkinson's disease. Nucleic Acids Res. 36, e38. https://doi.org/10.1093/nar/gkn084
- Gustot, A., Gallea, J.I., Sarroukh, R., Celej, M.S., Ruysschaert, J.-M., Raussens, V., 2015. Amyloid fibrils are the molecular trigger of inflammation in Parkinson's disease. Biochem. J. 471, 323–333. https://doi.org/10.1042/BJ20150617
- Haehner, A., Hummel, T., Hummel, C., Sommer, U., Junghanns, S., Reichmann, H., 2007. Olfactory loss may be a first sign of idiopathic Parkinson's disease. Mov. Disord. 22, 839–842. https://doi.org/10.1002/mds.21413
- Hajjar, I., Okafor, M., Wan, L., Yang, Z., Nye, J.A., Bohsali, A., Shaw, L.M., Levey, A.I., Lah, J.J., Calhoun, V.D., Moore, R.H., Goldstein, F.C., 2022. Safety and biomarker effects of candesartan in non-hypertensive adults with prodromal Alzheimer's disease. Brain Commun. 4, fcac270. https://doi.org/10.1093/braincomms/fcac270
- Hakimi, M., Selvanantham, T., Swinton, E., Padmore, R.F., Tong, Y., Kabbach, G., Venderova, K., Girardin, S.E., Bulman, D.E., Scherzer, C.R., LaVoie, M.J., Gris, D.,

Park, D.S., Angel, J.B., Shen, J., Philpott, D.J., Schlossmacher, M.G., 2011. Parkinson's disease-linked LRRK2 is expressed in circulating and tissue immune cells and upregulated following recognition of microbial structures. J. Neural Transm. 118, 795–808. https://doi.org/10.1007/s00702-011-0653-2

- Hancock, C.R., Han, D.-H., Chen, M., Terada, S., Yasuda, T., Wright, D.C., Holloszy, J.O., 2008. High-fat diets cause insulin resistance despite an increase in muscle mitochondria. Proc. Natl. Acad. Sci. 105, 7815–7820. https://doi.org/10.1073/pnas.0802057105
- Hansen, C., Angot, E., Bergström, A.-L., Steiner, J.A., Pieri, L., Paul, G., Outeiro, T.F., Melki, R., Kallunki, P., Fog, K., Li, J.-Y., Brundin, P., 2011. α-Synuclein propagates from mouse brain to grafted dopaminergic neurons and seeds aggregation in cultured human cells. J. Clin. Invest. 121, 715–725. https://doi.org/10.1172/JCI43366
- Harms, A.S., Barnum, C.J., Ruhn, K.A., Varghese, S., Treviño, I., Blesch, A., Tansey, M.G., 2011. Delayed Dominant-Negative TNF Gene Therapy Halts Progressive Loss of Nigral Dopaminergic Neurons in a Rat Model of Parkinson's Disease. Mol. Ther. 19, 46–52. https://doi.org/10.1038/mt.2010.217
- Harms, A.S., Cao, S., Rowse, A.L., Thome, A.D., Li, X., Mangieri, L.R., Cron, R.Q., Shacka, J.J., Raman, C., Standaert, D.G., 2013. MHCII Is Required for α-Synuclein-Induced Activation of Microglia, CD4 T Cell Proliferation, and Dopaminergic Neurodegeneration. J. Neurosci. 33, 9592–9600. https://doi.org/10.1523/JNEUROSCI.5610-12.2013
- Harms, A.S., Delic, V., Thome, A.D., Bryant, N., Liu, Z., Chandra, S., Jurkuvenaite, A., West, A.B., 2017. α-Synuclein fibrils recruit peripheral immune cells in the rat brain prior to neurodegeneration. Acta Neuropathol. Commun. 5, 85. https://doi.org/10.1186/s40478-017-0494-9
- Hasan, U., Chaffois, C., Gaillard, C., Saulnier, V., Merck, E., Tancredi, S., Guiet, C., Brière, F., Vlach, J., Lebecque, S., Trinchieri, G., Bates, E.E.M., 2005. Human TLR10 Is a Functional Receptor, Expressed by B Cells and Plasmacytoid Dendritic Cells, Which Activates Gene Transcription through MyD88. J. Immunol. 174, 2942–2950. https://doi.org/10.4049/jimmunol.174.5.2942
- Hasegawa, Y., Inagaki, T., Sawada, M., Suzumura, A., 2000. Impaired cytokine production by peripheral blood mononuclear cells and monocytes/macrophages in Parkinson's disease. Acta Neurol. Scand. 101, 159–164. https://doi.org/10.1034/j.1600-0404.2000.101003159.x
- Hashikawa-Hobara, N., Hashikawa, N., Inoue, Y., Sanda, H., Zamami, Y., Takatori, S., Kawasaki, H., 2012. Candesartan Cilexetil Improves Angiotensin II Type 2 Receptor– Mediated Neurite Outgrowth via the PI3K-Akt Pathway in Fructose-Induced Insulin-Resistant Rats. Diabetes 61, 925–932. https://doi.org/10.2337/db11-1468
- Hawkes, C.H., Del Tredici, K., Braak, H., 2007. Parkinson's disease: a dual-hit hypothesis. Neuropathol. Appl. Neurobiol. 33, 599–614. https://doi.org/10.1111/j.1365-2990.2007.00874.x

- Hely, M.A., Reid, W.G.J., Adena, M.A., Halliday, G.M., Morris, J.G.L., 2008. The Sydney multicenter study of Parkinson's disease: the inevitability of dementia at 20 years. Mov. Disord. Off. J. Mov. Disord. Soc. 23, 837–844. https://doi.org/10.1002/mds.21956
- Herkenham, M., Little, M.D., Bankiewicz, K., Yang, S.C., Markey, S.P., Johannessen, J.N., 1991. Selective retention of MPP+ within the monoaminergic systems of the primate brain following MPTP administration: an in vivo autoradiographic study. Neuroscience 40, 133–158. https://doi.org/10.1016/0306-4522(91)90180-v
- Hermida-Ameijeiras, Á., Méndez-Álvarez, E., Sánchez-Iglesias, S., Sanmartín-Suárez, C., Soto-Otero, R., 2004. Autoxidation and MAO-mediated metabolism of dopamine as a potential cause of oxidative stress: role of ferrous and ferric ions. Neurochem. Int. 45, 103–116. https://doi.org/10.1016/j.neuint.2003.11.018
- Herrera, A.J., Castaño, A., Venero, J.L., Cano, J., Machado, A., 2000. The Single Intranigral Injection of LPS as a New Model for Studying the Selective Effects of Inflammatory Reactions on Dopaminergic System. Neurobiol. Dis. 7, 429–447. https://doi.org/10.1006/nbdi.2000.0289
- Hilton, D., Stephens, M., Kirk, L., Edwards, P., Potter, R., Zajicek, J., Broughton, E., Hagan, H., Carroll, C., 2014. Accumulation of α-synuclein in the bowel of patients in the preclinical phase of Parkinson's disease. Acta Neuropathol. (Berl.) 127, 235–241. https://doi.org/10.1007/s00401-013-1214-6
- Hoebe, K., Beutler, B., 2004. LPS, dsRNA and the interferon bridge to adaptive immune responses: Trif, Tram, and other TIR adaptor proteins. J. Endotoxin Res. 10, 130–136. https://doi.org/10.1179/096805104225004031
- Hoffmann, A., Ettle, B., Bruno, A., Kulinich, A., Hoffmann, A.-C., von Wittgenstein, J., Winkler, J., Xiang, W., Schlachetzki, J.C.M., 2016. Alpha-synuclein activates BV2 microglia dependent on its aggregation state. Biochem. Biophys. Res. Commun. 479, 881–886. https://doi.org/10.1016/j.bbrc.2016.09.109
- Höglinger, G.U., Alvarez-Fischer, D., Arias-Carrión, O., Djufri, M., Windolph, A., Keber, U., Borta, A., Ries, V., Schwarting, R.K.W., Scheller, D., Oertel, W.H., 2015. A new dopaminergic nigro-olfactory projection. Acta Neuropathol. (Berl.) 130, 333–348. https://doi.org/10.1007/s00401-015-1451-y
- Holmans, P., Moskvina, V., Jones, L., Sharma, M., International Parkinson's Disease Genomics Consortium, Vedernikov, A., Buchel, F., Saad, M., Sadd, M., Bras, J.M., Bettella, F., Nicolaou, N., Simón-Sánchez, J., Mittag, F., Gibbs, J.R., Schulte, C., Durr, A., Guerreiro, R., Hernandez, D., Brice, A., Stefánsson, H., Majamaa, K., Gasser, T., Heutink, P., Wood, N.W., Martinez, M., Singleton, A.B., Nalls, M.A., Hardy, J., Morris, H.R., Williams, N.M., 2013. A pathway-based analysis provides additional support for an immune-related genetic susceptibility to Parkinson's disease. Hum. Mol. Genet. 22, 1039–1049. https://doi.org/10.1093/hmg/dds492
- Hong, W., Wang, Z., Liu, W., O'Malley, T.T., Jin, M., Willem, M., Haass, C., Frosch, M.P., Walsh, D.M., 2018. Diffusible, highly bioactive oligomers represent a critical minority of soluble Aβ in Alzheimer's disease brain. Acta Neuropathol. (Berl.) 136, 19–40. https://doi.org/10.1007/s00401-018-1846-7

- Hornung, V., Rothenfusser, S., Britsch, S., Krug, A., Jahrsdörfer, B., Giese, T., Endres, S., Hartmann, G., 2002. Quantitative Expression of Toll-Like Receptor 1–10 mRNA in Cellular Subsets of Human Peripheral Blood Mononuclear Cells and Sensitivity to CpG Oligodeoxynucleotides. J. Immunol. 168, 4531–4537. https://doi.org/10.4049/jimmunol.168.9.4531
- Hsu, L.J., Sagara, Y., Arroyo, A., Rockenstein, E., Sisk, A., Mallory, M., Wong, J., Takenouchi, T., Hashimoto, M., Masliah, E., 2000. α-Synuclein Promotes Mitochondrial Deficit and Oxidative Stress. Am. J. Pathol. 157, 401–410. https://doi.org/10.1016/S0002-9440(10)64553-1
- Hu, G., Jousilahti, P., Bidel, S., Antikainen, R., Tuomilehto, J., 2007. Type 2 Diabetes and the Risk of Parkinson's Disease. Diabetes Care 30, 842–847. https://doi.org/10.2337/dc06-2011
- Huang, M., Li, Y., Tian, T., Wang, K., Wang, Y., Yan, W., Yang, H., 2020. Knockdown of TLR4 Represses the Paraquat-Induced Neuroinflammation and Microglial M1 Polarization. Neurotox. Res. 38, 741–750. https://doi.org/10.1007/s12640-020-00261-6
- Hughes, A.J., Daniel, S.E., Blankson, S., Lees, A.J., 1993. A Clinicopathologic Study of 100 Cases of Parkinson's Disease. Arch. Neurol. 50, 140–148. https://doi.org/10.1001/archneur.1993.00540020018011
- Hughes, C.D., Choi, M.L., Ryten, M., Hopkins, L., Drews, A., Botía, J.A., Iljina, M., Rodrigues, M., Gagliano, S.A., Gandhi, S., Bryant, C., Klenerman, D., 2019. Picomolar concentrations of oligomeric alpha-synuclein sensitizes TLR4 to play an initiating role in Parkinson's disease pathogenesis. Acta Neuropathol. (Berl.) 137, 103–120. https://doi.org/10.1007/s00401-018-1907-y
- Iannaccone, S., Cerami, C., Alessio, M., Garibotto, V., Panzacchi, A., Olivieri, S., Gelsomino, G., Moresco, R.M., Perani, D., 2013. In vivo microglia activation in very early dementia with Lewy bodies, comparison with Parkinson's disease. Parkinsonism Relat. Disord. 19, 47–52. https://doi.org/10.1016/j.parkreldis.2012.07.002
- Ii, M., Matsunaga, N., Hazeki, K., Nakamura, K., Takashima, K., Seya, T., Hazeki, O., Kitazaki, T., Iizawa, Y., 2006. A Novel Cyclohexene Derivative, Ethyl (6R)-6-[N-(2-Chloro-4-fluorophenyl)sulfamoyl]cyclohex-1-ene-1-carboxylate (TAK-242), Selectively Inhibits Toll-Like Receptor 4-Mediated Cytokine Production through Suppression of Intracellular Signaling. Mol. Pharmacol. 69, 1288–1295. https://doi.org/10.1124/mol.105.019695
- Imai, Y., Ibata, I., Ito, D., Ohsawa, K., Kohsaka, S., 1996. A Novel Geneiba1in the Major Histocompatibility Complex Class III Region Encoding an EF Hand Protein Expressed in a Monocytic Lineage. Biochem. Biophys. Res. Commun. 224, 855–862. https://doi.org/10.1006/bbrc.1996.1112
- International Parkinson Disease Genomics Consortium, Nalls, M.A., Plagnol, V., Hernandez, D.G., Sharma, M., Sheerin, U.-M., Saad, M., Simón-Sánchez, J., Schulte, C., Lesage, S., Sveinbjörnsdóttir, S., Stefánsson, K., Martinez, M., Hardy, J., Heutink, P., Brice, A., Gasser, T., Singleton, A.B., Wood, N.W., 2011. Imputation of sequence variants for

identification of genetic risks for Parkinson's disease: a meta-analysis of genome-wide association studies. Lancet Lond. Engl. 377, 641–649. https://doi.org/10.1016/S0140-6736(10)62345-8

- Ito, D., Imai, Y., Ohsawa, K., Nakajima, K., Fukuuchi, Y., Kohsaka, S., 1998. Microgliaspecific localisation of a novel calcium binding protein, Iba1. Mol. Brain Res. 57, 1–9. https://doi.org/10.1016/S0169-328X(98)00040-0
- Iwaki, H., Blauwendraat, C., Leonard, H.L., Makarious, M.B., Kim, J.J., Liu, G., Maple-Grødem, J., Corvol, J.-C., Pihlstrøm, L., van Nimwegen, M., Smolensky, L., Amondikar, N., Hutten, S.J., Frasier, M., Nguyen, K.-D.H., Rick, J., Eberly, S., Faghri, F., Auinger, P., Scott, K.M., Wijeyekoon, R., Van Deerlin, V.M., Hernandez, D.G., Gibbs, R.J., Day-Williams, A.G., Brice, A., Alves, G., Noyce, A.J., Tysnes, O.-B., Evans, J.R., Breen, D.P., Estrada, K., Wegel, C.E., Danjou, F., Simon, D.K., Andreassen, O.A., Ravina, B., Toft, M., Heutink, P., Bloem, B.R., Weintraub, D., Barker, R.A., Williams-Gray, C.H., van de Warrenburg, B.P., Van Hilten, J.J., Scherzer, C.R., Singleton, A.B., Nalls, M.A., 2021. Differences in the Presentation and Progression of Parkinson's Disease by Sex. Mov. Disord. Off. J. Mov. Disord. Soc. 36, 106–117. https://doi.org/10.1002/mds.28312
- Jack, C.S., Arbour, N., Manusow, J., Montgrain, V., Blain, M., McCrea, E., Shapiro, A., Antel, J.P., 2005. TLR Signaling Tailors Innate Immune Responses in Human Microglia and Astrocytes. J. Immunol. 175, 4320–4330. https://doi.org/10.4049/jimmunol.175.7.4320
- Jackson-Lewis, V., Przedborski, S., 2007. Protocol for the MPTP mouse model of Parkinson's disease. Nat. Protoc. 2, 141–151. https://doi.org/10.1038/nprot.2006.342
- Jang, Y., Koo, J.-H., Kwon, I., Kang, E.-B., Um, H.-S., Soya, H., Lee, Y., Cho, J.-Y., 2017. Neuroprotective effects of endurance exercise against neuroinflammation in MPTPinduced Parkinson's disease mice. Brain Res. 1655, 186–193. https://doi.org/10.1016/j.brainres.2016.10.029
- Jenner, P., 2003. Oxidative stress in Parkinson's disease. Ann. Neurol. 53, S26–S38. https://doi.org/10.1002/ana.10483
- Jin, D., Takai, S., Sugiyama, T., Hayashi, T., Fukumoto, M., Oku, H., Kitaura, Y., Ikeda, T., Miyazaki, M., 2009. Long-Term Angiotensin II Blockade May Improve Not Only Hyperglycemia but Also Age-Associated Cardiac Fibrosis. J. Pharmacol. Sci. 109, 275–284. https://doi.org/10.1254/jphs.08210FP
- Jin, M.S., Kim, S.E., Heo, J.Y., Lee, M.E., Kim, H.M., Paik, S.-G., Lee, H., Lee, J.-O., 2007. Crystal structure of the TLR1-TLR2 heterodimer induced by binding of a tri-acylated lipopeptide. Cell 130, 1071–1082. https://doi.org/10.1016/j.cell.2007.09.008
- Jun, H.-S., Hwang, K., Kim, Y., Park, T., 2008. High-fat Diet Alters PP2A, TC10, and CIP4 Expression in Visceral Adipose Tissue of Rats. Obesity 16, 1226–1231. https://doi.org/10.1038/oby.2008.220
- Jurga, A.M., Paleczna, M., Kuter, K.Z., 2020. Overview of General and Discriminating Markers of Differential Microglia Phenotypes. Front. Cell. Neurosci. 14.

- Kang, J.Y., Nan, X., Jin, M.S., Youn, S.-J., Ryu, Y.H., Mah, S., Han, S.H., Lee, H., Paik, S.-G., Lee, J.-O., 2009. Recognition of Lipopeptide Patterns by Toll-like Receptor 2-Tolllike Receptor 6 Heterodimer. Immunity 31, 873–884. https://doi.org/10.1016/j.immuni.2009.09.018
- Karachi, C., Grabli, D., Bernard, F.A., Tandé, D., Wattiez, N., Belaid, H., Bardinet, E., Prigent, A., Nothacker, H.-P., Hunot, S., Hartmann, A., Lehéricy, S., Hirsch, E.C., François, C., 2010. Cholinergic mesencephalic neurons are involved in gait and postural disorders in Parkinson disease. J. Clin. Invest. 120, 2745–2754. https://doi.org/10.1172/JCI42642
- Karampetsou, M., Ardah, M.T., Semitekolou, M., Polissidis, A., Samiotaki, M., Kalomoiri, M., Majbour, N., Xanthou, G., El-Agnaf, O.M.A., Vekrellis, K., 2017. Phosphorylated exogenous alpha-synuclein fibrils exacerbate pathology and induce neuronal dysfunction in mice. Sci. Rep. 7, 16533. https://doi.org/10.1038/s41598-017-15813-8
- Kass, M.D., Czarnecki, L.A., Moberly, A.H., McGann, J.P., 2017. Differences in peripheral sensory input to the olfactory bulb between male and female mice. Sci. Rep. 7, 45851. https://doi.org/10.1038/srep45851
- Kawai, T., Akira, S., 2011. Toll-like receptors and their crosstalk with other innate receptors in infection and immunity. Immunity 34, 637–650. https://doi.org/10.1016/j.immuni.2011.05.006
- Kawamoto, T., Ii, M., Kitazaki, T., Iizawa, Y., Kimura, H., 2008. TAK-242 selectively suppresses Toll-like receptor 4-signaling mediated by the intracellular domain. Eur. J. Pharmacol. 584, 40–48. https://doi.org/10.1016/j.ejphar.2008.01.026
- Keshavarzian, A., Green, S.J., Engen, P.A., Voigt, R.M., Naqib, A., Forsyth, C.B., Mutlu, E., Shannon, K.M., 2015. Colonic bacterial composition in Parkinson's disease. Mov. Disord. 30, 1351–1360. https://doi.org/10.1002/mds.26307
- Khan, M.M., Kempuraj, D., Thangavel, R., Zaheer, A., 2013. Protection of MPTP-induced neuroinflammation and neurodegeneration by Pycnogenol. Neurochem. Int. 62, 379–388. https://doi.org/10.1016/j.neuint.2013.01.029
- Kim, C., Ho, D.-H., Suk, J.-E., You, S., Michael, S., Kang, J., Joong Lee, S., Masliah, E., Hwang, D., Lee, H.-J., Lee, S.-J., 2013. Neuron-released oligomeric α-synuclein is an endogenous agonist of TLR2 for paracrine activation of microglia. Nat. Commun. 4, 1562. https://doi.org/10.1038/ncomms2534
- Kim, C., Kwon, S., Iba, M., Spencer, B., Rockenstein, E., Mante, M., Adame, A., Shin, S.J., Fields, J.A., Rissman, R.A., Lee, S.-J., Masliah, E., 2021. Effects of innate immune receptor stimulation on extracellular α-synuclein uptake and degradation by brain resident cells. Exp. Mol. Med. 53, 281–290. https://doi.org/10.1038/s12276-021-00562-6
- Kim, C., Lee, H.-J., Masliah, E., Lee, S.-J., 2016. Non-cell-autonomous Neurotoxicity of αsynuclein Through Microglial Toll-like Receptor 2. Exp. Neurobiol. 25, 113–119. https://doi.org/10.5607/en.2016.25.3.113
- Kim, C., Rockenstein, E., Spencer, B., Kim, H.-K., Adame, A., Trejo, M., Stafa, K., Lee, H.-J., Lee, S.-J., Masliah, E., 2015. Antagonizing Neuronal Toll-like Receptor 2 Prevents

Synucleinopathy by Activating Autophagy. Cell Rep. 13, 771–782. https://doi.org/10.1016/j.celrep.2015.09.044

- Kim, C., Spencer, B., Rockenstein, E., Yamakado, H., Mante, M., Adame, A., Fields, J.A., Masliah, D., Iba, M., Lee, H.-J., Rissman, R.A., Lee, S.-J., Masliah, E., 2018. Immunotherapy targeting toll-like receptor 2 alleviates neurodegeneration in models of synucleinopathy by modulating α-synuclein transmission and neuroinflammation. Mol. Neurodegener. 13, 43. https://doi.org/10.1186/s13024-018-0276-2
- Kim, R., Kim, H.-J., Shin, J.H., Lee, C.Y., Jeon, S.H., Jeon, B., 2022. Serum Inflammatory Markers and Progression of Nonmotor Symptoms in Early Parkinson's Disease. Mov. Disord. 37, 1535–1541. https://doi.org/10.1002/mds.29056
- Kim, S., Kwon, S.-H., Kam, T.-I., Panicker, N., Karuppagounder, S.S., Lee, S., Lee, J.H., Kim, W.R., Kook, M., Foss, C.A., Shen, C., Lee, H., Kulkarni, S., Pasricha, P.J., Lee, G., Pomper, M.G., Dawson, V.L., Dawson, T.M., Ko, H.S., 2019. Transneuronal Propagation of Pathologic α-Synuclein from the Gut to the Brain Models Parkinson's Disease. Neuron 103, 627-641.e7. https://doi.org/10.1016/j.neuron.2019.05.035
- Kingsbury, A.E., Daniel, S.E., Sangha, H., Eisen, S., Lees, A.J., Foster, O.J.F., 2004. Alteration in α-synuclein mRNA expression in Parkinson's disease. Mov. Disord. 19, 162–170. https://doi.org/10.1002/mds.10683
- Kirik, D., Annett, L.E., Burger, C., Muzyczka, N., Mandel, R.J., Björklund, A., 2003. Nigrostriatal α-synucleinopathy induced by viral vector-mediated overexpression of human α-synuclein: A new primate model of Parkinson's disease. Proc. Natl. Acad. Sci. 100, 2884–2889. https://doi.org/10.1073/pnas.0536383100
- Kirik, D., Rosenblad, C., Burger, C., Lundberg, C., Johansen, T.E., Muzyczka, N., Mandel, R.J., Björklund, A., 2002. Parkinson-Like Neurodegeneration Induced by Targeted Overexpression of α-Synuclein in the Nigrostriatal System. J. Neurosci. 22, 2780–2791. https://doi.org/10.1523/JNEUROSCI.22-07-02780.2002
- Klegeris, A., Pelech, S., Giasson, B.I., Maguire, J., Zhang, H., McGeer, E.G., McGeer, P.L., 2008. α-Synuclein activates stress signaling protein kinases in THP-1 cells and microglia. Neurobiol. Aging 29, 739–752. https://doi.org/10.1016/j.neurobiolaging.2006.11.013
- Klein, S.L., Flanagan, K.L., 2016. Sex differences in immune responses. Nat. Rev. Immunol. 16, 626–638. https://doi.org/10.1038/nri.2016.90
- Klingelhoefer, L., Reichmann, H., 2015. Pathogenesis of Parkinson disease--the gut-brain axis and environmental factors. Nat. Rev. Neurol. 11, 625–636. https://doi.org/10.1038/nrneurol.2015.197
- Knott, C., Stern, G., Wilkin, G.P., 2000. Inflammatory Regulators in Parkinson's Disease: iNOS, Lipocortin-1, and Cyclooxygenases-1 and -2. Mol. Cell. Neurosci. 16, 724–739. https://doi.org/10.1006/mcne.2000.0914
- Koo, J.-H., Jang, Y.-C., Hwang, D.-J., Um, H.-S., Lee, N.-H., Jung, J.-H., Cho, J.-Y., 2017. Treadmill exercise produces neuroprotective effects in a murine model of Parkinson's

disease by regulating the TLR2/MyD88/NF-κB signaling pathway. Neuroscience 356, 102–113. https://doi.org/10.1016/j.neuroscience.2017.05.016

- Kordower, J.H., 2003. In vivo gene delivery of glial cell line–derived neurotrophic factor for Parkinson's disease. Ann. Neurol. 53, S120–S134. https://doi.org/10.1002/ana.10485
- Kordower, J.H., Chu, Y., Hauser, R.A., Freeman, T.B., Olanow, C.W., 2008. Lewy body–like pathology in long-term embryonic nigral transplants in Parkinson's disease. Nat. Med. 14, 504–506. https://doi.org/10.1038/nm1747
- Kordower, J.H., Dodiya, H.B., Kordower, A.M., Terpstra, B., Paumier, K., Madhavan, L., Sortwell, C., Steece-Collier, K., Collier, T.J., 2011. Transfer of host-derived α synuclein to grafted dopaminergic neurons in rat. Neurobiol. Dis. 43, 552–557. https://doi.org/10.1016/j.nbd.2011.05.001
- Kordower, J.H., Freeman, T.B., Snow, B.J., Vingerhoets, F.J., Mufson, E.J., Sanberg, P.R., Hauser, R.A., Smith, D.A., Nauert, G.M., Perl, D.P., 1995. Neuropathological evidence of graft survival and striatal reinnervation after the transplantation of fetal mesencephalic tissue in a patient with Parkinson's disease. N. Engl. J. Med. 332, 1118– 1124. https://doi.org/10.1056/NEJM199504273321702
- Kotagal, V., Albin, R.L., Müller, M.L.T.M., Koeppe, R.A., Frey, K.A., Bohnen, N.I., 2013. Diabetes is associated with postural instability and gait difficulty in Parkinson disease. Parkinsonism Relat. Disord. 19, 522–526. https://doi.org/10.1016/j.parkreldis.2013.01.016
- Kouli, A., Camacho, M., Allinson, K., Williams-Gray, C.H., 2020. Neuroinflammation and protein pathology in Parkinson's disease dementia. Acta Neuropathol. Commun. 8, 211. https://doi.org/10.1186/s40478-020-01083-5
- Kouli, A., Horne, C.B., Williams-Gray, C.H., 2019. Toll-like receptors and their therapeutic potential in Parkinson's disease and α-synucleinopathies. Brain. Behav. Immun. 81, 41–51. https://doi.org/10.1016/j.bbi.2019.06.042
- Krebiehl, G., Ruckerbauer, S., Burbulla, L.F., Kieper, N., Maurer, B., Waak, J., Wolburg, H., Gizatullina, Z., Gellerich, F.N., Woitalla, D., Riess, O., Kahle, P.J., Proikas-Cezanne, T., Krüger, R., 2010. Reduced basal autophagy and impaired mitochondrial dynamics due to loss of Parkinson's disease-associated protein DJ-1. PloS One 5, e9367. https://doi.org/10.1371/journal.pone.0009367
- Kuan, W., Alfaidi, M., Horne, C.B., Vallin, B., Fox, S., Fazal, S.V., Williams-Gray, C.H., Barker, R.A., 2022. Selective neurodegeneration generated by intravenous α-synuclein pre-formed fibril administration is not associated with endogenous α-synuclein levels in the rat brain. Brain Pathol. 33, e13128. https://doi.org/10.1111/bpa.13128
- Kuan, W.-L., Poole, E., Fletcher, M., Karniely, S., Tyers, P., Wills, M., Barker, R.A., Sinclair, J.H., 2011. A novel neuroprotective therapy for Parkinson's disease using a viral noncoding RNA that protects mitochondrial Complex I activity. J. Exp. Med. 209, 1– 10. https://doi.org/10.1084/jem.20111126
- Kuan, W.-L., Stott, K., He, X., Wood, T.C., Yang, S., Kwok, J.C.F., Hall, K., Zhao, Y., Tietz, O., Aigbirhio, F.I., Vernon, A.C., Barker, R.A., 2019. Systemic α-synuclein injection

triggers selective neuronal pathology as seen in patients with Parkinson's disease. Mol. Psychiatry 1–12. https://doi.org/10.1038/s41380-019-0608-9

- Kulisevsky, J., Martínez-Horta, S., Campolongo, A., Pascual-Sedano, B., Marín-Lahoz, J., Bejr-kasem, H., Labandeira-Garcia, J.L., Lanciego, J.L., Puig-Davi, A., Horta-Barba, A., Pagonabarraga, J., Rodríguez-Antigüedad, J., 2023. A randomized clinical trial of candesartan for cognitive impairment in Parkinson's disease. Parkinsonism Relat. Disord. 110. https://doi.org/10.1016/j.parkreldis.2023.105367
- Kumar, P., Wu, H., McBride, J.L., Jung, K.-E., Hee Kim, M., Davidson, B.L., Kyung Lee, S., Shankar, P., Manjunath, N., 2007. Transvascular delivery of small interfering RNA to the central nervous system. Nature 448, 39–43. https://doi.org/10.1038/nature05901
- Kupsky, W.J., Grimes, M.M., Sweeting, J., Bertsch, R., Cote, L.J., 1987. Parkinson's disease and megacolon: Concentric hyaline inclusions (Lewy bodies) in enteric ganglion cells. Neurology 37, 1253–1253. https://doi.org/10.1212/WNL.37.7.1253
- La Vitola, P., Balducci, C., Cerovic, M., Santamaria, G., Brandi, E., Grandi, F., Caldinelli, L., Colombo, L., Morgese, M.G., Trabace, L., Pollegioni, L., Albani, D., Forloni, G., 2018.
 Alpha-synuclein oligomers impair memory through glial cell activation and via Tolllike receptor 2. Brain. Behav. Immun. 69, 591–602. https://doi.org/10.1016/j.bbi.2018.02.012
- Labandeira, C.M., Pedrosa, M.A., Quijano, A., Valenzuela, R., Garrido-Gil, P., Sanchez-Andrade, M., Suarez-Quintanilla, J.A., Rodriguez-Perez, A.I., Labandeira-Garcia, J.L., 2022. Angiotensin type-1 receptor and ACE2 autoantibodies in Parkinson's disease. Npj Park. Dis. 8, 1–12. https://doi.org/10.1038/s41531-022-00340-9
- Lafon, M., Megret, F., Lafage, M., Prehaud, C., 2006. The innate immune facet of brain. J. Mol. Neurosci. 29, 185–194. https://doi.org/10.1385/JMN:29:3:185
- Langston, J.W., Ballard, P., Tetrud, J.W., Irwin, I., 1983. Chronic Parkinsonism in humans due to a product of meperidine-analog synthesis. Science 219, 979–980. https://doi.org/10.1126/science.6823561
- Lawson, L.J., Perry, V.H., Dri, P., Gordon, S., 1990. Heterogeneity in the distribution and morphology of microglia in the normal adult mouse brain. Neuroscience 39, 151–170. https://doi.org/10.1016/0306-4522(90)90229-w
- Le, W., Rowe, D., Xie, W., Ortiz, I., He, Y., Appel, S.H., 2001. Microglial activation and dopaminergic cell injury: an in vitro model relevant to Parkinson's disease. J. Neurosci. Off. J. Soc. Neurosci. 21, 8447–8455.
- Lebouvier, T., Chaumette, T., Damier, P., Coron, E., Touchefeu, Y., Vrignaud, S., Naveilhan, P., Galmiche, J.-P., Varannes, S.B. des, Derkinderen, P., Neunlist, M., 2008. Pathological lesions in colonic biopsies during Parkinson's disease. Gut 57, 1741– 1743. https://doi.org/10.1136/gut.2008.162503
- Lee, C.S., Samii, A., Sossi, V., Ruth, T.J., Schulzer, M., Holden, J.E., Wudel, J., Pal, P.K., de la Fuente-Fernandez, R., Calne, D.B., Stoessl, A.J., 2000. In vivo positron emission tomographic evidence for compensatory changes in presynaptic dopaminergic nerve terminals in Parkinson's disease. Ann. Neurol. 47, 493–503.

- Lee, H.-J., Cho, E.-D., Lee, K.W., Kim, J.-H., Cho, S.-G., Lee, S.-J., 2013. Autophagic failure promotes the exocytosis and intercellular transfer of α-synuclein. Exp. Mol. Med. 45, e22. https://doi.org/10.1038/emm.2013.45
- Lee, H.-J., Suk, J.-E., Bae, E.-J., Lee, S.-J., 2008. Clearance and deposition of extracellular alpha-synuclein aggregates in microglia. Biochem. Biophys. Res. Commun. 372, 423–428. https://doi.org/10.1016/j.bbrc.2008.05.045
- Lee, J.-W., Nam, H., Kim, L.E., Jeon, Y., Min, H., Ha, S., Lee, Y., Kim, S.-Y., Lee, S.J., Kim, E.-K., Yu, S.-W., 2019. TLR4 (toll-like receptor 4) activation suppresses autophagy through inhibition of FOXO3 and impairs phagocytic capacity of microglia. Autophagy 15, 753–770. https://doi.org/10.1080/15548627.2018.1556946
- Li, J.-Y., Englund, E., Holton, J.L., Soulet, D., Hagell, P., Lees, A.J., Lashley, T., Quinn, N.P., Rehncrona, S., Björklund, A., Widner, H., Revesz, T., Lindvall, O., Brundin, P., 2008. Lewy bodies in grafted neurons in subjects with Parkinson's disease suggest host-tograft disease propagation. Nat. Med. 14, 501–503. https://doi.org/10.1038/nm1746
- Li, X., Sundquist, J., Sundquist, K., 2012. Subsequent Risks of Parkinson Disease in Patients with Autoimmune and Related Disorders: A Nationwide Epidemiological Study from Sweden. Neurodegener. Dis. 10, 277–284. https://doi.org/10.1159/000333222
- Lindvall, O., Sawle, G., Widner, H., Rothwell, J.C., Björklund, A., Brooks, D., Brundin, P., Frackowiak, R., Marsden, C.D., Odin, P., Rehncrona, S., 1994. Evidence for long-term survival and function of dopaminergic grafts in progressive Parkinson's disease. Ann. Neurol. 35, 172–180. https://doi.org/10.1002/ana.410350208
- Liu, B., Dluzen, D.E., 2007. Oestrogen and Nigrostriatal Dopaminergic Neurodegeneration: Animal Models and Clinical Reports of Parkinson's Disease. Clin. Exp. Pharmacol. Physiol. 34, 555–565. https://doi.org/10.1111/j.1440-1681.2007.04616.x
- Liu, B., Fang, F., Pedersen, N.L., Tillander, A., Ludvigsson, J.F., Ekbom, A., Svenningsson, P., Chen, H., Wirdefeldt, K., 2017. Vagotomy and Parkinson disease: A Swedish register–based matched-cohort study. Neurology 88, 1996–2002. https://doi.org/10.1212/WNL.00000000003961
- Liu, C.-W., Corboy, M.J., DeMartino, G.N., Thomas, P.J., 2003. Endoproteolytic Activity of the Proteasome. Science 299, 408–411. https://doi.org/10.1126/science.1079293
- Liu, G., Zhang, C., Yin, J., Li, X., Cheng, F., Li, Y., Yang, H., Uéda, K., Chan, P., Yu, S., 2009. α-Synuclein is differentially expressed in mitochondria from different rat brain regions and dose-dependently down-regulates complex I activity. Neurosci. Lett. 454, 187–192. https://doi.org/10.1016/j.neulet.2009.02.056
- Liu, S., Liu, Y., Hao, W., Wolf, L., Kiliaan, A.J., Penke, B., Rübe, C.E., Walter, J., Heneka, M.T., Hartmann, T., Menger, M.D., Fassbender, K., 2012. TLR2 Is a Primary Receptor for Alzheimer's Amyloid β Peptide To Trigger Neuroinflammatory Activation. J. Immunol. 188, 1098–1107. https://doi.org/10.4049/jimmunol.1101121
- Liu, W.J., Ye, L., Huang, W.F., Guo, L.J., Xu, Z.G., Wu, H.L., Yang, C., Liu, H.F., 2016. p62 links the autophagy pathway and the ubiqutin–proteasome system upon ubiquitinated

protein degradation. Cell. Mol. Biol. Lett. 21, 29. https://doi.org/10.1186/s11658-016-0031-z

- Liu, Z., Chan, R.B., Cai, Z., Liu, X., Wu, Y., Yu, Z., Feng, T., Yang, Y., Zhang, J., 2022. α-Synuclein-containing erythrocytic extracellular vesicles: essential contributors to hyperactivation of monocytes in Parkinson's disease. J. Neuroinflammation 19, 53. https://doi.org/10.1186/s12974-022-02413-1
- Lo Bianco, C., Ridet, J.-L., Schneider, B.L., Déglon, N., Aebischer, P., 2002. α-Synucleinopathy and selective dopaminergic neuron loss in a rat lentiviral-based model of Parkinson's disease. Proc. Natl. Acad. Sci. 99, 10813–10818. https://doi.org/10.1073/pnas.152339799
- Loiarro, M., Sette, C., Gallo, G., Ciacci, A., Fantò, N., Mastroianni, D., Carminati, P., Ruggiero, V., 2005. Peptide-mediated Interference of TIR Domain Dimerization in MyD88 Inhibits Interleukin-1-dependent Activation of NF-κB *. J. Biol. Chem. 280, 15809–15814. https://doi.org/10.1074/jbc.C400613200
- Lu, X., Bing, G., Hagg, T., 2000. Naloxone prevents microglia-induced degeneration of dopaminergic substantia nigra neurons in adult rats. Neuroscience 97, 285–291. https://doi.org/10.1016/S0306-4522(00)00033-6
- Luk, K.C., Kehm, V., Carroll, J., Zhang, B., O'Brien, P., Trojanowski, J.Q., Lee, V.M.-Y., 2012. Pathological α-synuclein transmission initiates Parkinson-like neurodegeneration in nontransgenic mice. Science 338, 949–953. https://doi.org/10.1126/science.1227157
- Luk, K.C., Song, C., O'Brien, P., Stieber, A., Branch, J.R., Brunden, K.R., Trojanowski, J.Q., Lee, V.M.-Y., 2009. Exogenous alpha-synuclein fibrils seed the formation of Lewy body-like intracellular inclusions in cultured cells. Proc. Natl. Acad. Sci. U. S. A. 106, 20051–20056. https://doi.org/10.1073/pnas.0908005106
- Ma, Y., Li, J., Chiu, I., Wang, Y., Sloane, J.A., Lü, J., Kosaras, B., Sidman, R.L., Volpe, J.J., Vartanian, T., 2006. Toll-like receptor 8 functions as a negative regulator of neurite outgrowth and inducer of neuronal apoptosis. J. Cell Biol. 175, 209–215. https://doi.org/10.1083/jcb.200606016
- Mahase, E., 2022. Lecanemab trial finds slight slowing of cognitive decline, but clinical benefits are uncertain. BMJ 379, o2912. https://doi.org/10.1136/bmj.o2912
- Malek, N., Swallow, D., Grosset, K.A., Anichtchik, O., Spillantini, M., Grosset, D.G., 2014.
 Alpha-synuclein in peripheral tissues and body fluids as a biomarker for Parkinson's disease a systematic review. Acta Neurol. Scand. 130, 59–72. https://doi.org/10.1111/ane.12247
- Mariucci, G., Pagiotti, R., Galli, F., Romani, L., Conte, C., 2018. The Potential Role of Toll-Like Receptor 4 in Mediating Dopaminergic Cell Loss and Alpha-Synuclein Expression in the Acute MPTP Mouse Model of Parkinson's Disease. J. Mol. Neurosci. 64, 611–618. https://doi.org/10.1007/s12031-018-1057-7
- Marsden, C.D., 1990. Parkinson's disease. The Lancet, Originally published as Volume 1, Issue 8695 335, 948–949. https://doi.org/10.1016/0140-6736(90)91006-V
- Martin, H.L., Santoro, M., Mustafa, S., Riedel, G., Forrester, J.V., Teismann, P., 2016. Evidence for a role of adaptive immune response in the disease pathogenesis of the MPTP mouse model of Parkinson's disease. Glia 64, 386–395. https://doi.org/10.1002/glia.22935
- Martinez-Martin, P., Falup Pecurariu, C., Odin, P., van Hilten, J.J., Antonini, A., Rojo-Abuin, J.M., Borges, V., Trenkwalder, C., Aarsland, D., Brooks, D.J., Ray Chaudhuri, K., 2012. Gender-related differences in the burden of non-motor symptoms in Parkinson's disease. J. Neurol. 259, 1639–1647. https://doi.org/10.1007/s00415-011-6392-3
- McCabe, K., Concannon, R.M., McKernan, D.P., Dowd, E., 2017. Time-course of striatal Tolllike receptor expression in neurotoxic, environmental and inflammatory rat models of Parkinson's disease. J. Neuroimmunol. 310, 103–106. https://doi.org/10.1016/j.jneuroim.2017.07.007
- McClure, R., Massari, P., 2014. TLR-Dependent Human Mucosal Epithelial Cell Responses to Microbial Pathogens. Front. Immunol. 5, 386. https://doi.org/10.3389/fimmu.2014.00386
- McColl, B.W., Rothwell, N.J., Allan, S.M., 2007. Systemic inflammatory stimulus potentiates the acute phase and CXC chemokine responses to experimental stroke and exacerbates brain damage via interleukin-1- and neutrophil-dependent mechanisms. J. Neurosci. Off. J. Soc. Neurosci. 27, 4403–4412. https://doi.org/10.1523/JNEUROSCI.5376-06.2007
- McCoy, M.K., Martinez, T.N., Ruhn, K.A., Szymkowski, D.E., Smith, C.G., Botterman, B.R., Tansey, K.E., Tansey, M.G., 2006. Blocking Soluble Tumor Necrosis Factor Signaling with Dominant-Negative Tumor Necrosis Factor Inhibitor Attenuates Loss of Dopaminergic Neurons in Models of Parkinson's Disease. J. Neurosci. 26, 9365–9375. https://doi.org/10.1523/JNEUROSCI.1504-06.2006
- McCoy, M.K., Ruhn, K.A., Martinez, T.N., McAlpine, F.E., Blesch, A., Tansey, M.G., 2008. Intranigral Lentiviral Delivery of Dominant-negative TNF Attenuates Neurodegeneration and Behavioral Deficits in Hemiparkinsonian rats. Mol. Ther. 16, 1572–1579. https://doi.org/10.1038/mt.2008.146
- McGeer, P.L., Itagaki, S., Boyes, B.E., McGeer, E.G., 1988. Reactive microglia are positive for HLA-DR in the substantia nigra of Parkinson's and Alzheimer's disease brains. Neurology 38, 1285–1291. https://doi.org/10.1212/wnl.38.8.1285
- McNaught, K.S., Jenner, P., 2001. Proteasomal function is impaired in substantia nigra in Parkinson's disease. Neurosci. Lett. 297, 191–194. https://doi.org/10.1016/s0304-3940(00)01701-8
- McNaught, K.S.P., Belizaire, R., Isacson, O., Jenner, P., Olanow, C.W., 2003. Altered proteasomal function in sporadic Parkinson's disease. Exp. Neurol. 179, 38–46. https://doi.org/10.1006/exnr.2002.8050
- Metcalf, T.U., Cubas, R.A., Ghneim, K., Cartwright, M.J., Grevenynghe, J.V., Richner, J.M., Olagnier, D.P., Wilkinson, P.A., Cameron, M.J., Park, B.S., Hiscott, J.B., Diamond, M.S., Wertheimer, A.M., Nikolich-Zugich, J., Haddad, E.K., 2015. Global analyses

revealed age-related alterations in innate immune responses after stimulation of pathogen recognition receptors. Aging Cell 14, 421–432. https://doi.org/10.1111/acel.12320

- Miller, D.W., Hague, S.M., Clarimon, J., Baptista, M., Gwinn-Hardy, K., Cookson, M.R., Singleton, A.B., 2004. α-Synuclein in blood and brain from familial Parkinson disease with SNCA locus triplication. Neurology 62, 1835–1838. https://doi.org/10.1212/01.WNL.0000127517.33208.F4
- Moehle, M.S., Webber, P.J., Tse, T., Sukar, N., Standaert, D.G., DeSilva, T.M., Cowell, R.M., West, A.B., 2012. LRRK2 Inhibition Attenuates Microglial Inflammatory Responses. J. Neurosci. 32, 1602–1611. https://doi.org/10.1523/JNEUROSCI.5601-11.2012
- Mogi, M., Harada, M., Kondo, T., Riederer, P., Inagaki, H., Minami, M., Nagatsu, T., 1994a. Interleukin-1 beta, interleukin-6, epidermal growth factor and transforming growth factor-alpha are elevated in the brain from parkinsonian patients. Neurosci. Lett. 180, 147–150.
- Mogi, M., Harada, M., Riederer, P., Narabayashi, H., Fujita, K., Nagatsu, T., 1994b. Tumor necrosis factor-alpha (TNF-alpha) increases both in the brain and in the cerebrospinal fluid from parkinsonian patients. Neurosci. Lett. 165, 208–210.
- Mogi, M., Togari, A., Tanaka, K., Ogawa, N., Ichinose, H., Nagatsu, T., 2000. Increase in level of tumor necrosis factor-α in 6-hydroxydopamine-lesioned striatum in rats is suppressed by immunosuppressant FK506. Neurosci. Lett. 289, 165–168. https://doi.org/10.1016/S0304-3940(00)01275-1
- Morris, J.K., Bomhoff, G.L., Gorres, B.K., Davis, V.A., Kim, J., Lee, P.-P., Brooks, W.M., Gerhardt, G.A., Geiger, P.C., Stanford, J.A., 2011. Insulin resistance impairs nigrostriatal dopamine function. Exp. Neurol. 231, 171–180. https://doi.org/10.1016/j.expneurol.2011.06.005
- Morris, J.K., Bomhoff, G.L., Stanford, J.A., Geiger, P.C., 2010. Neurodegeneration in an animal model of Parkinson's disease is exacerbated by a high-fat diet. Am. J. Physiol.-Regul. Integr. Comp. Physiol. 299, R1082–R1090. https://doi.org/10.1152/ajpregu.00449.2010
- Mosser, D.M., Edwards, J.P., 2008. Exploring the full spectrum of macrophage activation. Nat. Rev. Immunol. 8, 958–969. https://doi.org/10.1038/nri2448
- Müller-Fielitz, H., Markert, A., Wittmershaus, C., Pahlke, F., Jöhren, O., Raasch, W., 2011. Weight loss and hypophagia after high-dose AT1-blockade is only observed after high dosing and depends on regular leptin signalling but not blood pressure. Naunyn. Schmiedebergs Arch. Pharmacol. 383, 373–384. https://doi.org/10.1007/s00210-011-0602-5
- Murphy, D.D., Rueter, S.M., Trojanowski, J.Q., Lee, V.M.-Y., 2000. Synucleins Are Developmentally Expressed, and α-Synuclein Regulates the Size of the Presynaptic Vesicular Pool in Primary Hippocampal Neurons. J. Neurosci. 20, 3214–3220. https://doi.org/10.1523/JNEUROSCI.20-09-03214.2000

- Murphy, K.E., Gysbers, A.M., Abbott, S.K., Spiro, A.S., Furuta, A., Cooper, A., Garner, B., Kabuta, T., Halliday, G.M., 2015. Lysosomal-associated membrane protein 2 isoforms are differentially affected in early Parkinson's disease. Mov. Disord. Off. J. Mov. Disord. Soc. 30, 1639–1647. https://doi.org/10.1002/mds.26141
- Muzio, M., Bosisio, D., Polentarutti, N., D'amico, G., Stoppacciaro, A., Mancinelli, R., van't Veer, C., Penton-Rol, G., Ruco, L.P., Allavena, P., Mantovani, A., 2000. Differential Expression and Regulation of Toll-Like Receptors (TLR) in Human Leukocytes: Selective Expression of TLR3 in Dendritic Cells1. J. Immunol. 164, 5998–6004. https://doi.org/10.4049/jimmunol.164.11.5998
- Nagatsu, T., 1995. Tyrosine hydroxylase: human isoforms, structure and regulation in physiology and pathology. Essays Biochem. 30, 15–35.
- Neystat, M., Lynch, T., Przedborski, S., Kholodilov, N., Rzhetskaya, M., Burke, R.E., 1999. Alpha-synuclein expression in substantia nigra and cortex in Parkinson's disease. Mov. Disord. Off. J. Mov. Disord. Soc. 14, 417–422. https://doi.org/10.1002/1531-8257(199905)14:3<417::aid-mds1005>3.0.co;2-x
- Nicklas, W.J., Youngster, S.K., Kindt, M.V., Heikkila, R.E., 1987. IV. MPTP, MPP+ and mitochondrial function. Life Sci. 40, 721–729. https://doi.org/10.1016/0024-3205(87)90299-2
- Nimmerjahn, A., Kirchhoff, F., Helmchen, F., 2005. Resting microglial cells are highly dynamic surveillants of brain parenchyma in vivo. Science 308, 1314–1318. https://doi.org/10.1126/science.1110647
- Nishimura, Y., Ito, T., Hoe, K.-L., Saavedra, J.M., 2000. Chronic peripheral administration of the angiotensin II AT1 receptor antagonist Candesartan blocks brain AT1 receptors. Brain Res. 871, 29–38. https://doi.org/10.1016/S0006-8993(00)02377-5
- Nissen, S.K., Shrivastava, K., Schulte, C., Otzen, D.E., Goldeck, D., Berg, D., Møller, H.J., Maetzler, W., Romero-Ramos, M., 2019. Alterations in Blood Monocyte Functions in Parkinson's Disease. Mov. Disord. Off. J. Mov. Disord. Soc. 34, 1711–1721. https://doi.org/10.1002/mds.27815
- Noelker, C., Morel, L., Lescot, T., Osterloh, A., Alvarez-Fischer, D., Breloer, M., Henze, C., Depboylu, C., Skrzydelski, D., Michel, P.P., Dodel, R.C., Lu, L., Hirsch, E.C., Hunot, S., Hartmann, A., 2013. Toll like receptor 4 mediates cell death in a mouse MPTP model of Parkinson disease. Sci. Rep. 3, 1393. https://doi.org/10.1038/srep01393
- Nyugen, J., Agrawal, S., Gollapudi, S., Gupta, S., 2010. Impaired Functions of Peripheral Blood Monocyte Subpopulations in Aged Humans. J. Clin. Immunol. 30, 806–813. https://doi.org/10.1007/s10875-010-9448-8
- Okafor, M., Wan, L., Saleh, S., Moore, R.H., Shaw, L.M., Goldstein, F.C., Hajjar, I., 2021. Candesartan is associated with lower amyloid accumulation in non-hypertensive prodromal Alzheimer's disease. Alzheimers Dement. 17, e056594. https://doi.org/10.1002/alz.056594
- Olanow, C.W., Prusiner, S.B., 2009. Is Parkinson's disease a prion disorder? Proc. Natl. Acad. Sci. 106, 12571–12572. https://doi.org/10.1073/pnas.0906759106

- Olson, J.K., Miller, S.D., 2004. Microglia Initiate Central Nervous System Innate and Adaptive Immune Responses through Multiple TLRs. J. Immunol. 173, 3916–3924. https://doi.org/10.4049/jimmunol.173.6.3916
- Otsuka, M., Takahashi, H., Shiratori, M., Chiba, H., Abe, S., 2004. Reduction of bleomycin induced lung fibrosis by candesartan cilexetil, an angiotensin II type 1 receptor antagonist. Thorax 59, 31–38. https://doi.org/10.1136/thx.2003.000893
- Ou, Z., Pan, J., Tang, S., Duan, D., Yu, D., Nong, H., Wang, Z., 2021. Global Trends in the Incidence, Prevalence, and Years Lived With Disability of Parkinson's Disease in 204 Countries/Territories From 1990 to 2019. Front. Public Health 9.
- Ouchi, Y., Yoshikawa, E., Sekine, Y., Futatsubashi, M., Kanno, T., Ogusu, T., Torizuka, T., 2005. Microglial activation and dopamine terminal loss in early Parkinson's disease. Ann. Neurol. 57, 168–175. https://doi.org/10.1002/ana.20338
- Palacino, J.J., Sagi, D., Goldberg, M.S., Krauss, S., Motz, C., Wacker, M., Klose, J., Shen, J., 2004. Mitochondrial dysfunction and oxidative damage in parkin-deficient mice. J. Biol. Chem. 279, 18614–18622. https://doi.org/10.1074/jbc.M401135200
- Palakurthi, B., Burugupally, S.P., 2019. Postural Instability in Parkinson's Disease: A Review. Brain Sci. 9, 239. https://doi.org/10.3390/brainsci9090239
- Panaro, M.A., Lofrumento, D.D., Saponaro, C., Nuccio, F.D., Cianciulli, A., Mitolo, V., Nicolardi, G., 2008. Expression of TLR4 and CD14 in the Central Nervous System (CNS) in a MPTP Mouse Model of Parkinson's-Like Disease. Immunopharmacol. Immunotoxicol. 30, 729–740. https://doi.org/10.1080/08923970802278557
- Pang, Y., Lin, S., Wright, C., Shen, J., Carter, K., Bhatt, A., Fan, L.-W., 2016. Intranasal insulin protects against substantia nigra dopaminergic neuronal loss and alleviates motor deficits induced by 6-OHDA in rats. Neuroscience 318, 157–165. https://doi.org/10.1016/j.neuroscience.2016.01.020
- Paumier, K.L., Luk, K.C., Manfredsson, F.P., Kanaan, N.M., Lipton, J.W., Collier, T.J., Steece-Collier, K., Kemp, C.J., Celano, S., Schulz, E., Sandoval, I.M., Fleming, S., Dirr, E., Polinski, N.K., Trojanowski, J.Q., Lee, V.M., Sortwell, C.E., 2015. Intrastriatal injection of pre-formed mouse α-synuclein fibrils into rats triggers αsynuclein pathology and bilateral nigrostriatal degeneration. Neurobiol. Dis. 82, 185– 199. https://doi.org/10.1016/j.nbd.2015.06.003
- Perez-Pardo, P., Dodiya, H.B., Broersen, L.M., Douna, H., van Wijk, N., Lopes da Silva, S., Garssen, J., Keshavarzian, A., Kraneveld, A.D., 2018. Gut–brain and brain–gut axis in Parkinson's disease models: Effects of a uridine and fish oil diet. Nutr. Neurosci. 21, 391–402. https://doi.org/10.1080/1028415X.2017.1294555
- Perez-Pardo, P., Dodiya, H.B., Engen, P.A., Forsyth, C.B., Huschens, A.M., Shaikh, M., Voigt, R.M., Naqib, A., Green, S.J., Kordower, J.H., Shannon, K.M., Garssen, J., Kraneveld, A.D., Keshavarzian, A., 2019. Role of TLR4 in the gut-brain axis in Parkinson's disease: a translational study from men to mice. Gut 68, 829–843. https://doi.org/10.1136/gutjnl-2018-316844

- Peter, I., Dubinsky, M., Bressman, S., Park, A., Lu, C., Chen, N., Wang, A., 2018. Anti–Tumor Necrosis Factor Therapy and Incidence of Parkinson Disease Among Patients With Inflammatory Bowel Disease. JAMA Neurol. 75, 939–946. https://doi.org/10.1001/jamaneurol.2018.0605
- Pike, A.F., Varanita, T., Herrebout, M.A.C., Plug, B.C., Kole, J., Musters, R.J.P., Teunissen, C.E., Hoozemans, J.J.M., Bubacco, L., Veerhuis, R., 2021. α-Synuclein evokes NLRP3 inflammasome-mediated IL-1β secretion from primary human microglia. Glia 69, 1413–1428. https://doi.org/10.1002/glia.23970
- Ping, Z., Xiaomu, W., Xufang, X., Liang, S., 2018. Vinpocetine regulates levels of circulating TLRs in Parkinson's disease patients. Neurol. Sci. https://doi.org/10.1007/s10072-018-3592-y
- Pinke, K.H., Calzavara, B., Faria, P.F., do Nascimento, M.P.P., Venturini, J., Lara, V.S., 2013. Proinflammatory profile of in vitro monocytes in the ageing is affected by lymphocytes presence. Immun. Ageing 10, 22. https://doi.org/10.1186/1742-4933-10-22
- Poehler, A.-M., Xiang, W., Spitzer, P., May, V.E.L., Meixner, H., Rockenstein, E., Chutna, O., Outeiro, T.F., Winkler, J., Masliah, E., Klucken, J., 2014. Autophagy modulates SNCA/α-synuclein release, thereby generating a hostile microenvironment. Autophagy 10, 2171–2192. https://doi.org/10.4161/auto.36436
- Polfliet, M.M.J., Fabriek, B.O., Daniëls, W.P., Dijkstra, C.D., van den Berg, T.K., 2006. The rat macrophage scavenger receptor CD163: Expression, regulation and role in inflammatory mediator production. Immunobiology, European Macrophage and Dendritic Cell Society 211, 419–425. https://doi.org/10.1016/j.imbio.2006.05.015
- Poltorak, A., He, X., Smirnova, I., Liu, M.-Y., Huffel, C.V., Du, X., Birdwell, D., Alejos, E., Silva, M., Galanos, C., Freudenberg, M., Ricciardi-Castagnoli, P., Layton, B., Beutler, B., 1998. Defective LPS Signaling in C3H/HeJ and C57BL/10ScCr Mice: Mutations in Tlr4 Gene. Science 282, 2085–2088. https://doi.org/10.1126/science.282.5396.2085
- Polymeropoulos, M.H., Lavedan, C., Leroy, E., Ide, S.E., Dehejia, A., Dutra, A., Pike, B., Root, H., Rubenstein, J., Boyer, R., Stenroos, E.S., Chandrasekharappa, S., Athanassiadou, A., Papapetropoulos, T., Johnson, W.G., Lazzarini, A.M., Duvoisin, R.C., Iorio, G.D., Golbe, L.I., Nussbaum, R.L., 1997. Mutation in the α-Synuclein Gene Identified in Families with Parkinson's Disease. Science 276, 2045–2047. https://doi.org/10.1126/science.276.5321.2045
- Ponsen, M.M., Stoffers, D., Booij, J., van Eck-Smit, B.L.F., Wolters, E.Ch., Berendse, H.W., 2004. Idiopathic hyposmia as a preclinical sign of Parkinson's disease. Ann. Neurol. 56, 173–181. https://doi.org/10.1002/ana.20160
- Postuma, R.B., Gagnon, J.-F., Bertrand, J.-A., Génier Marchand, D., Montplaisir, J.Y., 2015. Parkinson risk in idiopathic REM sleep behavior disorder. Neurology 84, 1104–1113. https://doi.org/10.1212/WNL.00000000001364
- Prediger, R.D.S., Batista, L.C., Medeiros, R., Pandolfo, P., Florio, J.C., Takahashi, R.N., 2006. The risk is in the air: Intranasal administration of MPTP to rats reproducing clinical

features of Parkinson's disease. Exp. Neurol. 202, 391–403. https://doi.org/10.1016/j.expneurol.2006.07.001

- Protopsaltis, T.S., Boniello, A.J., Schwab, F.J., 2016. Management of Spinal Deformity in Adult Patients With Neuromuscular Disease. JAAOS - J. Am. Acad. Orthop. Surg. 24, 634–644. https://doi.org/10.5435/JAAOS-D-15-00421
- Prusiner, S.B., 1998. Prions. Proc. Natl. Acad. Sci. 95, 13363–13383. https://doi.org/10.1073/pnas.95.23.13363
- Purisai, M.G., McCormack, A.L., Cumine, S., Li, J., Isla, M.Z., Di Monte, D.A., 2007. Microglial activation as a priming event leading to paraquat-induced dopaminergic cell degeneration. Neurobiol. Dis. 25, 392–400. https://doi.org/10.1016/j.nbd.2006.10.008
- Qie, S., Ran, Y., Lu, X., Su, W., Li, W., Xi, J., Gong, W., Liu, Z., 2020. Candesartan modulates microglia activation and polarization via NF-κB signaling pathway. Int. J. Immunopathol. Pharmacol. 34, 2058738420974900. https://doi.org/10.1177/2058738420974900
- Qin, L., Wu, X., Block, M.L., Liu, Y., Breese, G.R., Hong, J.-S., Knapp, D.J., Crews, F.T., 2007. Systemic LPS causes chronic neuroinflammation and progressive neurodegeneration. Glia 55, 453–462. https://doi.org/10.1002/glia.20467
- Qin, X.-Y., Zhang, S.-P., Cao, C., Loh, Y.P., Cheng, Y., 2016. Aberrations in Peripheral Inflammatory Cytokine Levels in Parkinson Disease: A Systematic Review and Metaanalysis. JAMA Neurol. 73, 1316–1324. https://doi.org/10.1001/jamaneurol.2016.2742
- Racette, B.A., Gross, A., Vouri, S.M., Camacho-Soto, A., Willis, A.W., Searles Nielsen, S., 2018. Immunosuppressants and risk of Parkinson disease. Ann. Clin. Transl. Neurol. 5, 870–875. https://doi.org/10.1002/acn3.580
- Ramalingam, M., Kim, S.-J., 2016. The Neuroprotective Role of Insulin Against MPP+-Induced Parkinson's Disease in Differentiated SH-SY5Y Cells. J. Cell. Biochem. 117, 917–926. https://doi.org/10.1002/jcb.25376
- Rannikko, E.H., Weber, S.S., Kahle, P.J., 2015. Exogenous α-synuclein induces toll-like receptor 4 dependent inflammatory responses in astrocytes. BMC Neurosci. 16, 57. https://doi.org/10.1186/s12868-015-0192-0
- Ransohoff, R.M., 2016. How neuroinflammation contributes to neurodegeneration. Science 353, 777–783. https://doi.org/10.1126/science.aag2590
- Reale, M., Iarlori, C., Thomas, A., Gambi, D., Perfetti, B., Di Nicola, M., Onofrj, M., 2009. Peripheral cytokines profile in Parkinson's disease. Brain. Behav. Immun. 23, 55–63. https://doi.org/10.1016/j.bbi.2008.07.003
- Rey, N.L., Steiner, J.A., Maroof, N., Luk, K.C., Madaj, Z., Trojanowski, J.Q., Lee, V.M.-Y., Brundin, P., 2016. Widespread transneuronal propagation of α-synucleinopathy triggered in olfactory bulb mimics prodromal Parkinson's disease. J. Exp. Med. 213, 1759–1778. https://doi.org/10.1084/jem.20160368

- Ribeiro, M., Rosenstock, T.R., Oliveira, A.M., Oliveira, C.R., Rego, A.C., 2014. Insulin and IGF-1 improve mitochondrial function in a PI-3K/Akt-dependent manner and reduce mitochondrial generation of reactive oxygen species in Huntington's disease knock-in striatal cells. Free Radic. Biol. Med. 74, 129–144. https://doi.org/10.1016/j.freeradbiomed.2014.06.023
- Rice, T.W., Wheeler, A.P., Bernard, G.R., Vincent, J.-L., Angus, D.C., Aikawa, N., Demeyer, I., Sainati, S., Amlot, N., Cao, C., Ii, M., Matsuda, H., Mouri, K., Cohen, J., 2010. A randomized, double-blind, placebo-controlled trial of TAK-242 for the treatment of severe sepsis*. Crit. Care Med. 38, 1685. https://doi.org/10.1097/CCM.0b013e3181e7c5c9
- Roberts, H.L., Brown, D.R., 2015. Seeking a mechanism for the toxicity of oligomeric αsynuclein. Biomolecules 5, 282–305. https://doi.org/10.3390/biom5020282
- Rochester, L., Yarnall, A.J., Baker, M.R., David, R.V., Lord, S., Galna, B., Burn, D.J., 2012. Cholinergic dysfunction contributes to gait disturbance in early Parkinson's disease. Brain J. Neurol. 135, 2779–2788. https://doi.org/10.1093/brain/aws207
- Rodrigues, L.S., Targa, A.D.S., Noseda, A.C.D., Aurich, M.F., Da Cunha, C., Lima, M.M.S., 2014. Olfactory impairment in the rotenone model of Parkinson's disease is associated with bulbar dopaminergic D2 activity after REM sleep deprivation. Front. Cell. Neurosci. 8, 383. https://doi.org/10.3389/fncel.2014.00383
- Rodriguez-Pallares, J., Rey, P., Parga, J.A., Muñoz, A., Guerra, M.J., Labandeira-Garcia, J.L., 2008. Brain angiotensin enhances dopaminergic cell death via microglial activation and NADPH-derived ROS. Neurobiol. Dis. 31, 58–73. https://doi.org/10.1016/j.nbd.2008.03.003
- Rodriguez-Perez, A.I., Sucunza, D., Pedrosa, M.A., Garrido-Gil, P., Kulisevsky, J., Lanciego, J.L., Labandeira-Garcia, J.L., 2018. Angiotensin Type 1 Receptor Antagonists Protect Against Alpha-Synuclein-Induced Neuroinflammation and Dopaminergic Neuron Death. Neurotherapeutics 15, 1063–1081. https://doi.org/10.1007/s13311-018-0646-z
- Roguski, A., Rayment, D., Whone, A.L., Jones, M.W., Rolinski, M., 2020. A Neurologist's Guide to REM Sleep Behavior Disorder. Front. Neurol. 11.
- Roodveldt, C., Labrador-Garrido, A., Gonzalez-Rey, E., Lachaud, C.C., Guilliams, T., Fernandez-Montesinos, R., Benitez-Rondan, A., Robledo, G., Hmadcha, A., Delgado, M., Dobson, C.M., Pozo, D., 2013. Preconditioning of Microglia by α-Synuclein Strongly Affects the Response Induced by Toll-like Receptor (TLR) Stimulation. PLOS ONE 8, e79160. https://doi.org/10.1371/journal.pone.0079160
- Ros-Bernal, F., Hunot, S., Herrero, M.T., Parnadeau, S., Corvol, J.-C., Lu, L., Alvarez-Fischer, D., Carrillo-de Sauvage, M.A., Saurini, F., Coussieu, C., Kinugawa, K., Prigent, A., Höglinger, G., Hamon, M., Tronche, F., Hirsch, E.C., Vyas, S., 2011. Microglial glucocorticoid receptors play a pivotal role in regulating dopaminergic neurodegeneration in parkinsonism. Proc. Natl. Acad. Sci. U. S. A. 108, 6632–6637. https://doi.org/10.1073/pnas.1017820108

- Ross, G.W., Petrovitch, H., Abbott, R.D., Nelson, J., Markesbery, W., Davis, D., Hardman, J., Launer, L., Masaki, K., Tanner, C.M., White, L.R., 2004. Parkinsonian signs and substantia nigra neuron density in decendents elders without PD. Ann. Neurol. 56, 532– 539. https://doi.org/10.1002/ana.20226
- Roubenoff, R., Harris, T.B., Abad, L.W., Wilson, P.W.F., Dallal, G.E., Dinarello, C.A., 1998.
 Monocyte Cytokine Production in an Elderly Population: Effect of Age and Inflammation. J. Gerontol. Ser. A 53A, M20–M26. https://doi.org/10.1093/gerona/53A.1.M20
- Russo, I., Bubacco, L., Greggio, E., 2014. LRRK2 and neuroinflammation: partners in crime in Parkinson's disease? J. Neuroinflammation 11, 52. https://doi.org/10.1186/1742-2094-11-52
- Saiki, M., Baker, A., Williams-Gray, C.H., Foltynie, T., Goodman, R.S., Taylor, C.J., Compston, D.A.S., Barker, R.A., Sawcer, S.J., Goris, A., 2010. Association of the human leucocyte antigen region with susceptibility to Parkinson's disease. J. Neurol. Neurosurg. Psychiatry 81, 890–891. https://doi.org/10.1136/jnnp.2008.162883
- Sampson, T.R., Debelius, J.W., Thron, T., Janssen, S., Shastri, G.G., Ilhan, Z.E., Challis, C., Schretter, C.E., Rocha, S., Gradinaru, V., Chesselet, M.-F., Keshavarzian, A., Shannon, K.M., Krajmalnik-Brown, R., Wittung-Stafshede, P., Knight, R., Mazmanian, S.K., 2016. Gut Microbiota Regulate Motor Deficits and Neuroinflammation in a Model of Parkinson's Disease. Cell 167, 1469-1480.e12. https://doi.org/10.1016/j.cell.2016.11.018
- Saunders, J.A.H., Estes, K.A., Kosloski, L.M., Allen, H.E., Dempsey, K.M., Torres-Russotto, D.R., Meza, J.L., Santamaria, P.M., Bertoni, J.M., Murman, D.L., Ali, H.H., Standaert, D.G., Mosley, R.L., Gendelman, H.E., 2012. CD4+ Regulatory and Effector/Memory T Cell Subsets Profile Motor Dysfunction in Parkinson's Disease. J. Neuroimmune Pharmacol. 7, 927–938. https://doi.org/10.1007/s11481-012-9402-z
- Schapira, A.H.V., 2009. Neurobiology and treatment of Parkinson's disease. Trends Pharmacol. Sci. 30, 41–47. https://doi.org/10.1016/j.tips.2008.10.005
- Scheperjans, F., Aho, V., Pereira, P.A.B., Koskinen, K., Paulin, L., Pekkonen, E., Haapaniemi, E., Kaakkola, S., Eerola-Rautio, J., Pohja, M., Kinnunen, E., Murros, K., Auvinen, P., 2015. Gut microbiota are related to Parkinson's disease and clinical phenotype. Mov. Disord. 30, 350–358. https://doi.org/10.1002/mds.26069
- Scherman, D., Desnos, C., Darchen, F., Pollak, P., Javoy-Agid, F., Agid, Y., 1989. Striatal dopamine deficiency in Parkinson's disease: role of aging. Ann. Neurol. 26, 551–557. https://doi.org/10.1002/ana.410260409
- Schlachetzki, J.C.M., Prots, I., Tao, J., Chun, H.B., Saijo, K., Gosselin, D., Winner, B., Glass, C.K., Winkler, J., 2018. A monocyte gene expression signature in the early clinical course of Parkinson's disease. Sci. Rep. 8, 10757. https://doi.org/10.1038/s41598-018-28986-7

- Schneeberger, A., Tierney, L., Mandler, M., 2016. Active immunization therapies for Parkinson's disease and multiple system atrophy. Mov. Disord. 31, 214–224. https://doi.org/10.1002/mds.26377
- Scott, K.M., Kouli, A., Yeoh, S.L., Clatworthy, M.R., Williams-Gray, C.H., 2018. A Systematic Review and Meta-Analysis of Alpha Synuclein Auto-Antibodies in Parkinson's Disease. Front. Neurol. 9, 815. https://doi.org/10.3389/fneur.2018.00815
- Sedelis, M., Schwarting, R.K.W., Huston, J.P., 2001. Behavioral phenotyping of the MPTP mouse model of Parkinson's disease. Behav. Brain Res. 125, 109–125. https://doi.org/10.1016/S0166-4328(01)00309-6
- Serdar, M., Kempe, K., Herrmann, R., Picard, D., Remke, M., Herz, J., Bendix, I., Felderhoff-Müser, U., Sabir, H., 2020. Involvement of CXCL1/CXCR2 During Microglia Activation Following Inflammation-Sensitized Hypoxic-Ischemic Brain Injury in Neonatal Rats. Front. Neurol. 11, 540878. https://doi.org/10.3389/fneur.2020.540878
- Shannon, K.M., Keshavarzian, A., Dodiya, H.B., Jakate, S., Kordower, J.H., 2012. Is alphasynuclein in the colon a biomarker for premotor Parkinson's Disease? Evidence from 3 cases. Mov. Disord. 27, 716–719. https://doi.org/10.1002/mds.25020
- Shao, Q., Chen, Y., Li, F., Wang, S., Zhang, X., Yuan, Y., Chen, N., 2019. TLR4 deficiency has a protective effect in the MPTP/probenecid mouse model of Parkinson's disease. Acta Pharmacol. Sin. 1–10. https://doi.org/10.1038/s41401-019-0280-2
- Shao, Q.-H., Yan, W.-F., Zhang, Z., Ma, K.-L., Peng, S.-Y., Cao, Y.-L., Yuan, Y.-H., Chen, N.-H., 2018a. Nurr1: A vital participant in the TLR4-NF-κB signal pathway stimulated by α-synuclein in BV-2 cells. Neuropharmacology. https://doi.org/10.1016/j.neuropharm.2018.04.008
- Shao, Q.-H., Yan, W.-F., Zhang, Z., Ma, K.-L., Peng, S.-Y., Cao, Y.-L., Yuan, Y.-H., Chen, N.-H., 2018b. Nurr1: A vital participant in the TLR4-NF-κB signal pathway stimulated by α-synuclein in BV-2 cells. Neuropharmacology. https://doi.org/10.1016/j.neuropharm.2018.04.008
- Shimazu, R., Akashi, S., Ogata, H., Nagai, Y., Fukudome, K., Miyake, K., Kimoto, M., 1999. MD-2, a molecule that confers lipopolysaccharide responsiveness on Toll-like receptor 4. J. Exp. Med. 189, 1777–1782. https://doi.org/10.1084/jem.189.11.1777
- Shimozawa, A., Ono, M., Takahara, D., Tarutani, A., Imura, S., Masuda-Suzukake, M., Higuchi, M., Yanai, K., Hisanaga, S., Hasegawa, M., 2017. Propagation of pathological α-synuclein in marmoset brain. Acta Neuropathol. Commun. 5, 12. https://doi.org/10.1186/s40478-017-0413-0
- Shin, W.-H., Jeon, M.-T., Leem, E., Won, S.-Y., Jeong, K.H., Park, S.-J., McLean, C., Lee, S.J., Jin, B.K., Jung, U.J., Kim, S.R., 2015. Induction of microglial toll-like receptor 4 by prothrombin kringle-2: a potential pathogenic mechanism in Parkinson's disease. Sci. Rep. 5, 14764. https://doi.org/10.1038/srep14764
- Simón-Sánchez, J., Schulte, C., Bras, J.M., Sharma, M., Gibbs, J.R., Berg, D., Paisan-Ruiz, C., Lichtner, P., Scholz, S.W., Hernandez, D.G., Krüger, R., Federoff, M., Klein, C., Goate, A., Perlmutter, J., Bonin, M., Nalls, M.A., Illig, T., Gieger, C., Houlden, H., Steffens,

M., Okun, M.S., Racette, B.A., Cookson, M.R., Foote, K.D., Fernandez, H.H., Traynor, B.J., Schreiber, S., Arepalli, S., Zonozi, R., Gwinn, K., van der Brug, M., Lopez, G., Chanock, S.J., Schatzkin, A., Park, Y., Hollenbeck, A., Gao, J., Huang, X., Wood, N.W., Lorenz, D., Deuschl, G., Chen, H., Riess, O., Hardy, J.A., Singleton, A.B., Gasser, T., 2009. Genome-wide association study reveals genetic risk underlying Parkinson's disease. Nat. Genet. 41, 1308–1312. https://doi.org/10.1038/ng.487

- Singleton, A.B., Farrer, M., Johnson, J., Singleton, A., Hague, S., Kachergus, J., Hulihan, M., Peuralinna, T., Dutra, A., Nussbaum, R., Lincoln, S., Crawley, A., Hanson, M., Maraganore, D., Adler, C., Cookson, M.R., Muenter, M., Baptista, M., Miller, D., Blancato, J., Hardy, J., Gwinn-Hardy, K., 2003. α-Synuclein Locus Triplication Causes Parkinson's Disease. Science 302, 841–841. https://doi.org/10.1126/science.1090278
- Skultetyova, D., Filipova, S., Riecansky, I., Skultety, J., 2007. The Role of Angiotensin Type 1 Receptor in Inflammation and Endothelial Dysfunction. Recent Patents Cardiovasc. Drug Discov. 2, 23–27. https://doi.org/10.2174/157489007779606130
- Smajić, S., Prada-Medina, C.A., Landoulsi, Z., Ghelfi, J., Delcambre, S., Dietrich, C., Jarazo, J., Henck, J., Balachandran, S., Pachchek, S., Morris, C.M., Antony, P., Timmermann, B., Sauer, S., Pereira, S.L., Schwamborn, J.C., May, P., Grünewald, A., Spielmann, M., 2022. Single-cell sequencing of human midbrain reveals glial activation and a Parkinson-specific neuronal state. Brain 145, 964–978. https://doi.org/10.1093/brain/awab446
- Smith, L., Schapira, A.H.V., 2022. GBA Variants and Parkinson Disease: Mechanisms and Treatments. Cells 11, 1261. https://doi.org/10.3390/cells11081261
- Sommer, A., Marxreiter, F., Krach, F., Fadler, T., Grosch, J., Maroni, M., Graef, D., Eberhardt, E., Riemenschneider, M.J., Yeo, G.W., Kohl, Z., Xiang, W., Gage, F.H., Winkler, J., Prots, I., Winner, B., 2018. Th17 Lymphocytes Induce Neuronal Cell Death in a Human iPSC-Based Model of Parkinson's Disease. Cell Stem Cell 23, 123-131.e6. https://doi.org/10.1016/j.stem.2018.06.015
- Sorokowski, P., Karwowski, M., Misiak, M., Marczak, M.K., Dziekan, M., Hummel, T., Sorokowska, A., 2019. Sex Differences in Human Olfaction: A Meta-Analysis. Front. Psychol. 10. https://doi.org/10.3389/fpsyg.2019.00242
- Spillantini, M.G., Schmidt, M.L., Lee, V.M.-Y., Trojanowski, J.Q., Jakes, R., Goedert, M., 1997. α-Synuclein in Lewy bodies. Nature 388, 839–840. https://doi.org/10.1038/42166
- Sriram, K., Matheson, J.M., Benkovic, S.A., Miller, D.B., Luster, M.I., O'Callaghan, J.P., 2002. Mice deficient in TNF receptors are protected against dopaminergic neurotoxicity: implications for Parkinson's disease. FASEB J. Off. Publ. Fed. Am. Soc. Exp. Biol. 16, 1474–1476. https://doi.org/10.1096/fj.02-0216fje
- Srivastava, G., Dixit, A., Yadav, S., Patel, D.K., Prakash, O., Singh, M.P., 2012. Resveratrol potentiates cytochrome P450 2d22-mediated neuroprotection in maneb- and paraquatinduced parkinsonism in the mouse. Free Radic. Biol. Med. 52, 1294–1306. https://doi.org/10.1016/j.freeradbiomed.2012.02.005

- Stefanis, L., 2012. α-Synuclein in Parkinson's Disease. Cold Spring Harb. Perspect. Med. 2, a009399. https://doi.org/10.1101/cshperspect.a009399
- Stefanova, N., Fellner, L., Reindl, M., Masliah, E., Poewe, W., Wenning, G.K., 2011. Toll-like receptor 4 promotes α-synuclein clearance and survival of nigral dopaminergic neurons. Am. J. Pathol. 179, 954–963. https://doi.org/10.1016/j.ajpath.2011.04.013
- Stokholm, M.G., Danielsen, E.H., Hamilton-Dutoit, S.J., Borghammer, P., 2016. Pathological α-synuclein in gastrointestinal tissues from prodromal Parkinson disease patients. Ann. Neurol. 79, 940–949. https://doi.org/10.1002/ana.24648
- Stoll, A.C., Sortwell, C.E., 2022. Leveraging the preformed fibril model to distinguish between alpha-synuclein inclusion- and nigrostriatal degeneration-associated immunogenicity. Neurobiol. Dis. 171, 105804. https://doi.org/10.1016/j.nbd.2022.105804
- Stolzenberg, E., Berry, D., Yang, D., Lee, E.Y., Kroemer, A., Kaufman, S., Wong, G.C.L., Oppenheim, J.J., Sen, S., Fishbein, T., Bax, A., Harris, B., Barbut, D., Zasloff, M.A., 2017. A Role for Neuronal Alpha-Synuclein in Gastrointestinal Immunity. J. Innate Immun. 9, 456–463. https://doi.org/10.1159/000477990
- Su, X., Fischer, D.L., Li, X., Bankiewicz, K., Sortwell, C.E., Federoff, H.J., 2017. Alpha-Synuclein mRNA Is Not Increased in Sporadic PD and Alpha-Synuclein Accumulation Does Not Block GDNF Signaling in Parkinson's Disease and Disease Models. Mol. Ther. 25, 2231–2235. https://doi.org/10.1016/j.ymthe.2017.04.018
- Sulzer, D., Alcalay, R.N., Garretti, F., Cote, L., Kanter, E., Agin-Liebes, J., Liong, C., McMurtrey, C., Hildebrand, W.H., Mao, X., Dawson, V.L., Dawson, T.M., Oseroff, C., Pham, J., Sidney, J., Dillon, M.B., Carpenter, C., Weiskopf, D., Phillips, E., Mallal, S., Peters, B., Frazier, A., Lindestam Arlehamn, C.S., Sette, A., 2017. T cells from patients with Parkinson's disease recognize α-synuclein peptides. Nature 546, 656–661. https://doi.org/10.1038/nature22815
- Sun, L., Jiang, W.-W., Wang, Y., Yuan, Y.-S., Rong, Z., Wu, J., Fan, Y., Lu, M., Zhang, K.-Z., 2021. Phosphorylated α-synuclein aggregated in Schwann cells exacerbates peripheral neuroinflammation and nerve dysfunction in Parkinson's disease through TLR2/NF-κB pathway. Cell Death Discov. 7, 1–10. https://doi.org/10.1038/s41420-021-00676-w
- Sun, X., Shi, B., Zheng, H., Min, L., Yang, J., Li, X., Liao, X., Huang, W., Zhang, M., Xu, S., Zhu, Z., Cui, H., Liu, X., 2018. Senescence-associated secretory factors induced by cisplatin in melanoma cells promote non-senescent melanoma cell growth through activation of the ERK1/2-RSK1 pathway. Cell Death Dis. 9, 260. https://doi.org/10.1038/s41419-018-0303-9
- Sun, X., Yao, H., Douglas, R.M., Gu, X.Q., Wang, J., Haddad, G.G., 2010. Insulin/PI3K signaling protects dentate neurons from oxygen-glucose deprivation in organotypic slice cultures. J. Neurochem. 112, 377–388. https://doi.org/10.1111/j.1471-4159.2009.06450.x

- Surmeier, D.J., Obeso, J.A., Halliday, G.M., 2017. Selective neuronal vulnerability in Parkinson disease. Nat. Rev. Neurosci. 18, 101–113. https://doi.org/10.1038/nrn.2016.178
- Suzuki, Y., Ruiz-Ortega, M., Lorenzo, O., Ruperez, M., Esteban, V., Egido, J., 2003. Inflammation and angiotensin II. Int. J. Biochem. Cell Biol., Renin-Angiotensin Systems: State of the Art 35, 881–900. https://doi.org/10.1016/S1357-2725(02)00271-6
- Svensson, E., Horváth-Puhó, E., Thomsen, R.W., Djurhuus, J.C., Pedersen, L., Borghammer, P., Sørensen, H.T., 2015a. Vagotomy and subsequent risk of Parkinson's disease. Ann. Neurol. 78, 522–529. https://doi.org/10.1002/ana.24448
- Svensson, E., Horváth-Puhó, E., Thomsen, R.W., Djurhuus, J.C., Pedersen, L., Borghammer, P., Sørensen, H.T., 2015b. Does vagotomy reduce the risk of Parkinson's disease: The authors reply. Ann. Neurol. 78, 1012–1013. https://doi.org/10.1002/ana.24518
- Świątkiewicz, M., Zaremba, M., Joniec, I., Członkowski, A., Kurkowska-Jastrzębska, I., 2013. Potential neuroprotective effect of ibuprofen, insights from the mice model of Parkinson's disease. Pharmacol. Rep. 65, 1227–1236. https://doi.org/10.1016/S1734-1140(13)71480-4
- Tagliaferro, P., Burke, R.E., 2016. Retrograde Axonal Degeneration in Parkinson Disease. J. Park. Dis. 6, 1–15. https://doi.org/10.3233/JPD-150769
- Taguchi, T., Ikuno, M., Yamakado, H., Takahashi, R., 2020. Animal Model for Prodromal Parkinson's Disease. Int. J. Mol. Sci. 21, 1961. https://doi.org/10.3390/ijms21061961
- Tan, E.-K., Chandran, V.R., Fook-Chong, S., Shen, H., Yew, K., Teoh, M.-L., Yuen, Y., Zhao, Y., 2005. Alpha-synuclein mRNA expression in sporadic Parkinson's disease. Mov. Disord. Off. J. Mov. Disord. Soc. 20, 620–623. https://doi.org/10.1002/mds.20391
- Tang, S.-C., Arumugam, T.V., Xu, X., Cheng, A., Mughal, M.R., Jo, D.G., Lathia, J.D., Siler, D.A., Chigurupati, S., Ouyang, X., Magnus, T., Camandola, S., Mattson, M.P., 2007. Pivotal role for neuronal Toll-like receptors in ischemic brain injury and functional deficits. Proc. Natl. Acad. Sci. U. S. A. 104, 13798–13803. https://doi.org/10.1073/pnas.0702553104
- Tanik, S.A., Schultheiss, C.E., Volpicelli-Daley, L.A., Brunden, K.R., Lee, V.M.Y., 2013. Lewy body-like α-synuclein aggregates resist degradation and impair macroautophagy. J. Biol. Chem. 288, 15194–15210. https://doi.org/10.1074/jbc.M113.457408
- Thomsen, M.B., Ferreira, S.A., Schacht, A.C., Jacobsen, J., Simonsen, M., Betzer, C., Jensen, P.H., Brooks, D.J., Landau, A.M., Romero-Ramos, M., 2021. PET imaging reveals early and progressive dopaminergic deficits after intra-striatal injection of preformed alpha-synuclein fibrils in rats. Neurobiol. Dis. 149, 105229. https://doi.org/10.1016/j.nbd.2020.105229
- Tian, J., Song, T., Wang, H., Wang, W., Ma, X., Hu, Y., 2021. Toll-Like Receptor 2 Antagonist Ameliorates Type 2 Diabetes Mellitus Associated Neuropathic Pain by Repolarizing Pro-inflammatory Macrophages. Neurochem. Res. 46, 2276–2284. https://doi.org/10.1007/s11064-021-03365-3

- Tian, Y.-M., Chen, X., Luo, D.-Z., Zhang, X.-H., Xue, H., Zheng, L.-F., Yang, N., Wang, X.-M., Zhu, J.-X., 2008. Alteration of dopaminergic markers in gastrointestinal tract of different rodent models of Parkinson's disease. Neuroscience 153, 634–644. https://doi.org/10.1016/j.neuroscience.2008.02.033
- Trapp, B.D., Wujek, J.R., Criste, G.A., Jalabi, W., Yin, X., Kidd, G.J., Stohlman, S., Ransohoff, R., 2007. Evidence for synaptic stripping by cortical microglia. Glia 55, 360–368. https://doi.org/10.1002/glia.20462
- Trudler, D., Farfara, D., Frenkel, D., 2010. Toll-Like Receptors Expression and Signaling in Glia Cells in Neuro-Amyloidogenic Diseases: Towards Future Therapeutic Application [WWW Document]. Mediators Inflamm. https://doi.org/10.1155/2010/497987
- Trudler, D., Nazor, K.L., Eisele, Y.S., Grabauskas, T., Dolatabadi, N., Parker, J., Sultan, A., Zhong, Z., Goodwin, M.S., Levites, Y., Golde, T.E., Kelly, J.W., Sierks, M.R., Schork, N.J., Karin, M., Ambasudhan, R., Lipton, S.A., 2021. Soluble α-synuclein–antibody complexes activate the NLRP3 inflammasome in hiPSC-derived microglia. Proc. Natl. Acad. Sci. 118, e2025847118. https://doi.org/10.1073/pnas.2025847118
- Tu, H.-Y., Yuan, B.-S., Hou, X.-O., Zhang, X.-J., Pei, C.-S., Ma, Y.-T., Yang, Y.-P., Fan, Y., Qin, Z.-H., Liu, C.-F., Hu, L.-F., 2021. α-synuclein suppresses microglial autophagy and promotes neurodegeneration in a mouse model of Parkinson's disease. Aging Cell 20, e13522. https://doi.org/10.1111/acel.13522
- Ullrich, C., Mlekusch, R., Kuschnig, A., Marksteiner, J., Humpel, C., 2010. Ubiquitin enzymes, ubiquitin and proteasome activity in blood mononuclear cells of MCI, Alzheimer and Parkinson patients. Curr. Alzheimer Res. 7, 549–555. https://doi.org/10.2174/156720510792231766
- Valkovic, P., Minar, M., Singliarova, H., Harsany, J., Hanakova, M., Martinkova, J., Benetin, J., 2015. Pain in Parkinson's Disease: A Cross-Sectional Study of Its Prevalence, Types, and Relationship to Depression and Quality of Life. PLOS ONE 10, e0136541. https://doi.org/10.1371/journal.pone.0136541
- Venezia, S., Kaufmann, W.A., Wenning, G.K., Stefanova, N., 2021. Toll-like receptor 4 deficiency facilitates α-synuclein propagation and neurodegeneration in a mouse model of prodromal Parkinson's disease. Parkinsonism Relat. Disord. 91, 59–65. https://doi.org/10.1016/j.parkreldis.2021.09.007
- Venezia, S., Refolo, V., Polissidis, A., Stefanis, L., Wenning, G.K., Stefanova, N., 2017. Tolllike receptor 4 stimulation with monophosphoryl lipid A ameliorates motor deficits and nigral neurodegeneration triggered by extraneuronal α-synucleinopathy. Mol. Neurodegener. 12, 52. https://doi.org/10.1186/s13024-017-0195-7
- Visanji, N.P., Brotchie, J.M., 2005. MPTP-Induced Models of Parkinson's Disease in Mice and Non-Human Primates. Curr. Protoc. Pharmacol. 29, 5.42.1-5.42.13. https://doi.org/10.1002/0471141755.ph0542s29
- Volpicelli-Daley, L.A., Luk, K.C., Lee, V.M.-Y., 2014. Addition of exogenous α-Synuclein Pre-formed fibrils to Primary Neuronal Cultures to seed recruitment of endogenous α-

Synuclein to Lewy body and Lewy Neurite-like aggregates. Nat. Protoc. 9, 2135–2146. https://doi.org/10.1038/nprot.2014.143

- Volpicelli-Daley, L.A., Luk, K.C., Patel, T.P., Tanik, S.A., Riddle, D.M., Stieber, A., Meaney, D.F., Trojanowski, J.Q., Lee, V.M.-Y., 2011. Exogenous α-Synuclein Fibrils Induce Lewy Body Pathology Leading to Synaptic Dysfunction and Neuron Death. Neuron 72, 57–71. https://doi.org/10.1016/j.neuron.2011.08.033
- Wakabayashi, K., Takahashi, H., Takeda, S., Ohama, E., Ikuta, F., 1988. Parkinson's disease: the presence of Lewy bodies in Auerbach's and Meissner's plexuses. Acta Neuropathol. (Berl.) 76, 217–221. https://doi.org/10.1007/BF00687767
- Wallings, R.L., Tansey, M.G., 2019. LRRK2 regulation of immune-pathways and inflammatory disease. Biochem. Soc. Trans. 47, 1581–1595. https://doi.org/10.1042/BST20180463
- Walsh, S., Merrick, R., Milne, R., Brayne, C., 2021. Aducanumab for Alzheimer's disease? BMJ 374, n1682. https://doi.org/10.1136/bmj.n1682
- Watson, M.B., Richter, F., Lee, S.K., Gabby, L., Wu, J., Masliah, E., Effros, R.B., Chesselet, M.-F., 2012. Regionally-specific microglial activation in young mice over-expressing human wildtype alpha-synuclein. Exp. Neurol. 237, 318–334. https://doi.org/10.1016/j.expneurol.2012.06.025
- Webb, J.L., Ravikumar, B., Atkins, J., Skepper, J.N., Rubinsztein, D.C., 2003. α-Synuclein Is Degraded by Both Autophagy and the Proteasome*. J. Biol. Chem. 278, 25009–25013. https://doi.org/10.1074/jbc.M300227200
- White, A.J., Wijeyekoon, R.S., Scott, K.M., Gunawardana, N.P., Hayat, S., Solim, I.H., McMahon, H.T., Barker, R.A., Williams-Gray, C.H., 2018. The Peripheral Inflammatory Response to Alpha-Synuclein and Endotoxin in Parkinson's Disease. Front. Neurol. 9, 946. https://doi.org/10.3389/fneur.2018.00946
- Wijeyekoon, R.S., Kronenberg-Versteeg, D., Scott, K.M., Hayat, S., Kuan, W.-L., Evans, J.R., Breen, D.P., Cummins, G., Jones, J.L., Clatworthy, M.R., Floto, R.A., Barker, R.A., Williams-Gray, C.H., 2020. Peripheral innate immune and bacterial signals relate to clinical heterogeneity in Parkinson's disease. Brain. Behav. Immun. https://doi.org/10.1016/j.bbi.2020.01.018
- Williams-Gray, C.H., Mason, S.L., Evans, J.R., Foltynie, T., Brayne, C., Robbins, T.W., Barker, R.A., 2013. The CamPaIGN study of Parkinson's disease: 10-year outlook in an incident population-based cohort. J. Neurol. Neurosurg. Psychiatry 84, 1258–1264. https://doi.org/10.1136/jnnp-2013-305277
- Williams-Gray, C.H., Wijeyekoon, R., Yarnall, A.J., Lawson, R.A., Breen, D.P., Evans, J.R., Cummins, G.A., Duncan, G.W., Khoo, T.K., Burn, D.J., Barker, R.A., ICICLE-PD study group, 2016. Serum immune markers and disease progression in an incident Parkinson's disease cohort (ICICLE-PD). Mov. Disord. Off. J. Mov. Disord. Soc. 31, 995–1003. https://doi.org/10.1002/mds.26563

- Williams-Gray, C.H., Wijeyekoon, R.S., Scott, K.M., Hayat, S., Barker, R.A., Jones, J.L., 2018. Abnormalities of age-related T cell senescence in Parkinson's disease. J. Neuroinflammation 15, 166. https://doi.org/10.1186/s12974-018-1206-5
- Wilms, H., Rosenstiel, P., Sievers, J., Deuschl, G., Zecca, L., Lucius, R., 2003. Activation of microglia by human neuromelanin is NF-κB-dependent and involves p38 mitogenactivated protein kinase: implications for Parkinson's disease. FASEB J. 17, 1–20. https://doi.org/10.1096/fj.02-0314fje
- Winslow, A.R., Chen, C.-W., Corrochano, S., Acevedo-Arozena, A., Gordon, D.E., Peden, A.A., Lichtenberg, M., Menzies, F.M., Ravikumar, B., Imarisio, S., Brown, S., O'Kane, C.J., Rubinsztein, D.C., 2010. α-Synuclein impairs macroautophagy: implications for Parkinson's disease. J. Cell Biol. 190, 1023–1037. https://doi.org/10.1083/jcb.201003122
- Witoelar, A., Jansen, I.E., Wang, Y., Desikan, R.S., Gibbs, J.R., Blauwendraat, C., Thompson, W.K., Hernandez, D.G., Djurovic, S., Schork, A.J., Bettella, F., Ellinghaus, D., Franke, A., Lie, B.A., McEvoy, L.K., Karlsen, T.H., Lesage, S., Morris, H.R., Brice, A., Wood, N.W., Heutink, P., Hardy, J., Singleton, A.B., Dale, A.M., Gasser, T., Andreassen, O.A., Sharma, M., International Parkinson's Disease Genomics Consortium (IPDGC), North American Brain Expression Consortium (NABEC), and United Kingdom Brain Expression Consortium (UKBEC) Investigators, 2017. Genome-wide Pleiotropy Between Parkinson Disease and Autoimmune Diseases. JAMA Neurol. 74, 780–792. https://doi.org/10.1001/jamaneurol.2017.0469
- Wooten, G.F., Currie, L.J., Bovbjerg, V.E., Lee, J.K., Patrie, J., 2004. Are men at greater risk for Parkinson's disease than women? J. Neurol. Neurosurg. Psychiatry 75, 637–639. https://doi.org/10.1136/jnnp.2003.020982
- Wright, S.D., Ramos, R.A., Tobias, P.S., Ulevitch, R.J., Mathison, J.C., 1990. CD14, a receptor for complexes of lipopolysaccharide (LPS) and LPS binding protein. Science 249, 1431–1433. https://doi.org/10.1126/science.1698311
- Xia, Y., Zhang, G., Han, C., Ma, K., Guo, X., Wan, F., Kou, L., Yin, S., Liu, L., Huang, J., Xiong, N., Wang, T., 2019. Microglia as modulators of exosomal alpha-synuclein transmission. Cell Death Dis. 10, 1–15. https://doi.org/10.1038/s41419-019-1404-9
- Xu, Q., Park, Y., Huang, X., Hollenbeck, A., Blair, A., Schatzkin, A., Chen, H., 2011. Diabetes and Risk of Parkinson's Disease. Diabetes Care 34, 910–915. https://doi.org/10.2337/dc10-1922
- Yamamoto, M., Sato, S., Hemmi, H., Hoshino, K., Kaisho, T., Sanjo, H., Takeuchi, O., Sugiyama, M., Okabe, M., Takeda, K., Akira, S., 2003. Role of adaptor TRIF in the MyD88-independent toll-like receptor signaling pathway. Science 301, 640–643. https://doi.org/10.1126/science.1087262
- Yan, J.-Q., Yuan, Y.-H., Gao, Y.-N., Huang, J.-Y., Ma, K.-L., Gao, Y., Zhang, W.-Q., Guo, X.-F., Chen, N.-H., 2014. Overexpression of human E46K mutant α-synuclein impairs macroautophagy via inactivation of JNK1-Bcl-2 pathway. Mol. Neurobiol. 50, 685– 701. https://doi.org/10.1007/s12035-014-8738-1

- Yao, C., Oh, J.-H., Lee, D.H., Bae, J.-S., Jin, C.L., Park, C.-H., Chung, J.H., 2015. Toll-like receptor family members in skin fibroblasts are functional and have a higher expression compared to skin keratinocytes. Int. J. Mol. Med. 35, 1443–1450. https://doi.org/10.3892/ijmm.2015.2146
- Zarember, K.A., Godowski, P.J., 2002. Tissue expression of human Toll-like receptors and differential regulation of Toll-like receptor mRNAs in leukocytes in response to microbes, their products, and cytokines. J. Immunol. Baltim. Md 1950 168, 554–561. https://doi.org/10.4049/jimmunol.168.2.554
- Zeller, S., Choi, C.S., Uchil, P.D., Ban, H.-S., Siefert, A., Fahmy, T.M., Mothes, W., Lee, S.-K., Kumar, P., 2015. Attachment of cell-binding ligands to arginine-rich cellpenetrating peptides enables cytosolic translocation of complexed siRNA. Chem. Biol. 22, 50–62. https://doi.org/10.1016/j.chembiol.2014.11.009
- Zeng, Bai-Yun, Bukhatwa, S., Hikima, A., Rose, S., Jenner, P., 2006. Reproducible nigral cell loss after systemic proteasomal inhibitor administration to rats. Ann. Neurol. 60, 248– 252. https://doi.org/10.1002/ana.20932
- Zeng, B.-Y., Iravani, M.M., Lin, S.-T., Irifune, M., Kuoppamäki, M., Al-Barghouthy, G., Smith, L., Jackson, M.J., Rose, S., Medhurst, A.D., Jenner, P., 2006. MPTP treatment of common marmosets impairs proteasomal enzyme activity and decreases expression of structural and regulatory elements of the 26S proteasome. Eur. J. Neurosci. 23, 1766– 1774. https://doi.org/10.1111/j.1460-9568.2006.04718.x
- Zhang, D., Kanthasamy, Arthi, Yang, Y., Anantharam, V., Kanthasamy, Anumantha, 2007. Protein Kinase Cδ Negatively Regulates Tyrosine Hydroxylase Activity and Dopamine Synthesis by Enhancing Protein Phosphatase-2A Activity in Dopaminergic Neurons. J. Neurosci. 27, 5349–5362. https://doi.org/10.1523/JNEUROSCI.4107-06.2007
- Zhang, X., Sun, X., Wang, J., Tang, L., Xie, A., 2017. Prevalence of rapid eye movement sleep behavior disorder (RBD) in Parkinson's disease: a meta and meta-regression analysis. Neurol. Sci. Off. J. Ital. Neurol. Soc. Ital. Soc. Clin. Neurophysiol. 38, 163–170. https://doi.org/10.1007/s10072-016-2744-1
- Zhao, J., Han, X., Xue, L., Zhu, K., Liu, H., Xie, A., 2015. Association of TLR4 gene polymorphisms with sporadic Parkinson's disease in a Han Chinese population. Neurol. Sci. 36, 1659–1665. https://doi.org/10.1007/s10072-015-2227-9
- Zhao, L.-Q., Huang, J.-L., Yu, Y., Lu, Y., Fu, L.-J., Wang, J.-L., Wang, Y.-D., Yu, C., 2013. Candesartan inhibits LPS-induced expression increase of toll-like receptor 4 and downstream inflammatory factors likely via angiotensin II type 1 receptor independent pathway in human renal tubular epithelial cells. Sheng Li Xue Bao 65, 623–630.
- Zhao, X.-D., Wang, F.-X., Cao, W.-F., Zhang, Y.-H., Li, Y., 2016. TLR4 signaling mediates AP-1 activation in an MPTP-induced mouse model of Parkinson's disease. Int. Immunopharmacol. 32, 96–102. https://doi.org/10.1016/j.intimp.2016.01.010
- Zhou, P., Weng, R., Chen, Z., Wang, R., Zou, J., Liu, X., Liao, J., Wang, Y., Xia, Y., Wang, Q., 2016. TLR4 Signaling in MPP+-Induced Activation of BV-2 Cells. Neural Plast. 2016, 5076740. https://doi.org/10.1155/2016/5076740

- Zhou, Y., Ye, L., Wan, Q., Zhou, L., Wang, X., Li, J., Hu, S., Zhou, D., Ho, W., 2009. Activation of Toll-like receptors inhibits herpes simplex virus-1 infection of human neuronal cells. J. Neurosci. Res. 87, 2916–2925. https://doi.org/10.1002/jnr.22110
- Zhu, Y., Yuan, M., Liu, Y., Yang, F., Chen, W.-Z., Xu, Z.-Z., Xiang, Z.-B., Xu, R.-S., 2021. Association between inflammatory bowel diseases and Parkinson's disease: systematic review and meta-analysis. Neural Regen. Res. 17, 344–353. https://doi.org/10.4103/1673-5374.317981
- Zimprich, A., Biskup, S., Leitner, P., Lichtner, P., Farrer, M., Lincoln, S., Kachergus, J., Hulihan, M., Uitti, R.J., Calne, D.B., Stoessl, A.J., Pfeiffer, R.F., Patenge, N., Carbajal, I.C., Vieregge, P., Asmus, F., Müller-Myhsok, B., Dickson, D.W., Meitinger, T., Strom, T.M., Wszolek, Z.K., Gasser, T., 2004. Mutations in LRRK2 Cause Autosomal-Dominant Parkinsonism with Pleomorphic Pathology. Neuron 44, 601–607. https://doi.org/10.1016/j.neuron.2004.11.005